

Thomson, Sarah (2016) *Cryptosporidiosis in farm livestock*. PhD thesis.

http://theses.gla.ac.uk/7096/

Copyright and moral rights for this thesis are retained by the author

A copy can be downloaded for personal non-commercial research or study

This thesis cannot be reproduced or quoted extensively from without first obtaining permission in writing from the Author

The content must not be changed in any way or sold commercially in any format or medium without the formal permission of the Author

When referring to this work, full bibliographic details including the author, title, awarding institution and date of the thesis must be given

Glasgow Theses Service http://theses.gla.ac.uk/ theses@gla.ac.uk

Cryptosporidiosis in Farm Livestock

Sarah Thomson



Submitted in fulfilment of the requirements for the Degree of Doctor of Philosophy, Institute of Biodiversity Animal Health and Comparative Medicine University of Glasgow 2016

Abstract

Although diarrhoea caused by *Cryptosporidium* is prevalent in livestock species throughout the world relatively little is known about the species and subtypes of *Cryptosporidium* found in cattle on Scottish farms. In particular, little is known about the shedding profiles (age when calves become infected and duration of shedding) of the different species found in cattle and how calves become infected. There are several theories about how neonatal calves first become infected with the parasite but the role which adult cattle play in the transmission of the parasite has not been fully addressed. It was previously thought that adult cattle did not become infected with the same species of *Cryptosporidium* which causes disease in the young calves. Some studies have shown that this may not be true and with the advance of new techniques to discriminate species this is an area which should be revisited.

In addition, it is known that it is possible for humans to become infected with *Cryptosporidium* and show clinical disease early in life and then again later in adulthood. In livestock however, diarrhoea caused by the parasite is generally only seen in neonatal livestock while older animals tend to be asymptomatic. It is not known if this resistance to clinical disease at an older age is due to changes in the host with an increase in age or if prior infection "immunises" the animal and provides protection against re-infection. It is also not known if infection with one isolate of *C. parvum* will provide protection against infection with another or if the protection formed is species/isolate specific.

The main aims of this thesis were to: determine the species and subtypes of *Cryptosporidium* found in calves on a study farm over a one year period from birth; assess the role which adult cattle play in the transmission of the parasite to newborn calves; develop new typing tools to enable the rapid and easy differentiation of *Cryptosporidium* species found in cattle and to examine the host-pathogen interactions in animals given serial experimental challenges with distinct *Cryptosporidium parvum* isolates to determine if the resistance seen in older animals on farms is due to an increase in age or as a result of prior infection.

A variety of different approaches were taken to achieve these aims. Longitudinal experiments carried out on a study farm revealed that in calves <9 weeks of age the most common species of *Cryptosporidium* is *C. parvum* and that all calves in the group became infected with *Cryptosporidium* within the first two weeks of life. Sample collection from the same animals later in life (at 6 months of age) showed that contrary to most previous studies the most common species detected at in this age group was also *C. parvum* although, interestingly, the subtype which the calves were shedding was not the same subtype that they were shedding previously.

The longitudinal study which investigated the role of adult cattle in the transmission of *Cryptosporidium* also yielded some interesting results. It was found that most of the adult cattle on this farm were shedding *Cryptosporidium* albeit intermittently. Speciation of the positive samples revealed that, on this farm, the most predominant species of *Cryptosporidium* in adult cattle was also *C. parvum*. This is very unusual as most previous studies have not found this level of infection in older cattle and *C. parvum* is not usually found in this age group. A number of different subtypes were found in adult cattle and some animals shed more than one subtype over the course of the study. This contradicts prior findings which demonstrated that only one subtype is found on a single farm.

The experimental infection trial involving infection of young (<1 week old) and older (6 week old) lambs with distinct *C. parvum* isolates demonstrated that an increase in age at primary infection reduces the effect of clinical disease. Animals which were infected at <1 week of age were re-challenged at 6 weeks of age with either a homologous or heterologous infection. Results revealed that previous exposure does not protect against re-infection with the same or a different isolate of *C. parvum*. This study also demonstrated that an increase in infective dose leads to a shorter pre-patent period and that there are variations in the clinical manifestations of different isolates of the same *Cryptosporidium* species.

Table of contents

Abstract	. ii
List of Tables	X
List of Figures	xii
Acknowledgements	xv
Author's Declaration	xvi
Definitions/Abbreviations	. xvii
Chapter 1: General Introduction	1
1.1 Background	1
1.2 Life Cycle	2
1.3 Species of Cryptosporidium	4
1.3.1 Species of <i>Cryptosporidium</i> in farm livestock	5
1.3.1.1 Cattle	5
1.3.1.2 Sheep and Goats	5
1.3.1.3 Pigs	5
1.3.2 Species of <i>Cryptosporidium</i> in poultry	6
1.3.3 Species of <i>Cryptosporidium</i> in non-mammalian vertebrates	6
1.3.3.1 Fish	6
1.3.3.2 Amphibians and reptiles	7
1.3.4 Species of Cryptosporidium in humans	/
1.3.5 Other Cryptosporidium species	ð
1.3.5.1 Companion animals	ð
1.3.5.2 Horses	9
1.4 Diagnosis and identification of english	···· 9
1.4 Diagnosis and identification of species	10
	10
1.4.1 Enzyme Infinutioassays	۲۲
1.5 Cryptosporidium subtyping	12
1.5 an60 Subtyning	13
15.7 gpool subtyping	15
1.6 Transmission of Cryptosporidium	17
1.6.1 Infective Dose	18
1.6.2 Oocyst Shedding (calves)	19
1.7 Cryptosporidiosis	19
1.7.1 Disease in farm livestock	20
1.7.1.1 Cattle	21

1.7.1.2 Sheep and goats	22
1.7.1.3 Poultry	23
1.7.2 Humans	24
1.7.3 Immunity to disease	25
1.7.3.1 Host Immune Response	26
1.7.4 Control of cryptosporidiosis	28
1.7.5 Therapeutics	28
1.7.5.1 Livestock	28
1.7.5.2 Humans	29
1.7.6 Vaccines	29
1.7.7 Disinfectants	30
1.7.8 Farm management	30
1.8 Summary	31
1.9 Aims	32
Chapter 2: Materials and Methods	33
2.1 Cattle Sample Collection	33
2.1.1 Dairy Calves	33
2.1.2 Adult Dairy Cattle	33
2.2 Experimental Animals	33
2.2.1 Lambs	33
2.2.1.1 Pre-amplification of <i>Cryptosporidium</i> isolates	34
2.2.1.2 Study animals	34
2.2.2 Clinical Scoring	34
2.2.3 Blood Sampling	36
2.2.4 Faecal Collection	36
2.2.4.1 Harness and bag design	37
2.3 Cryptosporidium oocysts	37
2.3.1 Collection of faeces for oocyst purification	37
2.3.2 Identification, quantification, extraction and purification of p	ositive
samples	38
2.3.2.1 Identification and quantification	38
2.3.2.2 Extraction and purification of oocysts from positive sample	es 39
2.3.3 Viability of oocysts and inoculum preparation	39
2.3.3.1 Excystation assay	39
2.3.3.2 Preparation of inoculum	41
2.4 Sample Processing	41
2.4.1 Adult Dairy Cattle	41
2.4.2 Experimental Lambs	42
2.5 DNA Extraction	42

2.5.1	Adult Cattle Faecal Samples	42
2.5.2	Calf and Lamb Faecal Samples	43
2.6 Poly	ymerase Chain Reaction (PCR)	43
2.6.1	Cryptosporidium Specific 18S rRNA PCR	43
2.6.2	Nested Species Specific Multiplex PCR (nssm-PCR)	44
2.6.3	gp60 Subtyping	45
2.7 Seq	uencing	46
2.7.1	Purification of PCR products	46
2.7.2	DNA quantification	46
2.7.3	Sequencing	46
2.7.4	Sequence analysis	47
2.8 Sta	tistical Analysis	47
Chapter 3:	Can infection with one isolate of Cryptosporidium parvum	
protect aga	inst infection with another?	48
3.1 Intr	oduction	48
3.2 Aim	IS	51
3.3 Mat	erials and Methods	52
3.3.1		52
3.3.2	Experimental Design	52
3.3.3	Cryptosporidium parvum isolates	53
3.3.4	Analysis of Calf Faecal Samples	55
3.3.4	.1 Microsatellite PCR primers and conditions	55
3.3.4		56
3.3.4	3.3 gp60 subtyping	5/
3.3.5	Purification from faeces and preparation of inoculum	5/
3.3.5	- Pre-amplification in lambs	5/
3.3.6	Experimental Animals	58
3.3./	Collection of samples	59
3.3./	1.1 Faecal Collection	59
3.3./	.2 Blood Sampling	59
3.3.8	Clinical Assessment	60
3.3.9	Sample Processing	60
3.3.9	.1 Oocysts counts	61
3.3.10	Statistical Analysis	62
3.3.1	0.1 Clinical Disease	63
3.3.1	U.2 Oocyst Shedding	63
3.4 Res	ults	64
3.4.1	Cryptosporidium parvum isolates	64
3.4.1	.1 Microsatellite analysis to identify suitable isolates	64

vi

3.	4.1.2	Faecal Consistency and Oocyst Shedding in Sampled Calves.	64
3.	4.1.3	C. parvum isolates used in experimental trial	65
3.4.	2 Pr	oduction of inoculum	65
3.4.	3 Pro	e-amplification of <i>Cryptosporidium parvum</i> isolates	66
3.	4.3.1	Sample Processing	66
3.	4.3.2	Excystation Assay	67
3.4.	4 Ex	perimental Animals - clinical observations	67
3.	4.4.1	Age-related Susceptibility to Disease	68
3.	4.4.2	Difference between isolates	70
3.	4.4.3	Previous exposure (homologous and heterologous challenge)	73
3.4.	5 Ex	perimental Animals - oocyst shedding	74
3.	4.5.1	Group I	75
3.4.	6 Gr	oup II	78
3.4.	7 Gr	oup III	81
3.4.	8 Gr	oup V	86
3.1 I	Effect	of age and infective dose on shedding	89
3.1.	1 Va	riation between isolates	89
3.1.	2 Pr	evious Exposure	90
3.2 I	Discus	sion	92
3.2.	1 Ag	e-related clinical disease and oocyst shedding	92
3.2.	2 Ch	allenge dose, Clinical Disease and oocyst shedding	95
3.2.	3 Cr	yptosporidium parvum isolates	97
3.3	Conclu	ision	99
Chapter	4:	Development of a molecular tool to differentiate between t	he
most coi	mmon	species of Cryptosporidium in cattle	. 101
4.1	Introdi	Jction	101
4.2	Aims .		103
4.3 /	Materi	als and Methods	104
4.3.	1 De	sign of <i>Cryptosporidium</i> species specific primers	104
4.3.	Z Fa	ecal Sample Collection and DNA extraction	109
4.3. (nss	3 NE m-PCR	sted species specific multiplex - Polymerase Chain Reaction	110
4	331	Testing of individual species specific primers	110
4	3 3 2	Field Samples	110
4.	3.3.3	Optimisation of PCR cycling conditions	
4.3.	4 Se	quence Analysis	
4.4	Result	5	112
4.4	1 Pr	imer Design	112
4.4.	2 Inc	dividual Species PCR	112

4.4	.3 Op	otimisation of PCR cycling conditions	112
4.4	.4 Ne	ested Species Specific Multiplex PCR	112
4.5	Discus	sion	118
Chapte	r 5:	Cryptosporidium species and subtypes found in calves fro	m
birth to	one y	ear	120
5.1	Introd	uction	120
5.2	Aims .	· · · · · · · · ·	122
5.3	Materi	als and Methods	123
5.3	.1 Sa	mple Collection	123
5.3	.2 Sa	mple processing and DNA extraction	123
5.3	.3 PC	R amplification of the 18S rRNA gene	123
5.3	.4 Sp Itinley	ecies identification by nssm-PCR (nested species specific	174
53	5 PC	\dot{R} amplification and sequencing of the gn60 gene for C nar	
sub	otyping	in amplification and sequencing of the 3000 gene for e. par	124
5.4	Result	s	125
5.4 mo	.1 Pr nths 12	evalence of <i>Cryptosporidium</i> in calves from birth to twelve 5	
5.4	.2 Sp	eciation of Cryptosporidium-positive samples by nssm-PCR .	128
5.4	.3 Su	btyping of C. parvum by gp60 PCR	130
	_ .		
5.5	Discus	sion	131
5.5 5.6	Discus Conclu	sion	131 135
5.5 5.6 Chapte	Discus Conclu r 6:	sion Ision Transmission of <i>Cryptosporidium</i> from dam to calf	131 135 136
5.5 5.6 Chapte 6.1	Discus Conclu r 6: Introd	sion Ision Transmission of <i>Cryptosporidium</i> from dam to calf uction	131 135 136 136
5.5 5.6 Chapte 6.1 6.2	Discus Conclu r 6: Introd Aims .	sion Ision Transmission of <i>Cryptosporidium</i> from dam to calf uction	131 135 136 136 139
5.5 5.6 Chapte 6.1 6.2 6.3	Discus Conclu r 6: Introd Aims . Materi	sion Ision Transmission of <i>Cryptosporidium</i> from dam to calf uction als and Methods	131 135 136 136 139 140
5.5 5.6 Chapte 6.1 6.2 6.3 6.3 cat	Discus Conclu r 6: Introd Aims . Materi .1 Op tle fae	sion Ision Transmission of <i>Cryptosporidium</i> from dam to calf uction als and Methods otimisation of oocyst concentration and DNA extraction from ces	131 135 136 136 139 140 adult 140
5.5 5.6 Chapte 6.1 6.2 6.3 6.3 cat	Discus Conclu r 6: Introd Aims . Materi .1 Op tle fae	sion Ision Transmission of <i>Cryptosporidium</i> from dam to calf uction als and Methods otimisation of oocyst concentration and DNA extraction from ces Spiking Experiment	131 135 136 136 139 140 adult 140 140
5.5 5.6 Chapte 6.1 6.2 6.3 6.3 cat 6	Discus Conclu r 6: Introd Aims . Materi .1 Op tle fae 0.3.1.1 0.3.1.2	sion Ision Transmission of <i>Cryptosporidium</i> from dam to calf uction als and Methods otimisation of oocyst concentration and DNA extraction from ces Spiking Experiment Field Samples.	131 135 136 136 139 140 adult 140 140 140
5.5 5.6 Chapte 6.1 6.2 6.3 6.3 cat 6 6	Discus Conclu r 6: Introd Aims . Materi .1 Op tle fae 0.3.1.1 0.3.1.2	sion Ision Transmission of Cryptosporidium from dam to calf uction als and Methods otimisation of oocyst concentration and DNA extraction from ces Spiking Experiment Field Samples. Acid Flocculation	131 135 136 136 139 140 adult 140 140 140 140
5.5 5.6 Chapte 6.1 6.2 6.3 6.3 cat 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6	Discus Conclu r 6: Introd Aims . Materi .1 Or tle fae 0.3.1.1 0.3.1.2 0.3.1.3	sion Ision Transmission of <i>Cryptosporidium</i> from dam to calf uction als and Methods otimisation of oocyst concentration and DNA extraction from ces Spiking Experiment Field Samples. Acid Flocculation	131 135 136 136 139 140 adult 140 140 140 140 141
5.5 5.6 Chapte 6.1 6.2 6.3 6.3 cat 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6	Discus Conclu r 6: Introd Aims . Materi .1 Or tle fae 5.3.1.1 5.3.1.2 5.3.1.3 5.3.1.4 5.3.1.5	sion Ision Transmission of Cryptosporidium from dam to calf uction als and Methods otimisation of oocyst concentration and DNA extraction from ces Spiking Experiment Field Samples. Acid Flocculation Salt Flotation DNA Extraction	131 135 136 136 139 140 adult 140 140 140 141
5.5 5.6 Chapte 6.1 6.2 6.3 6.3 cat 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6	Discus Conclu r 6: Introd Aims . Materi .1 Op tle fae 5.3.1.1 5.3.1.2 5.3.1.3 5.3.1.4 5.3.1.5 5.3.1.6	sion Transmission of Cryptosporidium from dam to calf uction als and Methods otimisation of oocyst concentration and DNA extraction from ces Spiking Experiment Field Samples Acid Flocculation Salt Flotation DNA Extraction PCR amplification of the 18S rRNA gene	131 135 136 136 139 140 adult 140 140 140 141 141
5.5 5.6 Chapte 6.1 6.2 6.3 6.3 cat 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6	Discus Conclu r 6: Introd Aims . Materi .1 Op tle fae 0.3.1.1 0.3.1.2 0.3.1.3 0.3.1.4 0.3.1.5 0.3.1.6 .2 Sa	sion Transmission of Cryptosporidium from dam to calf uction als and Methods otimisation of oocyst concentration and DNA extraction from ces Spiking Experiment Field Samples Acid Flocculation Salt Flotation DNA Extraction PCR amplification of the 18S rRNA gene mple Collection	131 135 136 136 139 140 adult 140 140 140 141 141 141 142
5.5 5.6 Chapte 6.1 6.2 6.3 6.3 cat 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6	Discus Conclu r 6: Introd Aims . Materi .1 Op tle fae 0.3.1.1 0.3.1.2 0.3.1.3 0.3.1.4 0.3.1.5 0.3.1.6 0.3.1.6 0.3.2.1	sion Transmission of Cryptosporidium from dam to calf uction als and Methods otimisation of oocyst concentration and DNA extraction from ces Spiking Experiment Field Samples Acid Flocculation Salt Flotation DNA Extraction PCR amplification of the 18S rRNA gene mple Collection Adults	131 135 136 136 139 140 140 140 140 141 141 141 141 142 142
5.5 5.6 Chapte 6.1 6.2 6.3 6.3 cat 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6	Discus Conclu r 6: Introd Aims . Materi .1 Or tle fae 5.3.1.1 5.3.1.2 5.3.1.3 5.3.1.4 5.3.1.5 5.3.1.6 .2 Sa 5.3.2.1 5.3.2.2	sion Transmission of Cryptosporidium from dam to calf uction als and Methods otimisation of oocyst concentration and DNA extraction from ces Spiking Experiment Field Samples Acid Flocculation DNA Extraction PCR amplification of the 18S rRNA gene mple Collection Adults Calves	131 135 136 136 139 140 140 140 140 140 141 141 141 142 142 142
5.5 5.6 Chapte 6.1 6.2 6.3 6.3 cat 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6	Discus Conclu r 6: Introd Aims . Materi .1 Op tle fae 5.3.1.1 5.3.1.2 5.3.1.3 5.3.1.4 5.3.1.5 5.3.1.6 .2 Sa 5.3.2.1 5.3.2.2 .3 Sa	sion Transmission of Cryptosporidium from dam to calf uction als and Methods otimisation of oocyst concentration and DNA extraction from ces Spiking Experiment Field Samples Acid Flocculation Salt Flotation DNA Extraction PCR amplification of the 18S rRNA gene mple Collection Adults Calves mple processing and DNA isolation	131 135 136 136 139 140 140 140 140 140 141 141 141 142 142 142 143
5.5 5.6 Chapte 6.1 6.2 6.3 6.3 cat 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6	Discus Conclu r 6: Introd Aims . Materi .1 Op tle fae 5.3.1.1 5.3.1.2 5.3.1.3 5.3.1.4 5.3.1.5 5.3.1.6 .2 Sa 5.3.2.1 5.3.2.2 .3 Sa 5.3.3.1	sion	131 135 136 136 136 139 140 140 140 140 140 141 141 141 142 142 142 143 143

List of Tables

Table 1-1: Currently recognised Cryptosporidium species and the hosts theyinfect4
Table 2-1: Scoring system used to score lambs daily 35
Table 2-2: Primer sequences and size of amplicon when used with primer pair forthe 18S rRNA gene
Table 2-3: Primer sequences and size of amplicon for nssm-PCR. 45
Table 2-4: Primer sequences and amplicon sizes for gp60 PCR
Table 3-1: Pre-amplification of C. parvum isolates. 52
Table 3-2: Primers used for microsatellite analysis of <i>Cryptosporidium</i> isolates 56
Table 3-3: Outcome and explanatory variables 62
Table 3-4: Alleles assigned to each isolate
Table 3-5: Faecal consistancy and average oocysts counts associated withdifferent C. parvum isolates in naturally infected calves
Table 3-6: Oocyst counts from faeces obtained from calves farms BB & LH \dots 65
Table 3-7: Sporozoite: Shell ratios and excystation percentages for each isolate 66
Table 3-8: Sporozoite:Shell ratios and excystation percentages for inoculum usedin experimental trial67
Table 3-9a: Log-rank test on Kaplan-Meier survival curves for onset of diarrhoeaafter challenge at <1 week of age.
Table 3-10: Multivariate logistic regression model for analysis of factorsassociated with occurrence of diarrhoea in wk1 - wk4 and wk6 - wk8
Table 3-11: Samples collected and processed from each lamb in Group I75
Table 3-12: Samples collected and processed from each lamb in Group II 78
Table 3-13: Samples collected and processed from each lamb in Group III 81
Table 3-14: Samples collected and processed from each lamb in Group V 86
Table 4-1: Primer sequences and size of amplicon produced with each primerpair109
Table 4-2: Comparison of results of nssm-PCR with sequencing of 18S rRNA gene.

Table 5-1: Number of samples collected at each time point
Table 5-2: gp60 genotypes detected in calves at different ages
Table 6-1: Methods tested for optimisation of oocyst concentration before DNAextraction from cattle faeces.140
Table 6-2: Detection limits of each of the four concentration methods145
Table 6-3: gp60 subtypes detected in calves at different ages
Table 6-4: Summary of studies carried out investigating Cryptosporidium in adult cattle

List of Figures

Figure 1-1: Life cycle of Cryptosporidium 3
Figure 1-2: <i>Cryptosporidium</i> oocysts as seen through a microscope using different staining methods
Figure 1-3: <i>gp60</i> sequence showing microsatellite region with TCA, TCG and ACATCA repeats highlighted
Figure 1-4: Healthy gut villi (left) and damaged gut caused by <i>C. parvum</i> infection (right) in a calf
Figure 1-5: Species of <i>Cryptosporidium</i> generally associated with different age groups of cattle
Figure 1-6: Simplified diagram showing the most important steps and interactions in the acquired immune response against <i>Cryptosporidium</i> infection. 27
Figure 2-1: Lambs wearing harness and bags for collection of faeces
Figure 2-2: Example of the grid pattern on a haemocytometer
Figure 2-3: Excysted oocysts. Empty oocyst shells (dark circles) and sporozoites (bright "commas") can be seen
Figure 3-1: Experimental design
Figure 3-2: Lambs in specially designed feeder
Figure 3-3: Lambs with the harness and bag system to collect faeces
Figure 3-4: Sample processing 61
Figure 3-5: Shedding profile of lambs inoculated with BB (1A & 2A) and LH (1B & 2B) isolates
Figure 3-6: Average feed intake of each group
Figure 3-7: Number of days lambs produced a diarrhoeic sample following primary-infection at <1 week (Group III, IV, V & VI) and 6 weeks (Group I & II) of age
Figure 3-8: Time to onset of diarrhoea following primary challenge at <1 week (A) and 6 weeks (B) of age
Figure 3-9: Number of days lambs infected with BB (red) or LH (blue) isolate produced a diarrhoeic sample in the 3 weeks post-primary infection at <1 week of age
Figure 3-10: Occurrences of diarrhoea following challenge at 6 weeks

Figure 3-11: Oocyst counts from lambs in Group I76
Figure 3-12: Total number of oocysts shed by individual lambs in Group I (BB wk6)
Figure 3-13: Oocyst counts from lambs in Group II
Figure 3-14: Total number of oocysts shed by individual lambs in Group II 80
Figure 3-15: Oocyst counts from animals in Group III
Figure 3-16: Oocyst counts from animals in Group III excluding animal F 84
Figure 3-17: Total number of oocysts shed by individual lambs in Group III 85
Figure 3-18: Oocyst counts from Group V
Figure 3-19: Oocyst counts from Group V excluding animals E & F
Figure 3-20: Total number of oocysts shed by individual lambs in Group V 89
Figure 3-21 Average oocysts counts (Group I, II, III & V) following primary infection
Figure 4-1: Positions of species specific primers on the 18S rRNA gene plus the location of the 18S rRNA internal primers
Figure 4-2: Detection of species specific amplicons on a 2% agarose gel stained with GelRedTM (Biotium, UK)
Figure 5-1: Shedding profile of <i>Cryptosporidium</i> oocysts in calves from birth to 12 months
Figure 5-2: Positive and negative samples collected from each calf
Figure 5-3: Prevalence of <i>Cryptosporidium</i> species in each age group of cattle
Figure 6-1: Possible routes for transmission of <i>Cryptosporidium</i> to calves137
Figure 6-2: Spiking experiment to compare AF, SF, AF/SF and no concentration methods of oocyst detection from adult cattle faeces
Figure 6-3: Samples collected from each cow pre and post calving
Figure 6-4: Positive samples detected in adult cows pre- and post-calving 150
Figure 6-5: Number of samples collected from each calf
Figure 6-6: Shedding profile of Cryptosporidium oocysts in calves <9 weeks154

xiii

Figure 6-7: Positive and negative samples collected from each calf1	55
Figure 6-8: Species detected in individual cows1!	57
Figure 6-9: Diagrammatic representation of cows calving dates and <i>C. parvum</i> subtypes they shed10	66
Figure 6-10: Comparison between studies10	68

Acknowledgements

Firstly, I would like to thank my supervisors at the Moredun Research Institute, Dr. Frank(tastic) Katzer and Prof. Lee Innes very much for all their support and guidance over the past three and a bit years. You have been wonderful supervisors and it has been a pleasure working with you. Also many thanks to Prof. Nick Jonsson my University supervisor, particularly for help with statistical analysis. I am extremely grateful for all of the time and advice all of my supervisors have given me even when I am sure they were busy with more important things. Thank you also for enabling me to attend so many conferences and meetings to share my research with the world ©

I have thoroughly enjoyed my time at the Moredun Research Institute, it is a very friendly and fun place to work. I need to thank the Proto & Co team for making my experience here so positive. Thank you also for helping me with the huge lamb study, from making little poo bags and harnesses to feeding lambs and collecting samples. A special thank you to Alessandra Toroda for spending months in the lab making poo-smoothies and counting oocysts, to Dr. Emily Hotchkiss for assisting with my stats understanding and to Dr. Beth Wells for all the additional help she gave me above and beyond what was necessary.

I am very fortunate that my work colleagues are also my friends so I must thank them too, in particular Laura Doull for coming to pole with me so that we had something other than our PhD's to complain talk about. To Stefano Guido, Beth Wells, Marieke Opsteegh, Maria Parigi and Alessandra Toroda for being wonderful office mates and (sometimes) joining me in a spot of office yoga to relieve the stresses of the day as well as putting up with the general chaos that I cause (I <u>WILL</u> tidy my desk soon, I promise).

Finally I need to thank my family and friends from outside of work for their support and understanding over the course of my PhD and previous studies. I am sure that you all know at lot more about *Cryptosporidium* than you ever wanted to! A big thank you to Alistair who has listened with patience to my presentations and for offering constructive advice and just generally putting up with me ©

Author's Declaration

I declare that the work presented in this thesis is my own work unless stated. This work has not been previously submitted for any other degree or professional qualification.

Sarah Thomson

Definitions/Abbreviations

%	Percent
(NH ₄) ₂ SO ₄	Ammonium Sulphate
~	Approximately
<	Less Than
=	Equal To
>	More Than
°C	Degrees Centigrade
μg	Microgram
μι	Microlitre
μm	Micrometre
μΜ	Micromolar
ACVM	aniline-carbol-methyl-violet
AF	Acid Flocculation
AI	Artificial Insemination
AIDS	Acquired Immune Deficiency Syndrome
ANOVA	Analysis of Variance
AP	Auramine Phenol
APC	Antigen Presenting Cell
BB	C. parvum isolate BB
bp	base pairs
BSA	Bovine Serum Albumin
	T-helper cell expressing cluster of differentiation 4
CD4+ 1	glycoprotein
CE	Capillary Electrophoresis
COWP	Cryptosporidium Oocyst Wall Protein
d	Day
DAPI	4',6-diamidino-2-phenylindole
dH ₂ O	Distilled Water
DNA	Deoxyribonucleic Acid
dNTP	Deoxyribonucleic Triphosphate
DPI	Days Post-Infection
EDTA	Ethylenediaminetetraacetic Acid

EIA	Enzyme Immunoassay		
ELISA	Enzyme Linked Immunosorbent Assay		
EtOH	Ethanol		
FAM	Carboxyfluorescein		
fg	Femtogram		
FITC	Fluorescein Isothiocyanate		
g	grams		
g	Gravitational Force		
GP60	60-kDa glycoprotein		
gp60	60-kDa glycoprotein gene		
h	hour		
HBSS	Hanks Buffered Saline Solution		
HEX	Hexachlorofluorescein		
HIV	Human Immunodeficiency Virus		
HLA	Human Leukocyte Antigen		
HSP70	Heat Shock Protein 70		
ID ₅₀	Infectious Dose to 50% of Exposed Individuals		
IFAT	Immunofluorescence Antibody Test		
IFN-γ	Interferon-gamma		
IL	Interleukin		
IMS	Immunomagnetic Separation		
	C. parvum isolate which is derived from naturally infected		
Isolate	calves, differs at more than one loci and which differs in		
	clinical manifestation		
kDa	Kilo Dalton		
kg	Kilogram		
ι	Litre		
LH	C. parvum isolate LH		
mg	Milligram		
MgCl ₂	Magnesium Chloride		
mins	minutes		
ml	Millilitre		
MLG	Multilocus Genotype		
MLST	Multilocus Sequence Typing		
mm	Millimetre		

mΜ	Millimolar			
mZN	Modified Ziehl-Neelsen			
n	Number			
ng	Nanogram			
NGS	Next Generation Sequencing			
NK	Natural Killer			
nssm	Nested Species Specific Multiplex			
opg	Oocysts per gram			
PBS	Phosphate Buffered Saline			
PCR	Polymerase Chain Reaction			
рН	Negative log of the activity of the hydrogen ion in an			
	aqueous solution			
pmol	Pico mol			
qPCR	Quantitative PCR			
RFLP	Restriction Fragment Length Polymorphism			
rRNA	Ribonucleic Acid			
S	Second			
SDS	Sodium Dodecyl Sulphate			
SF	Salt Flotation			
spp.	Species			
SSU	Small Subunit			
TE	Tris(hydroxymethyl)aminomethane-			
	Ethylenediamintetraacetic Acid			
TNFα	Tumor Necrosis Factor Alpha			
TNFB	Tumor Necrosis Factor Beta			
Tris-HCl	Tris(hydroxymethyl)aminomethane-hydrochloride			
UHT	Ultra High Temperature			
UK	United Kingdom			
USA	United States of America			
UV	Ultra-Violet			
vs.	Versus			
wk	Week			
wk1	Infected at <1 week of age			
wk6	Infected at 6 weeks of age			

Chapter 1: General Introduction

1.1 Background

Cryptosporidiosis is the disease caused by a protozoan parasite of the genus *Cryptosporidium*, which was discovered in 1910 by Edward Ernst Tyzzer in the gastric glands of mice. Tyzzer noticed that this parasite did not contain sporocysts within the oocysts and it sporulates while still attached to the host wall. For this reason Tyzzer named the genus *Cryptosporidium*, from the Greek kruptos meaning 'hidden', and the parasite that he identified in the gastric glands, *Cryptosporidium muris*. Three years later Tyzzer identified another species of *Cryptosporidium* which was not infective in the gastric glands of mice but only in the small intestine (Tyzzer, 1910). This second species produced oocysts which were smaller in size (4-5 μ m) compared with *C. muris* oocysts (6-8 μ m), Tyzzer named this species *Cryptosporidium parvum* (parvum comes from the Latin for little). It was originally believed that there were only two different species of *Cryptosporidium* but there are now over 27 described species and many more genotypes (Chalmers and Katzer, 2013).

Cryptosporidiosis, the disease caused by some species of *Cryptosporidium* in susceptible hosts generally manifests itself as profuse watery diarrhoea, weightloss and inappetence. Depending on the parasite and host species other clinical signs may be seen such as vomiting and nausea in some humans infected with *Cryptosporidium hominis*. Despite being identified in many animal species, cryptosporidiosis was not thought to be a very important disease until the early 1980s and the HIV-AIDS epidemic, though it was first reported in humans in 1976 (Nime et al., 1976). Of the first seven reported cases of *Cryptosporidium* infection in humans, six were from immunocompromised patients leading to the belief that if the immune system was not compromised, cryptosporidiosis would not pose any health risk (Bird and Smith, 1980). Of course cryptosporidiosis is now recognised as an important zoonotic disease, which affects many species (Chalmers and Katzer, 2013).

Cryptosporidium was first associated with diarrhoea in calves in 1971 (Panciera et al., 1971) and since then has been identified as one of the most important

cause of neonatal enteritis in calves worldwide (de Graaf et al., 1999; Fayer et al., 1998; Thompson et al., 2005).

1.2 Life Cycle

Cryptosporidium is a monoxenous organism, (life cycle completed within a single host) which exists in the environment in thick walled oocysts. These oocysts are very resistant to many environmental factors such as high and low temperatures (Fujino et al., 2002) but they can be inactivated by desiccation (Robertson et al., 1992). *Cryptosporidium* oocysts survive well in cool, moist climates and can remain infective for many months (Robertson et al., 1992).

Infective *Cryptosporidium* oocysts (Figure 1-1(a)) are ingested by the host and when triggered by the conditions in the gastrointestinal tract (low pH, body temperature) excystation occurs (Hijjawi et al., 2002) and four sporozoites are released (Figure 1-1(b)). The sporozoites attach themselves to the host epithelial cells at the ileocaecal junction which they then invade and become engulfed by the cells forming a parasitophorous vacuole. Many other protozoan parasites also exist within a parasitophorous vacuole but *Cryptosporidium* also contains a unique structure known as the feeder organelle, which separates the cell and parasite cytoplasm (Pohlenz et al., 1987).

After the development of the feeder organelle the sporozoite itself becomes more spherical in shape and forms a trophozoite (Figure 1-1 (c)). The parasite begins asexual reproduction and develops into a Type I meront (Figure 1-1 (d)) or a Type II meront (Figure 1-1 (f)). Merozoites are formed within the Type I meront, each Type I meront releases 6-8 merozoites which can immediately reinfect the host, by invading neighbouring cells and beginning asexual reproduction again, or develop into a Type II meront (Figure 1-1 (f)).

Type II meronts release four merozoites which initiate the sexual reproductive cycle (Figure 1-1 (f)). The released merozoites invade host cells and differentiate into either macrogamonts (Figure 1-1 (h)) or microgamonts (Figure 1-1 (g)). Microgamonts develop multi-nuclei and release free microgametes which penetrate and fertilise the macrogamete, producing a zygote (Figure 1-1 (i)). Meiosis occurs and the zygote differentiates into four sporozoites as the

oocyst develops and is released from the lumen. The sporozoites may be released directly into the lumen from thin-walled oocysts and re-infect the host (Figure 1-1 (k)) or thick-walled oocysts which are shed in faeces and remain immediately infective for other hosts (Figure 1-1 (j)).



Figure 1-1: Life cycle of Cryptosporidium

Reproduced from Juranek DD. Cryptosporidiosis. In: Strickland GT, editor. Hunter's Tropical Medicine and Emerging Infectious Diseases, 8th ed. Philadelphia: WB Saunders; 2000. Originally adapted from the life cycle that appears in Current WL, Garcia LS. Cryptosporidiosis. Clinical Microbiol Rev 1991; 4:325-58.

1.3 Species of Cryptosporidium

There are 28 different species of *Cryptosporidium* currently recognised (Table 1-1) and many more genotypes (Chalmers and Katzer, 2013; Fayer, 2010). The most common species found in the UK and worldwide is the zoonotic species *C*. *parvum* (Chalmers and Katzer, 2013). Historically species of *Cryptosporidium* were named according to the host of origin on the assumption that most species were host-specific. While many of the species of *Cryptosporidium* do still appear to be host-specific others infect many different host species, for example, *C*. *parvum* and *C*. *ubiquitum* infect a wide variety of hosts.

Species Name	Synonym	Host Species	First Described By
C. andersoni*	Mouse genotype	Cattle	(Lindsay et al., 2000)
C. baileyi		Birds	(Current et al., 1986)
C. bovis	Genotype Bovine B	Cattle	(Fayer et al., 2005)
C. canis*	Canine genotype	Dogs	(Fayer et al., 2001)
C. cuniculus*	Rabbit genotype	Rabbit	(Robinson et al., 2010)
C. ducismarci		Tortoise	(Traversa, 2010)
C. fayeri	Opossum genotype 1	Marsupials	(Ryan et al., 2008)
C. felis*		Cats	(Iseki et al., 1989)
C. fragile		Toads	(Jirku et al., 2008)
C. galli	C. blagburni	Birds	(Ryan et al., 2003b)
C. hominis*	Genotype 1/genotype H	Humans	(Morgan-Ryan et al., 2002)
C. huwi		Fish	(Ryan et al., 2015)
C. marcopodum		Marsupials	(Power and Ryan, 2008)
C. meleagridis*		Birds	(, 1955a)
C. molnari		Fish	(Alvarez-Pellitero and Sitja-Bobadilla, 2002)
C. muris*		Mice	(Tyzzer, 1910)
C. parvum*	Genotype 2/genotype C	Mammals inc. humans	(Tyzzer, 1912)
C. ryanae	Deer-like genotype	Cattle	(Fayer et al., 2008)
C. scophthalmi		Fish	(Alvarez-Pellitero et al., 2004)
C. scruforum		Pigs	(Kvac et al., 2013)
C. serpentis		Snakes	(Levine, 1980)
C. suis*	Pig genotype 1	Pigs	(Ryan et al., 2004)
C. tyzzeri	Mouse genotype 1	Mice	(Ren et al., 2012)
C. ubiquitum*	Cervine genotype	Variety of animals	(Fayer et al., 2010b)
C. varanii	C. saurophilum	Lizards and Snakes	(Pavlasek and Ryan, 2008a)
C. viatorum		Humans	(Elwin et al., 2012b)
C. wrairi		Guinea Pigs	(Vetterling et al., 1971)
C. xiaoi	C. bovis-like genotype	Sheep	(Fayer and Santin, 2009)

Table 1-1: Currently recognised Cryptosporidium species and the hosts they infect

*also infects humans

1.3.1 Species of Cryptosporidium in farm livestock

Of the 27 or so *Cryptosporidium* species which have been identified about eight of these are known infect farm livestock (cattle, sheep and pigs).

1.3.1.1 Cattle

There are four species of *Cryptosporidium* which are commonly found in cattle; *C. parvum*, *C. bovis*, *C. ryanae* and *C. andersoni*. There are some reports of other species also being found; *C. hominis* has been reported in a 3 day old calf and a 6 year old cow from Scotland (Smith et al., 2005), *C. suis* in a calf (Fayer et al., 2006) and *C. felis* in a cow (Bornay-Llinares et al., 1999) but these reports are rare.

1.3.1.2 Sheep and Goats

Sheep and goats are predominantly infected with *C. parvum, C. xiaoi, C. bovis* and *C. ubiquitum* although rare occurrences of other species have been reported. *Cryptosporidium hominis* has been reported as the predominant species in sheep on the Scottish island of St Kilda (Connelly et al., 2013), in a sheep and a goat kid from a farm and petting zoo in the UK (Giles et al., 2009) and in a sheep in Australia (Ryan et al., 2005). *Cryptosporidium scruforum* and *C. andersoni* have been detected in sheep from flocks in Australia (Yang et al., 2014) and *C. fayeri* has been reported in a sheep (Ryan et al., 2005) again these occurrences of other species are rare.

1.3.1.3 Pigs

The predominant species of *Cryptosporidium* detected in pigs are *C. suis* and *C. scruforum*; *C. suis* is prevalent in pigs worldwide but causes few clinical signs (Enemark et al., 2003). *Cryptosporidium scruforum* is a fairly new species, only described in 2013 (Kvac et al., 2013), again, this species does not seem to cause any clinical disease in pigs yet is prevalent in adult pigs worldwide. It does not appear infective to pigs under 8 weeks of age (Kvac et al., 2014).

1.3.2 Species of *Cryptosporidium* in poultry

Three species of *Cryptosporidium* are currently known to infect birds; C. meleagridis, C. baileyi and C. galli. Avian cryptosporidiosis was first described in 1929 but was not formally recognised until 1955 when C. meleagridis was reported in turkeys (Slavin, 1955), C. meleagridis is found in the small and large intestine and bursa of infected birds. Cryptosporidium *meleagridis* has been reported in many different species of birds and also in humans (Ng-Hublin et al., 2013; Silverlas et al., 2012; Wang et al., 2014b). The second species of Cryptosporidium to be reported in birds was C. baileyi in chickens in 1986 (Current et al., 1986), this species is more commonly associated with respiratory cryptosporidiosis in broiler chickens and is the most common species of avian Cryptosporidium. Also described from chickens is C. galli which was discovered in 1999 but was redescribed in 2003 (Ryan et al., 2003a). Like C. meleagridis this species is also found in the small intestine, large intestine and in the epithelium of the proventriculus and is associated with clinical disease and mortality (Blagburn et al., 1991; Tumova et al., 2002). The three avian adapted species are morphologically different and can be distinguished by morphological difference in the size and shape of the oocysts.

1.3.3 Species of *Cryptosporidium* in non-mammalian vertebrates

1.3.3.1 Fish

In fish three species have been reported but it is possible that there are more which remain undiscovered as studies into the species of *Cryptosporidium* found in fish are scarce. The first species in fish to be described was *C. molnari* infecting the stomach of gilthead sea bream and European sea bass with few clinical signs (Alvarez-Pellitero and Sitja-Bobadilla, 2002). The other reported species of *Cryptosporidium* infecting fish is *C. scophthalmi* which is found in the intestinal epithelium of turbot. So far it appears that juvenile fish are most frequently infected with *Cryptosporidium* while it has rarely been detected in adults (Alvarez-Pellitero et al., 2004). Infected fish tend to show no signs of clinical infection but do show intestinal distension at necropsy. A more recent study, has also detected *C. parvum* in fish from Papua New Guinea (Koinari et al., 2013) as well as a novel *Cryptosporidium* species which is also zoonotic. The

most recently described species of *Cryptosporidium* is a species identified in the stomach of a guppy (Ryan et al., 2015).

1.3.3.2 Amphibians and reptiles

Four species of *Cryptosporidium* have been reported in amphibians and reptiles, C. fragile, C. varanii, C. serpentis and C. ducismarci. As its name suggests, C. fragile is the most delicate species of Cryptosporidium and was first described in the stomach of the black-spined toad (Jirku et al., 2008) and was named due to its fragile nature. Unlike other species of Cryptosporidium, C. fragile will crumple on contact with hypertonic solutions and disintegrate after just 4 weeks in water (Jirku et al., 2008). *Cryptosporidium* has been reported in many species of snakes and other reptiles, the first to be identified was C. serpentis which is a gastric species and was found in four snake species in 1977 (Brownstein et al., 1977). It is believed that C. serpentis may actually be a name which covers many other species of Cryptosporidium infecting snakes which have not yet been described (Fayer, 2010) as morphological data from many different species of reptiles indicates the possibility of multiple unique groups (Upton et al., 1989; Xiao and Ryan, 2004; Xiao et al., 2004). Cryptosporidium varanii was first described in an Emerald Monitor lizard (Pavlasek and Ryan, 2008b); it has since been identified in many other lizard and snake species. The most recently discovered Cryptosporidium species in a reptile is C. ducismarci which was described from tortoises (Traversa, 2010) and which caused fatal intestinal disease in two different species of tortoise. As with fish it is possible that many more species of *Cryptosporidium* infecting reptiles and amphibians may exist but not enough studies have been carried out to confirm this.

1.3.4 Species of Cryptosporidium in humans

In humans there are two species which are routinely diagnosed in clinical cases of cryptosporidiosis; these are the zoonotic species *C. parvum* and the human adapted species *C. hominis* (Morgan-Ryan et al., 2002). These species account for over 90% of human infections worldwide (Chalmers and Giles, 2010) and 96% of clinical cases in the UK are attributed to these two species (Hadfield et al., 2011). *Cryptosporidium parvum* infections peak in the spring months while *C. hominis* is more common in the autumn months (Chalmers et al., 2009a), this is

thought to be related to springtime calving and lambing and an increase in people participating in outdoor activities at this time of year. Other species which commonly infect humans include; C. meleagridis (Chalmers and Giles, 2010; Xiao and Fayer, 2008) in the UK and C. felis, C. canis and C. meleagridis in other parts of the world (Cama et al., 2008). Sporadic infections with other *Cryptosporidium* species have also occurred although these are not always associated with disease and may just be asymptomatic infections (Helmy et al., 2013; Robinson et al., 2008; Waldron et al., 2010). While most outbreaks in the UK are associated with either C. hominis or C. parvum one outbreak in 2008 was caused by C. cuniculus (Chalmers et al., 2009b; Robinson et al., 2010). *Cryptosporidium cuniculus* was originally reported as causing infection in rabbits in 1979 (Inman and Takeuchi, 1979) and was believed at this time to be host specific. Following the human outbreak of 2008 it was discovered that C. cuniculus is infective to humans and rabbits and under experimental conditions immunosupressed gerbils and mice can become infected (Robinson et al., 2010). There are no clinical signs recorded for rabbits. This species is very similar to C. hominis at several loci including 18S SSU, heat shock protein 70 (HSP), actin, LIB13 and *Cryptosporidium* oocyst wall protein (COWP), in 2010 a redescription of C. cuniculus was published (Robinson et al., 2010). In 2012 a new human adapted species was recognised and named C. viatorum, this species is associated with travellers who have visited the Indian subcontinent (Elwin et al., 2012b).

1.3.5 Other Cryptosporidium species

1.3.5.1 Companion animals

In the UK other *Cryptosporidium* species are commonly found in domestic pets, in one study it was found that 20% of cats and dogs were shedding *C. felis* and *C. canis* respectively (Mtambo et al., 1991). Worldwide the prevalence of *Cryptosporidium* in these companion animals ranges from 0-29.4% in cats and 0.5-44.1% in dogs (Lucio-Forster et al., 2010), *C. parvum* has been detected in dogs previously but these animals are not thought to be an important factor in zoonotic transmission of the parasite (Giangaspero et al., 2006; Lucio-Forster et al., 2010).

1.3.5.2 Horses

Cryptosporidium was first reported in horses in the late 1970s in immunodeficient foals (Snyder et al., 1978). At this time it was generally believed that in animals, as well as humans, cryptosporidiosis was only a problem where the host was immunocompromised in some way. Since the initial report of *Cryptosporidium* in foals there have been many reports of the parasite in immunocompetent horses (Santin, 2013). The most common clinical sign of cryptosporidiosis in foals is diarrhoea (Grinberg et al., 2009), in adult horses most infections tend to be asymptomatic (Majewska et al., 2004; Sturdee et al., 2003) although diarrhoea in adult horses has been reported (McKenzie and Diffay, 2000). The most commonly detected species in horses is *C. parvum* (Chalmers and Grinberg, 2005; Cole et al., 1998; Perrucci et al., 2011).

1.3.5.3 Wildlife

In British wildlife (rabbits, rodents, foxes and deer) various species and genotypes of *Cryptosporidium* have been identified although relatively few studies into wildlife hosts in the UK have been carried out (Appelbee et al., 2005). One study which examined a large number of wildlife hosts as well as livestock and horses identified *Cryptosporidium* oocysts, believed at the time to be C. parvum, in ~30% of faeces from small wild mammals (Sturdee et al., 2003) and in another study, oocysts were identified in 63% of brown rat faecal samples (Webster and Macdonald, 1995). A study in the UK investigating *Cryptosporidium* species in a water catchment found C. ubiquitum in deer samples and in foxes C. bovis and C. parvum were identified (Robinson et al., 2011). Although this study found no evidence of Cryptosporidium in rabbits an outbreak in humans in 2008 was linked to the rabbit genotype, now identified as a separate species C. cuniculus (Chalmers et al., 2009b; Robinson et al., 2010). A recent study investigating *Cryptosporidium* species and genotypes found in livestock and deer in a water catchment in north east Scotland reported a prevalence of 69.23% in deer (80% in red deer and 33% in roe deer) the majority of the samples were identified as *C. parvum* with the remainder identified as the *Cryptosporidium*deer genotype (Wells et al., 2015).

In other parts of the world different species have been identified in other wildlife hosts; *C.wrairi* was first identified in guinea pigs in 1966, this species has only ever been identified from guinea pigs and appears to be host specific (Jervis et al., 1966; Vetterling et al., 1971). In marsupials two species have been reported, *C. fayeri* which was isolated from a red kangaroo (Ryan et al., 2008) and *C. macropodum* which was first reported in an eastern grey kangaroo (Power and Ryan, 2008) and has since been reported in other marsupials such as; swamp wallabies, western grey kangaroos and red kangaroos (Power, 2010). Neither *C. fayeri* nor *C. macropodum* have been associated with clinical disease in marsupials (Yang et al., 2011).

1.4 Diagnosis and identification of species

Differentiation of *Cryptosporidium* species found in both humans and farm livestock is important as it enables disease epidemiology involving different species to be understood and in human disease cases outbreaks may be traced back to the original source and help inform control strategies (Chalmers et al., 2009a; Gormley et al., 2011). Many studies have examined the epidemiology of Cryptosporidium in humans (Caccio, 2005; Chalmers et al., 2009a; Fayer et al., 2000a; Xiao, 2010). However, similar studies in farm livestock are few and often only focus on the neonatal age group. Longitudinal studies in livestock are not very common though some have been carried out in France, the USA, the UK, Australia and Scandinavia (Brook et al., 2009; Follet et al., 2011; Robertson et al., 2010; Santin et al., 2008; Sturdee et al., 2003; Yang et al., 2014). During human outbreaks of cryptosporidiosis it is usual for the parasite to be speciated and even genotyped whereas in veterinary diagnosis often only the presence of the parasite is confirmed. The most common methods for detection of Cryptosporidium in the UK are; microscopy, enzyme immunoassays (EIA) and polymerase chain reaction (PCR) (Chalmers et al., 2010).

1.4.1 Microscopic Identification

Identifying only the presence of the parasite by microscopy can be problematic for diagnosis as many *Cryptosporidium* species do not cause disease in some hosts, for example even under experimental conditions *C. ryanae* and *C. bovis* do not cause diarrhoea in calves (Fayer et al., 2008; Fayer et al., 2005). There

are few morphological differences between Cryptosporidium species and internal structures cannot be easily seen and it is not possible to speciate most *Cryptosporidium* species by microscopy alone. There are some exceptions to this rule with some species of Cryptosporidium such as C. andersoni and C. muris being larger in size than the others (Lindsay et al., 2000). Many species of *Cryptosporidium* are not host specific and multiple species can often be found in the same host. The oocysts of *Cryptosporidium* are very small (3-8 µm) and between species there is overlap in oocyst size. Under microscopic examination *Cryptosporidium* oocysts can be difficult to distinguish from other small particles such as yeasts, moulds and faecal debris. Various techniques for detection of oocysts by microscopy exist (Figure 1-2). The most common method for detection of *Cryptosporidium* by microscopy in the UK is auramine phenol staining. With this method the oocysts are stained a bright greenish yellow against a dark background and sensitivity and specificity of this staining method is good (~92% sensitivity and 100% specificity). The 2nd most common method used in UK diagnostic laboratories for human specimens is modified Ziehl-Neelsen (mZN). Using mZN the oocysts appear red on a blue/green background and the sensitivity of this method is ~75% with 100% specificity, this technique is also commonly applied in veterinary diagnostics as well. Finally another common method of detection by microscopy is 4,6' diamidino-2-phenylindole dihydrochloride (DAPI) staining which stains the nuclei of viable oocysts and they appear blue under UV light, oocysts can also be detected without staining, using phase contrast microscopy, (Chalmers et al., 2011)



Figure 1-2: *Cryptosporidium* oocysts as seen through a microscope using different staining methods.

Phenol Auramine (AP) staining (A) stains the oocysts bright green, Modified Ziehl-Neelsen (mZN) (B) stains oocysts pink against a blue/green background, DAPI stains the nuclei of viable oocysts making them appear blue (C) and it is possible to visualize oocysts using phase-contrast microscopy without any staining (D). Images from (Robinson et al., 2010).

Microscopy techniques are commonly used for diagnostic purposes but most often in a research setting molecular tools are used instead as these are much more sensitive and provide far more additional information (Chalmers and Katzer, 2013). Many conserved genes have been targeted for the purpose of species differentiation including; those encoding 18S rRNA (Xiao et al., 1999), 70 kDa heat shock protein (HSP70) (Sulaiman et al., 2000), *Cryptosporidium* oocyst wall protein (COWP) (Xiao et al., 2000b) and the actin gene (Kim et al., 1992). Probably the most commonly used of these is the 18S rRNA gene as in addition to containing regions that vary between species there are several regions which are conserved within the *Cryptosporidium* genus. In each *Cryptosporidium* oocyst there are 20 copies of this gene (Le Blancq et al., 1997; Morgan and Thompson, 1999) meaning that sensitivity is increased when compared with a single-copy gene.

1.4.1 Enzyme Immunoassays

Enzyme immunoassays (EIA) are very common in the UK for detection of *Cryptosporidium* in human faecal samples and are very sensitive in detecting *C. parvum* (Garcia et al., 1987) but the ability of these kits to detect other *Cryptosporidium* species has not been thoroughly assessed. Some studies show a reduction in sensitivity when other *Cryptosporidium* species (*C. andersoni*) are present (Lindergard et al., 2001). EIAs use rabbit anti-*Cryptosporidium* polyclonal antibodies to capture the *Cryptosporidium* antigen from the stool supernatant. A second set of goat anti-*Cryptosporidium* polyclonal antibodies are then added, which sandwiches the captured *Cryptosporidium* antigen (Ungar, 1990). Sometimes these assays can be combined to allow the detection of other parasites such as *Giardia* at the same time. EIA are very sensitive (100%) and specific (~91-93%) but can be expensive to perform (Chalmers et al., 2011) which may prohibit their use in large studies.

1.4.2 Molecular detection of Cryptosporidium

Species differentiation using molecular methods can be done in a few different ways; these begin with DNA extraction from oocysts and PCR amplification of the gene(s) of interest. DNA extracted from oocysts can be amplified using standard or nested PCR methods. In standard PCR one pair of forward and reverse primers are used to amplify a gene or region of a gene, whereas in nested PCR two sets of primers are used. The first pair (external) amplify the gene of interest and in a second round; internal primers amplify a shorter sequence of the amplicon produced in the first PCR. This is useful if the sample only contains a small amount of DNA as it results in more DNA copies than standard PCR. The PCR products are separated by electrophoresis on an agarose gel and visualised by staining with a stain such as ethidium bromide which binds to nucleic acids and can be viewed under UV light. Species can be differentiated using restriction enzymes to digest PCR products into fragments of different sizes (PCR-Restriction Fragment Length Polymorphism (RFLP)). These fragments can be visualised on the agarose gel, producing varying banding patterns depending on species (Spano et al., 1997; Sulaiman et al., 1999). Another commonly used method is DNA sequencing of the PCR products, the PCR product is purified and re-amplified using either the same primers in the initial PCR or primers specifically designed for sequencing and fluorescently labelled nucleotides. The labelled nucleotides emit light at specific wavelengths giving the sequence of the DNA fragment which can be analysed using various computer based programs.

Another PCR method which can be used to detect and speciate *Cryptosporidium* parasites is real-time PCR which is considered to be the "gold-standard" for *Cryptosporidium* detection (Chalmers et al., 2011) as this method is the most sensitive and can detect as few as 2 oocysts per PCR (Hadfield et al., 2011). In real-time PCR the amplification of the DNA can be tracked in "real time" and the amount of DNA present can be quantified, this is an advantage over standard PCR which can only indicate the presence of parasite DNA but cannot quantify the amount.

1.5 Cryptosporidium subtyping

As well as speciating *Cryptosporidium* positive samples it can be useful to further subtype particular species, which can be informative in epidemiological studies and disease outbreaks to identify origin or source of particular subtypes and to study population genetics (Gatei et al., 2008; Mallon et al., 2003; Morrison et al., 2008; Wang et al., 2014b). Subtyping can be carried out by multilocus sequence typing (MLST) using several microsatellites to identify alleles, assigning a parasite a multilocus genotype (MLG). MLST is useful for studying population genetics but it is not always necessary to examine the parasite in such detail. Many studies have subtyped certain *Cryptosporidium* species (*C. parvum*, *C. hominis*, *C. meleagridis*, *C. ubiquitum* and *C. cuniculus*) using the 60-kDa glycoprotein gene (*gp60*). This is informative enough to show inter-species variation and has also been linked to differences in clinical disease (Adamu et al., 2014; Cama et al., 2008) and, in some cases host specificity (Wang et al., 2014a). For example, some *gp60* subtype families (IIc, IIb, IIe-III) of *C. parvum* have only been detected in humans but never in animal hosts. This is expanded on in section1.5.2.

1.5.1 gp60 Subtyping

The GP60 protein is expressed on the surface of invading stages of the parasite and is a target for neutralising antibodies (O'Connor et al., 2007). It is formed from two cell surface glycoproteins *gp15* and *gp45*; *gp45* is also referred to as *gp40* and *S45* (Strong et al., 2000; Strong and Nelson, 2000; Winter et al., 2000) and *gp15* as *cp17* and *S16* (Winter et al., 2000). The *gp60* gene is similar to a microsatellite as it contains multiple repeats of three tri-nucleotides (TCA, TCG and TCT) at the 5' end of the gene.

Sequence analysis of the *gp60* gene is commonly used for genotyping *Cryptosporidium* species and often *gp15* is used as a microsatellite marker for multilocus genotyping to identify *Cryptosporidium* subtypes (Strong et al., 2000). In addition to variation in the number of tri-nucleotide repeats in the microsatellite region of gp60 there are also extensive sequence differences in non-repeat regions of the gene. These differences enable us to define *Cryptosporidium* species into subtype families (Sulaiman et al., 2005); the name of *gp60* subtypes always starts with identification of the subtype family e.g. IIa (this is the most commonly recognised *C. parvum* subtype family in the UK) followed by the number of TCA (A), TCG (G) and TCT (T) repeats within the microsatellite region (Sulaiman et al., 2005). In some subtypes of *C. parvum*, either within or immediately after the microsatellite region there is a six basepair repeat ACATCA which is counted and represented by "R", in addition, some *C. hominis* subtypes have a 13 or 15 base-pair repeat (5' -AAGACGGTGAAGG3'/ 5' -AGGACGGTGGTAAGG-3') at the end of the microsatellite region also represented by "R". The most commonly identified *C. parvum gp60* subtype found in the UK is IIaA15G2R1 which means that the parasite belongs to *C. parvum* subtype family IIa and contains 15 copies of the trinucleotide repeat TCA, two copies of TCG, no copies of TCT and one copy of the 6 base-pair sequence ACATCA (Figure 1-3).



The main advantage to using *gp60* as a single locus marker is that extensive use of this gene worldwide means comparisons of subtypes can easily be made between studies, countries and host species. It is also possible to link *gp60* subtypes with biological characteristics and clinical disease (Xiao, 2010) as, unlike many of the other genes which are used for subtyping, it is a functional protein which is involved in host cell attachment and parasite invasion (O'Connor et al., 2007). One disadvantage of using this single locus is that, at present, it can only be used to subtype certain species (*C. parvum*, *C. hominis*, *C. meleagridis*, *C. ubiquitum* and *C. cuniculus*).

1.5.2 C. parvum gp60 Subtype Diversity

In many countries high subtype diversity of *C. parvum* has been reported from both human and ruminant studies. In Northern Ireland a study which analysed the gp60 sequence of 214 neonatal calves found 15 different subtypes all from the IIa subtype family (Thompson et al., 2007) and in the Netherlands 160 *C. parvum* positive bovine samples were subtyped by *gp60* sequence analysis and 17 subtypes of IIa and III subtype families were identified (Wielinga et al., 2008).

Subtyping of *C. parvum* and *C. hominis* with *gp60* has been used worldwide, it has been shown that some *C. parvum* subtype families (IIa and IId) are found in

both humans and ruminants, whereas other C. parvum subtypes (IIC, IIb, IIe-III) have only been identified in humans (Alves et al., 2003; Cama et al., 2008; Hijjawi et al., 2010). Even within ruminants it has been found that some C. parvum subtypes preferentially infect one ruminant species over another. In areas of Spain where both IIa and IId are found IIa predominantly infects calves while IId predominantly infects lambs and goat kids (Quilez et al., 2008a; Quilez et al., 2008b). The most common subtype of *C. parvum* is IIaA15G2R1 which has a worldwide distribution and is often found in both humans and ruminants (Broglia et al., 2008). Within C. hominis subtypes significant variations in clinical disease and virulence have also been found in certain areas of Peru and Ethiopia in patients suffering from HIV (Adamu et al., 2014; Cama et al., 2008; Cama et al., 2007). It has been previously suggested that differences in clinical manifestations may be due to the immune status of the infected person (Flanigan, 1994) but the results from these studies suggest that there are differences in the parasite in addition to host immune factors. Between species, C. parvum was likely to be the most pathogenic as it was significantly associated with chronic diarrhoea and vomiting while *C. meleagridis* was less pathogenic. Within species, those infected with C. hominis subtype family Id were most likely to suffer from diarrhoea while those infected with subtype family Ia had fewer clinical signs. It is also possible that some people have a genetic predisposition that makes them more susceptible to infection with Cryptosporidium. Fayer et al (2000) showed that children with B*15 HLA I allele and the DQB1* 0301 HLA II allele were more likely to suffer from cryptosporidiosis than those with a different allele.

Although most studies subtyping *Cryptosporidium* species are studies examining human samples or outbreaks there are more and more studies being carried out to investigate the subtypes of *Cryptosporidium* infecting farm livestock. One cross-sectional study of neonatal calves from 41 farms in the South of England demonstrated that most farms have a single predominant *gp60* subtype (Brook et al., 2009). The findings in this study tie in with other studies which show that on individual farms *C. parvum gp60* subtypes seem stable but that there is wide genetic variation between farms. A study carried out in Scotland on the population genetics of *Cryptosporidium* species by microsatellite analysis of isolates from human, cattle and sheep sources showed that genetic diversity in
bovine isolates can be linked with cattle movement (Morrison et al., 2008). Some other studies which have looked at the *C. parvum gp60* subtypes of calves have found similar results to this but another study in the UK found that IIaA15G2R1 was not the most commonly detected subtype. The study examined 779 faecal samples collected from neonatal diarrhoeic calves from farms across Northern Ireland. All of the *C. parvum* positive samples (n=213), were subtyped at the *gp60* locus, in total 16 different subtypes were identified all from allele family IIa. The most prevalent subtype in this study was IIaA18G3R1 (n=120) (Thompson et al., 2007) which has also been identified in humans from Northern Ireland (Zintl et al., 2009).

1.6 Transmission of Cryptosporidium

Cryptosporidium oocysts are transmitted between hosts via the faecal-oral route, either directly from contact with faeces of infected animals or indirectly through environmental contamination or from ingestion of contaminated food or water.

There are some reports of veterinary students, typically from urban areas, becoming infected with *C. parvum* when they start working with farm livestock (usually calves) during their studies (Gait et al., 2008; Pohjola et al., 1986; Preiser et al., 2003) as well as outbreaks amongst members of the public associated with petting zoos or farm visits (Pritchard and Fleetwood, 1995) (Gormley et al., 2011). Drinking water outbreaks can be very serious as many people can become infected in a short space of time. The oocysts of *Cryptosporidium* are very small and as such it can be difficult to eliminate them from municipal supplies as the filters used are larger than the oocysts themselves. It is also not possible to chemically inactive the oocysts (Pollock et al., 2008) by chlorination or other methods commonly used to treat drinking water. More recently in some areas the use of ozone and UV light have been very useful in inactivation of Cryptosporidium oocysts in drinking water (Cho and Yoon, 2007). The world's largest waterborne Cryptosporidium outbreak occurred in Milwaukee in 1993. During this outbreak ~400,000 people were affected. The source of the outbreak was never confirmed at the time but run-off from cattle pastures due to ice melt was one suggestion (Mac Kenzie et al., 1994). Retrospective analysis of samples collected at the time of the outbreak has

shown that patients were in fact infected with *C. hominis* rather than *C. parvum*. This indicates that the source of the infection was anthroponotic rather than zoonotic . Foodborne transmission of *Cryptosporidium* can occur when food items, such as vegetables and salad items are washed with contaminated water during processing, from contamination of food by a food handler or contact with infected animal faeces. Foodborne transmission of cryptosporidiosis seems to be much lower than waterborne transmission, this is either because there is less foodborne transmission or because of under reporting of outbreaks (Robertson and Chalmers, 2013).

Anthroponotic transmission of *C. hominis* has been associated with those working in caring environments such as nursery and care workers who may have close contact with faeces from infected individuals (Joachim, 2004). There have also been waterborne outbreaks associated with human to human transmission via drinking water (Mason et al., 2010; Widerstrom et al., 2014) and recreational waters (Coetzee et al., 2008).

1.6.1 Infective Dose

Attempts have been made to quantify the minimum infective dose of *Cryptosporidium* oocysts of different species in different host species. The results vary between study even when the same *Cryptosporidium* and host species is being examined. One study carried out on human volunteers with *C. parvum* reported an infective dose of 132 oocysts (DuPont et al., 1995). As has been previously discussed variation in the outcome of clinical disease occurs depending on the isolate used, another paper reported differences in the infective dose among *C. parvum* isolates in human volunteers. The infective dose for *C. parvum* IOWA isolate was 87 oocysts, 1042 oocysts for *C. parvum* UCP isolate and only 9 oocysts for *C. parvum* TAMU isolate (Okhuysen et al., 1999). One study in calves (infected with *C. parvum*) found that the infective dose for oocyst shedding only was 5.8 oocysts, scour only was 9.7 oocysts and for oocyst shedding with diarrhoea was 16.6 oocysts (Zambriski et al., 2013b). The infective dose is thought to be very low due to the rapid life cycle and ability of the parasite to auto-infect host cells.

1.6.2 Oocyst Shedding (calves)

Infected hosts can shed huge numbers of oocysts per day, which are immediately infective to other susceptible hosts (Nydam et al., 2001; Zambriski et al., 2013a) meaning that infection can pass very quickly between animals kept in close contact to one another. Infected calves will begin to shed infective oocysts 4-9 days after initial infection (depending on infective dose) and can shed large numbers of oocysts each day (Faubert and Litvinsky, 2000). Studies have reported oocyst shedding in calf faeces to be around $10^4 - 10^8$ oocysts per gram of faeces (Fayer et al., 1998), the number of oocyst shed in the faeces depends on the initial infective dose (Fayer et al., 2010a; Rieux et al., 2013; Zambriski et al., 2013a). Excretion of oocysts normally lasts for 6-14 days (Fayer et al., 1998; Nydam et al., 2001; Zambriski et al., 2013a) and high oocyst excretion is more usually associated with calves younger than one month (O'Handley et al., 1999; Ongerth and Stibbs, 1989). Animals infected with a higher dose will not shed any longer than those receiving a lower dose but they will shed more oocysts throughout the duration of shedding (Zambriski et al., 2013a).

1.7 Cryptosporidiosis

Not all species of *Cryptosporidium* cause similar disease manifestations. In most mammals' infection with *C. parvum* causes self-limiting watery diarrhoea, loss of appetite and abdominal pain (Clark and Sears, 1996; Klein et al., 2008; Tzipori et al., 1983; Tzipori and Ward, 2002). In other host species different *Cryptosporidium* species will cause similar clinical disease such as; *C. hominis* in humans, *C. meleagridis* and *C. galli* in birds, usually chickens; *C. serpentis* in snakes; *C. varanii* in lizards and *C. xiaoi* in goats whereas other *Cryptosporidium* species manifest themselves in other ways. For example, in birds the most common form of cryptosporidiosis is respiratory cryptosporidium appear to be non-pathogenic in some hosts such as *C. ryanae* and *C. bovis*, these are commonly found in calves but experimental infections have shown that although infected animals shed oocysts infection with these species is not associated with clinical disease (Fayer et al., 2008; Fayer et al., 2005). In some cases and in immunosupressed individuals disease caused by *C. parvum* can be severe, lasting

for several weeks, spreading to other areas of the body, as opposed to the intestines, such as the respiratory tract causing respiratory disease and may even lead to death (Denkinger et al., 2008; Hojlyng and Jensen, 1988; Moore and Frenkel, 1991; Mor et al., 2010).

Diarrhoea caused by *C. parvum* can occur in two ways; malabsorptive diarrhoea is caused by loss of enterocytes and blunting of villi (Figure 1-4) which reduces the intestinal surface leading to decreased water and nutrient absorption (Klein et al., 2008; Tzipori and Ward, 2002). Secretory diarrhoea occurs when intestinal secretion is increased. Some infectious organisms produce toxins which stimulate intestinal secretion by invading host epithelial cells (Clark and Sears, 1996) and there have been reports of patients infected with *Cryptosporidium* suffering from profuse watery diarrhoea, which is very similar to the secretory diarrhoea that is caused by the cholera (*Vibrio cholera*) toxin. A study in 1994 identified enterotoxin-like activity in stool supernatant from *C. parvum*-infected calves (Guarino et al., 1994) and others believe that *C. parvum* might simulate proinflammatory cytokines. Intestinal damage caused by infection can lead to a slight reduction in growth rates but the intestinal absorption is generally restored three weeks after infection and no long-term damage to the gut occurs (Klein et al., 2008).



Figure 1-4: Healthy gut villi (left) and damaged gut caused by *C. parvum* infection (right) in a calf.

1.7.1 Disease in farm livestock

Cryptosporidiosis in farm livestock usually lasts around two weeks in otherwise healthy animals and has no long-term implications on animal health except in some severe cases when lambs and chickens may suffer from a reduction in growth (de Graaf et al., 1999; Goodwin et al., 1996; Sweeny et al., 2011a). The economic losses associated with *Cryptosporidium* have never been calculated but losses would include cost of treatment of enteritis, reduced feed conversion and production efficiency and, of course, losses due to animal death.

The majority of animals infected with *Cryptosporidium* survive but in some host species animals which have been infected may have lower growth rates than those that have not, lambs infected with *C. parvum* were shown to have reduced growth rate, live-weight and carcass quality than those not infected (Sweeny et al., 2011b).

Despite most reports stating that cryptosporidiosis is generally self-limiting with no long-term health implications there are several anecdotal incidences of very severe cryptosporidiosis leading to high morbidity and mortality within herds and flocks with some farmers having no choice but to leave the industry due to lack of profitability.

1.7.1.1 Cattle

Cryptosporidiosis was first reported in cattle in the early 1970's (Meuten et al., 1974; Panciera et al., 1971) but the diarrhoea caused in these animals was also associated with other viral and bacterial infections and could not be solely attributed to *Cryptosporidium*. In 1983 neonatal diarrhoea in calves was reported with *Cryptosporidium* species as the sole infective agent (Tzipori et al., 1983) and it is now recognised that the disease is endemic worldwide. Infection with *C. parvum* in cattle tends to be located in the small intestine at the ileocaecal junction but the parasite has also been found in the caecum, colon and duodenum (Pohlenz et al., 1978; Sanford and Josephson, 1982). Different species of *Cryptosporidium* are found in the abomasum (Lindsay et al., 2000). Death of young calves caused by *Cryptosporidium* is usually attributed to dehydration due to severe diarrhoea (Sanford and Josephson, 1982).

Different age groups of cattle are commonly infected with distinct species of *Cryptosporidium* (Figure 1-5) although there are sometimes exceptions to this. Clinical disease caused by *Cryptosporidium* is usually limited to neonatal

livestock with older animals (>6 weeks) exhibiting only asymptomatic shedding of oocysts (Fayer et al., 2006) and is caused by *C. parvum*; whereas the species that infect older calves (*C. bovis* and *C. ryanae*) have not been associated with any clinical disease (de Graaf et al., 1999; Fayer et al., 2008; Fayer et al., 2005). *Cryptosporidium andersoni* tends to affect the abomasum of adult cattle but this species does not cause diarrhoea and much fewer oocysts are shed compared with *C. parvum* infection (Kvac and Vitovec, 2003), infection with this species tends to cause mild clinical disease and affects weight gain and milk production (Anderson, 1987; Esteban and Anderson, 1995; Ralston et al., 2010).



Species found in Cattle

Figure 1-5: Species of *Cryptosporidium* generally associated with different age groups of cattle.

It is commonly accepted that different species of cattle-adapted *Cryptosporidium* species are found in different age groups of cattle. *C. parvum* is commonly found in young calves, *C. bovis* and *C. ryanae* in older calves and *C. andersoni* in adult cattle (Santin et al., 2004).

1.7.1.2 Sheep and goats

In sheep, cryptosporidiosis was first reported in lambs less than three weeks old suffering from diarrhoea on a farm in Australia (Barker and Carbonell, 1974). As with cattle, *Cryptosporidium* is now recognised as one of the main etiological agents of lamb diarrhoea and in lambs is associated with high morbidity and mortality in infected flocks (de Graaf et al., 1999; Sari et al., 2009). Cryptosporidiosis in lambs appears to cause more mortality than in calves, this is perhaps because lambs can become dehydrated much more quickly than calves. Australia was also the first place where cryptosporidiosis of goats was reported (Mason et al., 1981), it is now considered a very important disease of goats due to the high morbidity (up to 100%) and mortality (up to 50%) in some goat herds (Johnson et al., 1999; Paraud et al., 2010; Vieira et al., 1997). In small ruminants the infection causes depression, anorexia, abdominal pain and diarrhoea. The diarrhoea is usually accompanied by the shedding of large numbers of oocysts, usually lasting three to five days but in severe cases up to two weeks. In small ruminants such as sheep and goats infection normally occurs in the jejunum and ileum but can spread to other areas in the small and large intestine (Snodgrass et al., 1984).

1.7.1.3 Poultry

In domestic poultry the most common form of cryptosporidiosis is respiratory rather than diarrhoeic except in turkeys. Birds are also known to suffer from renal and ocular cryptosporidiosis.

Cryptosporidium meleagridis infects the intestinal tract, bursa of fabricius and cloaca (Ryan, 2010) in turkeys and chickens but clinical disease seems to affect only turkeys (Slavin, 1995), in other birds asymptomatic infections have been reported (Abe and Iseki, 2004). *C. baileyi*, in addition to infection of the intestinal tract, bursa of fabricius and cloaca also infects the respiratory and urinary tract in turkeys, chickens and ducks (Lindsay et al., 1986). It is the most common cause of cryptosporidiosis in poultry, especially chickens where it normally manifests itself as respiratory disease with only occasional intestinal disease. *C. baileyi* infection has also been associated with a lower body weight in broiler chickens (Goodwin, 1996). The third avian-adapted species, *C. galli*, does not develop in either the intestines or respiratory tract but instead develops in the proventriculus, causing diarrhoea; this species has been associated with high mortality in birds (Ryan et al., 2003a).

1.7.2 Humans

At present it is known that humans can be infected with ~12 of the currently recognised species (*C. hominis, C. parvum, C. viatorum, C. muris, C. felis, C. suis, C. canis, C. meleagridis, C. ubiquitum, C. fayeri, C. bovis* and *C. cuniculus*) though some of these infections may have been transient infections and infections with some species have only been reported once (Elwin et al., 2012a). Humans are most commonly infected with *C. hominis* and *C. parvum* (Hadfield et al., 2011) and it was originally thought that *C. hominis* infected only humans; however, there are some reports of *C. hominis* in other animals; a rhesus monkey (Ye et al., 2012); a dugong (Morgan et al., 2000); Canadian geese (Zhou et al., 2004); a sheep from Australia (Ryan et al., 2005); two cattle from Scotland (Smith et al., 2005) and a pygmy goat and a Texel lamb in the UK (Giles et al., 2009).

Cryptosporidiosis in humans is prevalent worldwide in both developed and developing countries. Children and the elderly are more susceptible to disease than young adults. Children tend to acquire the infection shortly after, or during weaning (Tzipori, 2002). Infection of immunocompetent people with *Cryptosporidium* tends to cause self-limiting diarrhoea, however, a recent report showed that *Cryptosporidium* is the second biggest cause of infant diarrhoea and death in Africa and Asia (Striepen, 2013) which demonstrates that cryptosporidiosis is not always self-limiting and mild. Time from infection to clinical signs can vary depending on host variables such as age and any previous exposure and also on the parasite (age of oocysts, species/genotype); C. hominis tends to cause more severe and extra-intestinal disease than other species (Cama et al., 2007). Infection generally occurs in the small intestine but can spread through the large intestine and into the stomach. Infections concentrated in the small intestine of humans have a tendency to be more severe and result in more watery diarrhoea than infections in the distal ileum or bowel, which are often asymptomatic. Where diarrhoea lasts more than seven days serious dehydration and weight-loss may occur (Tzipori and Ward, 2002).

Cryptosporidiosis, as with many other infectious diseases, can be much more severe in immunocompromised patients such as those suffering from AIDS, undergoing chemotherapy or those that have genetic disorders compared with immunocompetent patients. One of the earliest reports of cryptosporidiosis in humans was in an immunocompromised patient in 1976 (Meisel, 1976). Reports of cryptosporidiosis in people dramatically increased during the 1980s when the HIV AIDS epidemic began as the infected patients were unable to fight the infection. People suffering from HIV AIDS who have a CD4⁺ T-cell count of less than 150/ml are very susceptible to severe persistent infection by *Cryptosporidium* with profuse and often life-threatening diarrhoea (Flanigan, 1994; Huang and Zhou, 2007). In immunocompromised patients infection is rarely confined to the small intestine and can spread to the heptobilliary tract, pancreatic duct and also the lungs (Denkinger et al., 2008; Lopez-Velez et al., 1995).

1.7.3 Immunity to disease

Despite the role of immune compromise in the development of disease in humans, the degree of adaptive immunity to infection and disease does not appear to be strong or of long duration (Okhuysen et al., 1998). Many people are infected whilst young and then again in adulthood (Current and Bick, 1989). A similar situation occurs for some primate species and broiler chickens, except that these hosts remain susceptible to infection but are protected against symptomatic disease (Current and Snyder, 1988; Miller et al., 1990). In farm livestock there is some evidence that animals will become less susceptible to disease the older they are (Harp et al., 1990; Kvac et al., 2014; Ortega-Mora and Wright, 1994) though one experimental study in calves demonstrated that it does not matter what age the calf is when it is first challenged with C. parvum the animal will shed oocysts and may have diarrhoea (Harp et al., 1990). However, if a calf has previous exposure to the parasite and is subsequently challenged then the animal will not shed oocysts (Harp et al., 1990). Similar studies in mice show that mice do not need to be exposed to the parasite to develop resistance at an older age (Harp et al., 1992; Harp et al., 1988), it has also been shown that the age at which the mouse becomes resistant to infection is also the age at which the intestinal flora is fully mature (Harp et al., 1992). It is not known if it is possible to develop immunity which is specific to a species or subtype of Cryptosporidium but which will not provide protection to infection with a different species or subtype. Most experimental trials have only tested the

development of immunity to a homologous challenge (Harp et al., 1990; Miller et al., 1990; Ortega-Mora and Wright, 1994).

1.7.3.1 Host Immune Response

Immunity to *Cryptosporidium* involves both the innate and adaptive immune response. Studies on this subject are few and the nature of the immune response is poorly understood.

The innate immune response will provide an immediate defence against the parasite invading the epithelial cells by preventing attachment. It has been shown that mucin interferes with the sporozoite's ability to attach to the cell wall (Barnes et al., 1998; Cevallos et al., 2000) and other immune cells such as natural killer (NK) cells will secrete cytokines which may have a direct effect on the parasite or signal a further immune response.

Following invasion of the host epithelial cells cytokines are produced which signal NK cells into action. When stimulated by TNF α and IL-12, NK cells produce IFN- γ (and other cytokines but mainly IFN- γ), NK cells also produce TNF β , which can directly damage target cells and cause cell lysis. TNF α activates APC (antigen presenting cells) such as dendritic cells and macrophages which are found in the villi of the intestinal mucosa. Activation of these leads to the secretion of further cytokines and interaction with CD4⁺ T cells, these cells differentiate into either Th1 or Th2 cells. Th1 cells produce IL-12, IFN- γ and TNF β which enhance the cytotoxic response, Th2 cells produce other interleukins which stimulate the differentiation of B-cells into antibody producing cells (Figure 1-6).



Figure 1-6: Simplified diagram showing the most important steps and interactions in the acquired immune response against *Cryptosporidium* infection. Adapted from *Cryptosporidium* from molecules to disease.(Thompson et al., 2003)

Much of the knowledge of immune responses to *Cryptosporidium* has been gained through infection trials in murine models. There are some studies investigating the immune responses to *Cryptosporidium* in humans but very few in livestock hosts.

Experimental infections in mice have shown that Th1 cytokines, CD4⁺ T cells and IFN-γ are essential for elimination of *Cryptosporidium* infection (McDonald and Bancroft, 1994; McDonald et al., 1996). Mice lacking CD4⁺ T cells become seriously ill if infected compared with 'normal' mice, the same is assumed for humans as it is known that AIDS patients with low CD4⁺ T cell counts suffer from more severe disease which can be fatal (Hoepelman, 1996; Lopez-Velez et al., 1995; Tali et al., 2011).

Cytokines that play a key role in both innate and adaptive immunity include IFN- γ . In mice it has been established that IFN- γ is a major player in controlling *Cryptosporidium* infection, not only in cell-mediated immunity but in early innate immune responses as well (Riggs, 2002; Theodos, 1998). Mice deficient in IFN- γ and other cytokines (IL-12 and IL-18) have increased susceptibility to

infection (Mead and You, 1998; Tessema et al., 2009). Similar results have been reported for HIV-negative patients with IFN- γ deficiency (Gomez Morales et al., 1999) and *in vitro* studies have demonstrated that in human cells IFN- γ directly prevents the parasite from invading host cells (Pollok et al., 2001).

1.7.4 Control of cryptosporidiosis

Cryptosporidiosis is a difficult disease to control, as the oocysts are very environmentally stable and can survive for long periods of time in cool, moist conditions, infected animals shed huge numbers of infective oocysts (Fayer et al., 2010a) and infection can pass through a group of susceptible hosts very quickly. The oocysts are also resistant to many disinfectants (Chalmers and Giles, 2010), there are no vaccines available to prevent the disease, and treatment options are limited and often rely on rehydration therapy.

1.7.5 Therapeutics

At present there are very few products licensed in the UK for the treatment or prevention of cryptosporidiosis in livestock or humans, the few products which are available are not very effective and in most cases will only reduce the duration of shedding and have little or no effect on immunocompromised patients.

1.7.5.1 Livestock

The only licensed treatment for cryptosporidiosis in calves is halofuginone lactate which affects invading parasite stages of *Cryptosporidium* (Jarvie et al., 2005). This drug is approved for use in both prevention and treatment of cryptosporidiosis in calves at a dose of 0.10 mg/kg of body weight per day for seven consecutive days. As a preventative measure the drug should be given within 48 h of birth and as a therapeutic, within 24 h of the onset of symptoms. Treatment with halofuginone lactate does not completely prevent or cure disease but it does reduce oocyst shedding and duration of diarrhoea (Jarvie et al., 2005; Lefay et al., 2001; Trotz-Williams et al., 2011). There is no licensed treatment for cryptosporidiosis in sheep, goats or pigs. Several other chemotherapeutic agents have been tested for the treatment of cryptosporidiosis in livestock but none have resulted in a significant reduction in oocyst shedding; some antibiotics have shown efficacy against *Cryptosporidium* oocyst shedding, clinical disease and mortality in calves, lambs and goat kids but these are not registered for use in calves (Fayer and Ellis, 1993; Viu et al., 2000). A few coccidiostats, such as decoquinate have been tested against *Cryptosporidium* with limited or no reduction in oocyst shedding (Moore et al., 2003).

1.7.5.2 Humans

For human cases of cryptosporidiosis the only treatment available is nitazoxanide, which interferes with anaerobic metabolism of the parasite. Like halofuginone lactate in the treatment of cryptosporidiosis of livestock this drug cannot prevent or cure the disease but does reduce shedding of oocysts and severity of disease in immunocompetent patients (Rossignol et al., 2001). It has little effect on the severity of disease in immunocompromised patients (Amadi et al., 2002).

1.7.6 Vaccines

At present there are no vaccines available to prevent cryptosporidiosis in either farm livestock or humans. However, several attempts to develop such a vaccine have been made, some of which were partially successful under experimental conditions. Calves that were immunised with killed (gamma-irradiated or lyophilised) *C. parvum* oocysts showed a reduction in oocyst shedding and levels of diarrhoea compared to non-immunised calves (Harp and Goff, 1995; Jenkins et al., 2004). However, this vaccine was not effective when tested under field conditions (Harp and Goff, 1998).

Infection with *Cryptosporidium* often occurs within the first week of life so attempting to immunise the calves themselves is probably ineffective as this will not mediate a significant immune response prior to infection (de Graaf et al., 1999; Innes et al., 2011). Some attempts to immunise cows so that they produce antibodies against *Cryptosporidium* which can be passed to their calves in the colostrum have been made. Calves receiving colostrum from cows vaccinated

with a recombinant *C. parvum* were protected against diarrhoea and also a reduction in oocyst shedding compared to the calves which did not receive colostrum from vaccinated cows (Perryman et al., 1999).

1.7.7 Disinfectants

The oocysts are highly resistant to commonly used disinfectants (Virkon® S (peroxygen compounds (DuPont) (Ares-Mazas et al., 1997), FAM® 30 (lodophors (Evans Vanodine)), Sorgene® 5 (hydrogen peroxide (BASF)) at their recommended working concentrations, (reviewed by Casemore & Watkins (, 1998)), making them difficult to eliminate in cattle areas on farms.Some less common or harder to obtain disinfectants (Keno[™]Cox (amine based (CID Lines)), (Ox-Virin & Ox-Agua (peroxygen compounds (Edificio OX)) have been shown to be effective against *Cryptosporidium* oocysts, at least *in vitro* (Castro-Hermida et al., 2006b; Naciri et al., 2011; Quilez et al., 2005). There are very few field trials testing these disinfectants but one such study demonstrated that even though these may be effective *in vitro* they are not effective enough to rely on for the effective elimination of *Cryptosporidium* oocysts on farms (Keidel and Daugschies, 2013).

1.7.8 Farm management

Most recommendations for controlling cryptosporidiosis rely on good management and hygiene practices as the oocysts are so difficult to eliminate it is better to try and reduce the environmental contamination in the first instance. Frequent mucking out of calving areas and calf houses combined with steam-cleaning and disinfection can help to reduce environmental build up (Castro-Hermida et al., 2002) as can thorough cleaning with hot water followed by drying (Harp and Goff, 1998) as the oocysts are susceptible to extremes of temperature and desiccation (Robertson et al., 1992).

Other calf management practices such as housing can have an effect on infection rates but some of the results are conflicting. In one study it was shown that housing calves with their dams decreased the risk of *C. parvum* infection while calves removed from their dam and kept in individual pens were at higher risk of infection (Duranti et al., 2009). In contrast, other studies have shown that the

opposite is true (Quigley, III et al., 1994). It is also possible for infection to be passed between subsequent calves in a particular pen (O'Handley et al., 1999) if the pen is not thoroughly cleaned between calves. Calves housed outdoors are at less risk than those housed indoors regardless of the housing arrangements (Quigley, III et al., 1994).

1.8 Summary

In summary, cryptosporidiosis is a very serious cause of enteritis and disease in many vertebrate and invertebrate species. Detailed, longitudinal studies in farm livestock are few and since the development of new molecular tools the dynamics of parasite infection should be further investigated. There is a lack of diagnostic tools to enable the quick speciation of *Cryptosporidium* present in calf samples; it is also difficult to detect mixed infections meaning that zoonotic infections can be missed where the predominant species is a less-pathogenic one. In particular, the transmission of *C. parvum* from adult cattle to neonatal calves is much debated and the ability of animals to develop immunity to disease is not well understood. This PhD work hopes to answer, in part, some of these important questions.

1.9 Aims

The main aims of this PhD are:

- 1. To develop new typing tools (multiplex PCR) to distinguish *Cryptosporidium* species commonly found in cattle samples
- 2. To assess the shedding profile of *Cryptosporidium* species and genotypes found in calves on a study farm
- 3. To assess the role of adult cattle in the transmission and persistence of *Cryptosporidium parvum* on a study farm
- 4. To assess the host-pathogen interactions involved in animals given serial experimental challenges with distinct *Cryptosporidium parvum* isolates

Chapter 2: Materials and Methods

The methods described here are general protocols used throughout this PhD in various studies. Animal and sample numbers are described in more detail in the relevant chapter as well as any adaptations or changes to the general protocol.

2.1 Cattle Sample Collection

2.1.1 Dairy Calves

Faecal samples from dairy calves were collected directly from the rectum using a gloved finger and bag for collection; where possible the entire motion was collected. It was not always possible to collect a sample from each calf on each sampling day; therefore the number of samples collected per calf did vary. After collection the samples were transferred to the laboratory where a sub-sample was aliquoted into a bijoux tube and stored at -20°C.

2.1.2 Adult Dairy Cattle

Collection of faeces from adult dairy cattle was carried out by observing the cow until she defecated and picking the sample from the floor; the whole pat was collected where possible. The cows were housed in cubicle sheds before calving, in straw courts at calving and in slatted cubicle sheds after calving when they returned to the milking herd. After collection the samples were transferred to the laboratory where the entire motion was mixed and a sub-sample aliquoted into a 125 ml tub and stored at 4°C.

2.2 Experimental Animals

All experimental animals were used in accordance with Home Office regulations, with approval of the experimental design sought before the study commenced from the Moredun Research Institute's ethics committee.

2.2.1 Lambs

Forty newborn (<24 h old) Texel and Texel cross male lambs were obtained from Firth Mains farm, Roslin.

2.2.1.1 Pre-amplification of Cryptosporidium isolates

Four of the 40 lambs were obtained one month prior to the study start date. These lambs were separated into two groups of two and each group was inoculated with one of two different *C. parvum* isolates (see section 3.3.1).

2.2.1.2 Study animals

Animals used in the study were separated into six groups of six depending on experimental challenge (see section 3.3.2).

2.2.2 Clinical Scoring

Lambs were monitored a minimum of three times per day for the first six weeks and then a minimum of twice daily for the remainder of the study. Daily monitoring involved the recording of the volume of milk each lamb drank along with how well the animal drank, the demeanour of the animal and the appearance of the faeces produced (Table 2-1).

Table 2-1: Scoring system used to score lambs daily

	Demeanour Rising	Demeanour Standing	Feeding	Faecal Consistency
Score	Description	Description	Description	Description
1	Already on feet	-	Feeding vigorously	Solid Pellet (s)
	Happy to rise	Fit and well	Complete feed with interruptions (due to excitement) Complete feed with assistance (due to age)	
2	Reluctant to rise	Listless	Partial feed with assistance (due to age)	Soft Pellet (s)
3	Needs assistance to rise	Unsteady on feet	Complete feed with interruptions (due to illness)	Soft Mass
4	N/A	Cannot remain standing	Complete feed with assistance (due to illness)	Scour
5	N/A	N/A	Partial feed – no assistance	Liquid
6	N/A	N/A	Partial feed with assistance (due to illness)	N/A
7	N/A	N/A	Animal refuses to feed	N/A

2.2.3 Blood Sampling

Blood samples were taken weekly from each lamb by jugular vein puncture using a vacutainer needle (BD Vacutainer, USA). Blood samples were left overnight at 4° C to clot then centrifuged at 2000 × g for 10 mins and the serum transferred to a 1.5 ml tube. Serum samples were stored at -20°C.

2.2.4 Faecal Collection

Total faecal output was collected from each lamb using a specially designed harness and bag system (see section 2.2.4.1) which the lamb wore continuously. This enabled the lambs to be penned in groups rather than separately but still allowed collection of samples from each lamb individually. This system also ensured that after inoculation the lambs were not exposed to any faecal matter and were not re-infected by oocysts shed in the faeces. Bags were taken off and replaced with a new one clearly marked with group/animal/sample when required. Samples were stored in the bags at 4°C until they were processed (see section 2.5.2).

2.2.4.1 Harness and bag design

An adjustable body harness with attachment points for the collection bag was designed and made using 20 mm webbing and plastic buckles (Figure 2-1). Collection bags were made using resealable polythene bags with a pipe cleaner to keep the bag open, a hole for the tail and a length of India cotton tape to attach the bag to the harness.



Figure 2-1: Lambs wearing harness and bags for collection of faeces.

2.3 Cryptosporidium oocysts

In the experimental study (Chapter 3) three different *C. parvum* isolates were used, these are described in more detail in section 3.3.3.

2.3.1 Collection of faeces for oocyst purification

Faecal samples were collected from calves on three different farms known to have a problem with *Cryptosporidium* (history of cryptosporidiosis diagnosed by a veterinarian and regular occurrence of scour in neonatal calves). Samples were collected using the method described in section 2.1.1 from 2-6 scouring calves on each farm.

2.3.2 Identification, quantification, extraction and purification of positive samples

2.3.2.1 Identification and quantification

Each faecal sample was weighed and the starting weight recorded. If the sample was not sufficiently liquid (easy to pipette through a 1000 μ l pipette tip) then water was added and the sample re-weighed. Approximately 1 ml of the faecal suspension was added to a bijoux, weighed, and diluted 1:5 with dH₂O. The sample was then vortexed vigorously and 100 μ l added to 900 μ l malachite green stain (0.16% malachite green, 1% SDS). Using a haemocytometer 10 μ l of the stained faecal suspension was examined under the microscope for the presence of oocysts. The number of oocysts in each of the four, sixteen grid corner squares were counted (Figure 2-2) on both side of the haemocytometer and the average number of oocysts per gram of faeces.



Figure 2-2: Example of the grid pattern on a haemocytometer. The number of oocysts in each of the four sixteen grid squares (circled)

Quantification of the oocysts in each sample was done as follows (mean) \times (dilution factor) \times (10⁴) \times (sample weight).

Oocysts were extracted from positive samples with high oocyst numbers using the method described in section 2.3.2.2.

2.3.2.2 Extraction and purification of oocysts from positive samples

Oocysts were extracted from faeces that were found to be positive for *Cryptosporidium* when tested as described in section 2.3.2.1 using a sucrose flotation method. Sixty millilitres of faecal suspension was added to a 250 ml centrifuge bottle with 90 ml 45% sucrose solution and mixed thoroughly before being centrifuged at $1000 \times g$ for 20 mins. The meniscus was removed and diluted 1:5 with dH₂O and centrifuged at 1000 \times g for 20 mins. The supernatant was discarded and the pellet resuspended in 20 ml dH₂O in a 30 ml universal tube, this was centrifuged for 5 mins at $500 \times g$, the supernatant discarded and the pellet resuspended in 20 ml dH_2O . This step was repeated until the supernatant was clear (four washes on average). After the final wash the pellet was resuspended in 10 ml 10% sodium dodecyl sulphate (SDS) and left to stand at room temperature for 1 h. After 1 h the samples were centrifuged at $500 \times g$ for 5 mins, the supernatant discarded and the pellet resuspended in 10 ml dH_2O the centrifugation and resuspension step were repeated. The samples were stood for 10-15 mins to allow any debris to flocculate and the supernatant removed to a clean universal tube. If any debris still remained after this step 100 μ l 2% sulphuric acid was added to aid flocculation.

Once oocysts were extracted from the faeces they were counted using the method described in section 2.3.2.1 except 10 μ l oocyst suspension was added to 990 μ l 0.16% malachite green, 10% SDS.

2.3.3 Viability of oocysts and inoculum preparation

In order to test if the oocysts extracted from the calf faeces were infective a excystation assay was performed. A certain percentage of oocysts must be viable for the challenge dose to be effective.

2.3.3.1 Excystation assay

Forty microlitres of oocyst suspension was transferred to a 0.5 ml mini microcentrifuge tube and centrifuged at 12,500 × g for 30 s to pellet the oocysts. The supernatant was discarded and 40 μ l of 1x Hanks Buffered Saline Solution (HBSS) was added and the pellet resuspended, 50 μ l 1x trypsin (prepared in HBSS at pH3) was added and the sample mixed and incubated at 37°C for 1 h. Following incubation the sample was centrifuged at $500 \times g$ for 30 s and the supernatant discarded, the pellet was resuspended in 90 µl 1 × HBSS and 10 µl 2.2% sodium bicarbonate and 10 µl 1% sodium deoxycholate (in HBSS) added. The sample was mixed and incubated at 37°C for 40 mins. Following incubation the sample was centrifuged at 12,500 ×g for 30 s and the supernatant discarded. The pellet was resuspended in 50 µl 3% gluteraldehyde to fix the sample. To assess the viability of the oocysts 10 µl fixed sample was spotted onto a microscope slide a cover slip placed on top and examined under × 40 phase contrast magnification. The total numbers of oocysts, sporozoites and empty oocyst shells were counted until the combined total reached 250. The sporozoite per shell ratio was calculated as follows:

 $\frac{\text{Sporozoite Count}}{\text{Shell Count}} = \text{Sporozoites per Shell}$

And the excystation percentage as follows:

 $\frac{\text{Shell Count}}{\text{Oocyst Count} + \text{Shell Count}} \times 100 = \text{Percentage Excystation}$

Each oocyst can contain a maximum of 4 sporozoites. A sporozoite per shell ratio of >2 and an excystation percentage of >85% are considered good indicators of viability. Figure 2-3 shows excysted oocysts and sporozoites.



Figure 2-3: Excysted oocysts. Empty oocyst shells (dark circles) and sporozoites (bright "commas") can be seen.

2.3.3.2 Preparation of inoculum

Cryptosporidium oocysts were surface sterilised using 2 ml 70% EtOH (ethanol) for 30 mins before being resuspended in phosphate buffered saline (PBS). Oocysts were then counted and the oocyst suspension diluted with PBS to give the desired infective dose.

2.4 Sample Processing

2.4.1 Adult Dairy Cattle

A 50 g sub-sample of faeces was thoroughly mixed and added to a 1000 ml cylinder containing 600 ml water and 7 ml 2% sulphuric acid. The volume was then made up to 700 ml by the addition of water. The samples were mixed for 5 mins on a magnetic stirrer and then left to settle for 5 mins. The heavy, fibrous material sank to the bottom of the cylinder while the oocysts were suspended in the liquid fraction. Using a 25 ml pipette tip the supernatant was removed and added to two 250 ml centrifuge bottles, the bottles were then centrifuged at 1000 × g for 20 mins and the supernatant poured off. The thin layer of sediment at the bottom of the centrifuge bottle was resuspended in 7 ml dH₂O and poured into a 15 ml tube; two 3.5ml washes were used to remove the whole layer. The 15 ml tubes were centrifuged at 5000 ×g for 5 mins and the supernatant poured off. The pellet was then subjected to 8 ml saturated salt solution and

thoroughly mixed by vortexing then 2 ml dH₂O trickled on top of the salt solution before centrifugation at 1000 × g for 8 mins. Following centrifugation the water layer was gently swirled using a Pasteur pipette to create a vortex which draws the oocysts from the layer between the dH₂O and salt into the dH₂O, the layer was removed and added to 6 ml dH₂O, the total volume was made up to 10 ml, mixed by inverting and centrifuged at 5000 × g for 5 mins. The supernatant was poured off and the pellet resuspended in TE buffer (10 mM Tris-HCl, 0.5 mM EDTA) for DNA extraction.

2.4.2 Experimental Lambs

Samples collected from the lambs used in the experimental challenge (Chapter 3) were processed individually. The sample was weighed and the weight recorded, water was then added to the sample to make the faeces liquid enough to pipette. If the sample was already liquid then no additional water was added. The sample was then re-weighed and the new weight recorded. Using a food processor the sample was homogenised completely and a 2 ml sample aliquoted into a tube for storage for subsequent use. These aliquots were stored at -20°C. A small subsample ~850 µl was taken and used to quantify the oocysts present in the sample (see section 2.3.2.1). The remainder of the faecal material was discarded.

2.5 DNA Extraction

2.5.1 Adult Cattle Faecal Samples

DNA was extracted using a Macherey-Nagel Tissue Kit (Macherey-Nagel, Germany) following the manufacturer's instructions except $10 \times$ freeze-thaw cycles in liquid nitrogen were added. Briefly, the resuspended pellet (see section 2.4.1) was mixed vigorously using a vortex then centrifuged at 4,000 × g for 15 mins. The supernatant was poured off and the pellet resuspended in 200 µl buffer T1, $10 \times$ freeze-thaw cycles in liquid nitrogen were performed and 25 µl proteinase K (provided in the kit) added. The samples were then incubated at 56°C overnight. Following incubation the samples were briefly mixed using a vortex and then 200 µl lysis buffer B3 was added and incubated at 70°C for 10 mins, after incubation samples still containing insoluble particles were centrifuged at 11,000 × g for 5 mins and the supernatant transferred to a new 1.5 ml tube. The samples were mixed by vortexing and the binding conditions adjusted by adding 210 µl EtOH (100%). The whole sample was loaded into a spin column placed in a 2 ml collection tube, the samples were centrifuged at 11,000 × g for 1 min, the flow through discarded and the spin column placed into a new 2 ml collection tube. The silica membrane was washed twice, first using 500 µl wash buffer BW and centrifugation for 1 min at 11,000 × g and secondly using 600 µl wash buffer B5 followed by centrifugation for 1 min at 11,000 × g. After the initial wash the spin column was placed in a new 2 ml collection tube and after the second wash the flow through was discarded and the samples centrifuged at 11,000 × g for 1 min to dry the silica membrane. The DNA was eluted in 100 µl dH₂O and stored at -20°C.

2.5.2 Calf and Lamb Faecal Samples

DNA was extracted from calf and lamb faecal samples in much the same way as described for adult samples (section 2.5.1) except 200 μ g (or 200 μ l if liquid) of faecal material was added to TE buffer and no freeze-thaw cycles were performed as previous work in the laboratory had shown that free-thawing of faeces from calves or lambs did not improve the detection of *Cryptosporidium*.

2.6 Polymerase Chain Reaction (PCR)

2.6.1 Cryptosporidium Specific 18S rRNA PCR

A previously described nested PCR protocol (Xiao et al., 1999) was used to amplify a ~840 bp fragment of the 18S rRNA gene. Each 25 µl reaction contained 10 × PCR buffer (45 mM Tris-HCl pH 8.8, 11 mM (NH₄)₂SO₄, 4.5 mM MgCl₂, 4.4 µM EDTA, 113 µg ml-1 BSA, 1 mM each of the four deoxyribonucleotide triphosphates) (MacLeod et al., 1999), 0.5 units BioTaq (Bioline, UK), 10 µM of forward and reverse primers (Table 2-2) and 3 µl DNA in the primary round and 1 µl primary PCR product in the secondary round. The total volume was made up to 25 µl with dH₂O. In each PCR run one set of positive control DNA and negative controls consisting of dH₂O were included. All PCR reactions were carried out in triplicate.

Primer	Sequence 5'-3'	Fragment Size	
		(bp)	
AL1687(EF)	TTCTAGAGCTAATACATGCG	1370	
AL1691(ER)	CCCATTTCTTCGAAACAGGA	1010	
AL1598 (IF)	GGAAGGGTTGTATTTATTAGATAAAG	840	
AL3032 (IR)	AAGGAGTAAGGAACAACCTCCA		

Table 2-2: Primer sequences and size of amplicon when used with primer pair for the 18S rRNA gene.

Cycling conditions were 3 mins at 94°C, followed by 35 cycles of 45 seconds at 94°C, 45 seconds at 55°C and 1 min at 72°C. The final extension was 7 mins at 72°C. Secondary amplification products (~3 μ l) were visualised following electrophoresis on a 1.5% agarose gel stained with GelRedTM (Biotium, UK) on an Alphalmager 2000.

2.6.2 Nested Species Specific Multiplex PCR (nssm-PCR)

For the differentiation of *Cryptosporidium* species, the primary PCR products from the 18S rRNA nested PCR were amplified using a nested species specific multiplex PCR (nssm-PCR) (Chapter 4) which can distinguish the four most commonly detected bovine species of *Cryptosporidium*. The primary PCR products were diluted with 50 μ l dH₂O and 1 μ l of the dilution used as template in the secondary round. Secondary PCR reactions contained 2.5 μ l 10 × PCR buffer (section 2.6.1), 0.5 units BioTaq (Bioline, UK), 10 μ M of each primer (

Table 2-3) and 1 µl template. Cycling conditions were 3 mins at 94°C, followed by 35 cycles of 45 seconds at 94°C, 45 seconds at 56°C and 1 min at 72°C. The final extension was 7 mins at 72°C. Secondary PCR products (~2 µl) were visualised following electrophoresis on a 1.5% agarose gel stained with GelRedTM (Biotium, UK) on an AlphaImager 2000.

Primer	Sequence 5'-3'	Fragment Size (bp)
AL1598 (IF)	GGAAGGGTTGTATTTATTAGATAAAG	840
AL3032 (IR)	AAGGAGTAAGGAACAACCTCCA	
CaF	GCAAATTACCCAATCCTGAC	625
CrF	TGTTAATTTTTATATACAAT <u>R</u> CTACGG	415
CphF	AGAGTGCTTAAAGCAGGCATA	241
CbF	CTTCTTATTCCTTCTAGAATAAAAATG	305

Table 2-3: Primer sequences and size of amplicon for nssm-PCR.

The *C. ryanae* specific primer contains a degenerate base which is shown in bold and underlined. (CaF: *C. andersoni* specific forward primer, C.rF: *C. ryanae* specific forward primer, C.phF: *C. parvum* specific forward primer, CbF: *C. bovis* specific forward primer.

2.6.3 gp60 Subtyping

A nested PCR designed to amplify a 450 bp fragment of the gp60 gene (Brook et al., 2009) was used on a selection of *C. parvum* positive samples to identify the subtype. Each 25 μ l reaction contained 10 × PCR buffer (section 2.6.1), 0.5 units BioTaq (Bioline, UK), 10 μ M of forward and reverse primers (Table 2-4) and 3 μ l DNA in the primary round and 1 μ l primary PCR product in the secondary round. Cycling conditions were 3 mins at 94°C, followed by 35 cycles of 45 seconds at 94°C, 45 seconds at 55°C and 1 min at 72°C. The final extension was 7 mins at 72°C. Secondary PCR products (~2 μ l) were visualised following electrophoresis on a 1.5% agarose gel stained with GelRedTM (Biotium, UK) on an AlphaImager 2000.

Table 2 4. Thinki sequences and ampricen sizes for gpeer en				
Primer	Sequence 5'-3'	Fragment Size (bp)		
GP60 1F (EF)	ATAGTCTCCGCTGTATC	480		
GP60 1R (ER)	GAGATATATCTTGGTGCG	400		
GP60 2F (IF)	TCCGCTGTATTCTCAGCC	~375		
GP60 2R (IR)	CGAACCACATTACAAATGAAG	010		

 Table 2-4: Primer sequences and amplicon sizes for gp60 PCR

2.7 Sequencing

2.7.1 Purification of PCR products

PCR products which were sent for sequencing were purified using the Promega Wizard SV gel and PCR Clean-Up System (Promega, UK) following the manufacturer's instructions. Briefly, an equal volume of membrane-binding solution was added to the PCR in a 1.5 ml tube before being added to the minicolumn assembly. The solution was incubated at room temperature for 1 min then centrifuged at 16,000 \times g for 1 min. The supernatant was discarded and the minicolumn reinserted into the collection tube. Seven hundred microlitres of membrane wash solution was discarded and the assembly centrifuged at 16,000 \times g for 1 min. The supernatant was discarded and the wash step repeated with 500 µl membrane wash solution and centrifugation at 16,000 \times g for 5 mins. The supernatant was discarded and the minicolumn centrifuged for a further min to remove all residual ethanol. The minicolumn was inserted into a clean 1.5 ml tube and the DNA eluted in 50 µl dH₂O.

2.7.2 DNA quantification

DNA concentration of purified PCR products was quantified using a NanoDrop ND1000 spectrophotometer (Labtech International Ltd, UK).

2.7.3 Sequencing

Sequencing of PCR products was carried out by GATC (GATC, Germany) using their "Light-run" service. Five microlitres of purified PCR product was sent at a concentration of between 20-80 ng/ μ l with 5 μ l of the appropriate primer at a concentration of 5 pmol/ μ l. In some circumstances, instead of the light-run service, unpurified PCR products were sent to GATC for sequencing using their "Supremerun96" sequencing option which includes PCR purification. With this service 20 μ l PCR product was sent with a separate 1.5 ml tube containing 50 μ l of the appropriate primer at a concentration of 10 pmol/ μ l.

2.7.4 Sequence analysis

Sequences were analysed using BioEdit Sequence Alignment Software (www.mbio.ncsu.edu/bioedit/bioedit.html) and Chromas Lite (www.technelysium.com).

2.8 Statistical Analysis

Statistical analyses were performed using Minitab17 (Version 17.1.0.0, Minitab Inc.) P≤0.05 was considered significant. Survival curves were produced using R statistical package.

For association between binary outcome variables (Chapter 3), such as the presence/absence of diarrhoea, chi squared tests were used for univariate analysis and logistic regression for multivariate analysis. Continuous outcomes, such as number of days on which diarrhoea occurred, were tested for association with the explanatory variables using ANOVA. For the outcome variable of first day of diarrhoea, data was right censored (i.e. no values were recorded if the lamb did not develop diarrhoea) and Kaplan-Meier survival curves were analysed.

Chapter 3: Can infection with one isolate of *Cryptosporidium parvum* protect against infection with another?

3.1 Introduction

An experimental challenge in neonatal lambs was carried out to determine if lambs are able to develop immunity to infection with *C. parvum* following a primary infection. We also investigated the effect of an increase in age on susceptibility to disease when older naïve lambs were challenged with *C. parvum*.

There are very few studies which have investigated the effect of sequential infection with distinct strains of *C. parvum* on neonatal livestock. Two studies, one in sheep (Ortega-Mora and Wright, 1994) and one in cattle (Harp et al., 1990) have studied the effect of age on infection with Cryptosporidium species (believed at the time to be C. parvum). In the study in sheep, they noted an age-related susceptibility to infection with younger animals (<4 weeks of age) shedding larger numbers of oocysts for a more prolonged period compared with those infected a few weeks later. Both of these studies showed that animals exposed to the parasite at a younger age do not have such severe disease when exposed a second time showing that the animals were able to develop immunity to a homologous challenge. Most of the studies done in animals investigating immune responses to infection were carried out over 10 years ago before the discovery of some of the different species of *Cryptosporidium* which are now known to infect farm livestock. At this time there was not the knowledge of different isolates of C. parvum that we have now. It is not known if infection with one isolate of *C. parvum* will provide protection against infection with another isolate. It is possible that if one isolate does provide protection against another that there is potential for development of control measures using a less virulent (if one is identified) isolate to protect against infection from severe disease caused by a more virulent subtype. This strategy has been used previously to protect against other protozoan parasites, for example, the vaccine Toxovax® (MSD, Animal Health) uses a less virulent strain (S48) of the Toxoplasma gondii parasite to provide protection to pregnant ewes from more virulent strains of T. gondii.

Similar studies have been carried out in other host species with different *Cryptosporidium* species. Kvac et al (2014) investigated the age-related susceptibility of pigs to *C. scrofarum*; previous work had indicated that while *C. suis* (the other pig adapted *Cryptosporidium* species) infected all age groups of pigs it appeared that *C. scrofarum* only infected older animals. In their study the authors infected groups of pigs ranging from 4 - 8 weeks of age with *C. scrofarum* oocysts and tested all faecal samples produced from each animal. The presence of *Cryptosporidium* oocysts was confirmed using aniline-carbol-methyl-violet (ACMV) staining and 18S nested PCR.

The results of the study showed that *C. scrofarum* was not detected in the faeces of 4 week old pigs while in pigs aged 5-8 weeks, oocyst shedding was detectable by both microscopy and PCR from 6-8 DPI. The authors suggested that the animals became more susceptible to invading sporozoites at weaning when the gut was adapting to different food.

Another study investigating *Cryptosporidium* infection in chickens demonstrated an age-related susceptibility to infection (Hatkin et al., 1993). Chicks infected with *C. baileyi* at 1 day of age showed severe clinical disease (diarrhoea, respiratory distress and lesions detected by necropsy) and 28.7% (n=33) of the infected birds died. Naïve birds infected at 7 days of age had less severe clinical disease with a longer prepatent period and naïve birds infected at 14 days of age showed only asymptomatic oocyst shedding.

Some studies investigating other protozoan parasites have carried out similar trials and one such trial showed that, similarly to the age-related susceptibility of pigs to *C. scrofarum*, rabbits older than 19 days were susceptible to *Eimeria intestinalis* whereas younger rabbits (up to 16 days of age) were resistant to infection (Pakandl and Hlaskova, 2007).

In piglets, coccidiosis caused by *Isospora suis* is responsible for up to 20% of cases of piglet diarrhoea (Lindsay et al., 1985). One study carried out by Koudela (1999) examined the effects of age and sequential infection with *I. suis* in piglets.

Naïve piglets infected at 3 days of age showed clinical signs of disease (diarrhoea) at 2-4 DPI which lasted until 11 DPI following the primary challenge. Piglets which were re-challenged showed less severe clinical signs (less diarrhoea) at 5 d post-secondary challenge, and the piglets which were infected for the first time at 19 days of age showed similar clinical signs to piglets which were re-challenged. The authors suggested that this demonstrated that age has more of an effect on development of resistance to *I. suis* than acquired immunity by previous exposure.

The trial described in this chapter aims to answer the question of whether lambs are able to develop immunity following infection with *Cryptosporidium parvum* and if this will protect them against a homologous or heterologous infection with *C. parvum*. The effects of increasing age on development of resistance to disease will also be explored. The possibility of differences in isolate pathogenicity/virulence will also be examined.

3.2 Aims

- Determine whether lambs develop immunity following a primary infection with *C. parvum* to a homologous or heterologous challenge
- 2) Determine the effects of an increase in age on the resistance or susceptibility to infection by *C. parvum*
- 3) Determine if there are differences in the severity of disease caused by distinct *C. parvum* isolates

3.3 Materials and Methods

3.3.1 Animals

Forty Texel and Texel cross male lambs were obtained from Firth Mains farm, Roslin at <24 h of age and were housed in groups and kept *Cryptosporidium*-free until challenge.

3.3.2 Experimental Design

A total of 40 lambs were used in the experiment and were separated into eight groups. Four lambs were used for pre-amplification of two different *C. parvum* isolates to produce oocysts for inoculation, these lambs were separated into two groups each containing two lambs (Table 3-1).

Table 3-1: Pre-amplification of *C. parvum* isolates.

Group	C. parvum Isolate	Challenge	Number of Animals
Group A	BB	1 × 10⁵	n=2
Group B	LH	1 × 10⁵	n=2

The remaining 36 lambs were used for the experimental trial and were separated into six groups each containing six lambs. Group I and II were agerelated susceptibility challenge groups (not challenged until week 6), Group III and VI were homologous challenge groups and Group IV and V were heterologous challenge groups (Figure 3-1).


Figure 3-1: Experimental design.

Lambs were infected with two different *C. parvum* isolates at either 6 weeks of age only or at <1 week of age and re-infected with a homologous or heterologous challenge at 6 weeks of age.

3.3.3 Cryptosporidium parvum isolates

C. parvum isolates were obtained from calves from three dairy farms (Farm OV, Farm BB & Farm LH) in Scotland with known histories of cryptosporidiosis and assessed for suitability for use in the trial. Isolates from two farms (Farm BB & Farm LH) were chosen for use in the experimental trial.

The isolates used in the experimental trial were obtained from naturally infected calves on two dairy farms; one in Midlothian (also the study farm used in the studies described in Chapter 5 & 6 (Farm LH), Scotland and the other in the Scottish Borders (Farm BB), Scotland. In general the farm management practices on both farms was very similar. Farm LH milked approximately 240 dairy cows (Holstein Friesian and Norwegian Red) and farm BB milked approximately 500 dairy cows (Holstein Friesian). Both used artificial insemination (AI) to impregnate their cows, any cows which did not hold to AI were serviced naturally by the stock bull, on farm LH only heifers not holding to AI were serviced naturally. On both farms the only animals bought in was a new stock bull every 3-4 years.

For prevention of enteritis, in-calf cows and heifers were vaccinated with Rotavec[™] Corona vaccine (MSD Animal Health, UK) in accordance with the manufacturers recommendations.

In-calf cows and heifers were removed from the main herd before calving and are housed in cubicle sheds with automatic scrapers. Approximately 1 week prior to calving the cows and heifers were moved to a straw bedding court, the pens were regularly bedded (-3 × per week) with fresh straw and are mucked out and steam cleaned annually. On farm LH once the cows calved the calf was kept with its dam for a minimum of 24 h and was assumed to have received adequate colostrum if the calf is seen to suckle. On farm BB the stockworker was alerted to all newborn calves within 1 h of birth and they ensured that each calf received an adequate volume of colostrum in the 2 h following birth. Farm BB had a frozen supply of colostrum with 2 litres ready at anytime to give to a calf if the dam did not have a sufficient supply or if the calf was born to a heifer. Only cows colostrum is used to feed newborn calves on farm BB as it is higher in antibodies (Morrill et al., 2012) than colostrum from heifers and the stockworker believes that this is beneficial to the calves.

Following birth, calves on farm LH were kept in small groups (3-5 calves) for 5 days and were bottle fed calf milk replacer before being moved to a bigger pen of ~20 calves. The calves were housed in a large straw bedded pen in a well ventilated barn and had free access to hay, water and pelleted calf feed. The calves were fed milk replacer through an automated system which dispenses a set volume of milk for each individual calf. The volume of milk is set by the stockworker. The calves were weaned at 3 months and were then moved to a larger indoor straw bedded pen of ~40 calves until they reach 6 months of age. At this point, depending on the time of year they were either turned out to pasture or moved to a cubicle shed until turn out. The straw courts were regularly bedded (~3 × per week) with fresh straw and were mucked out, steam cleaned and disinfected between calf batches. Each year when the calves were turned out the calf pens were thoroughly cleaned, disinfected and painted.

Calves on Farm BB suffering from mild or asymptomatic infection whilst still shedding large numbers of oocysts Calves on farm BB were kept in individual calf pens for 2 days following birth and, depending on the condition of the calf and the weather conditions the calves may have had a calf jacket (Cosy Calf, UK) put on. The calf wore the jacket until the temperature was warm enough to remove the jacket. Whilst in individual pens the calves were bottle fed milk replacer. The individual pens were mucked out and disinfected between calves.

After 2 days the calves were removed from the individual pen and were housed in a straw bedded pen in a large well ventilated barn with approximately 15 other calves of a similar age group (<1 month). The calves were fed milk replacer through an automated system which dispenses a set volume of milk for each individual calf. The volume of milk was set by the stockworker. They also had free access to hay, water and pelleted calf feed. Once the calves were old enough they were moved into an adjacent pen with other older calves until weaning at 3 months. At weaning calves were either turned out or moved to a shed until turn out, depending on the time of year. The straw pens were regularly bedded (~3 × per week) with fresh straw and were mucked out, steam cleaned and disinfected between calf batches. Each year when the calves were turned out the calf pens were thoroughly cleaned and disinfected.

Faecal samples were collected directly from the rectum from calves (n=2-6) on each farm as described in section 2.1.1.

3.3.4 Analysis of Calf Faecal Samples

Following sample collection the oocysts present in each sample were counted as described in section 2.3.2.1 and DNA extracted from the faeces following the protocol in section 2.5.2.

3.3.4.1 Microsatellite PCR primers and conditions

A nested PCR protocol was used to amplify regions of five previously described microsatellite markers (Morrison et al., 2008). Each 20 µl reaction contained 10 × PCR buffer (45 mM Tris-HCl pH 8.8, 11 mM (NH₄)₂SO₄, 4.5 mM MgCl₂, 4.4 µM EDTA, 113 µg ml-1 BSA, 1 mM each of the four deoxyribonucleotide triphosphates), 0.5 units BioTaq (Bioline, UK), 10 µM of forward and reverse primers (Table 3-2) and 1 µl DNA in the primary round and 1 µl primary PCR product diluted 1:100 in the secondary round. In the secondary round 10 µM primers labelled with FAM or HEX were used (Table 3-2). The total volume was

made up to 20 μ l with dH₂O. In each PCR run one set of positive control DNA and negative controls consisting of dH₂O were included.

Primer	Sequence 5'-3'
MM5 (EF)	TCACAAGTTACCCCTTCTGATGCTG
MM5 (ER)	TCCACCTCCGGATTGGTTGTG
MM5 (IF)	CCTGGACTTGGATTTGGACTTACACC
MM5 (IR)	GGAGAAGATAAGCTAGCCGAATCT
MM18 (EF)	GTTCAGCTGATACGGGTTTGCAACA
MM18 (ER)	
MM18 (IF)	CTTTCTGGAGGGTTTGTTCCTCC
MM18 (IR)	
MM19 (EF)	TGGTTTTAGCTAAGGAAGCGATAG
MM19 (ER)	070070070770777
MM19 (IF)	GATTCTGTCAACTTTGAATTCAG
MM19 (IR)	
	CCAACCCCGAATTCATTTCCAAC
1P14 (EF)	GAGAAGGAGCAATGGGAGCA
TP14 (ER	TCCTCCTTTTTGCCCTTGAA
TP14 (IF)	CTAACGTTCACAGCCAACAGTACC
TP14 (IR)	CAATAAAGACCATTATTACCACC
MS1 (EF)	AAGGGTGAGGATGAGCAGAA
MS1 (ER)	TTCTTAACTTTCCATTTTGAGTGA
MS1 (IF)	TTAGTCGACCTCTTCAACAGTTGG
MS1 (IR)	GGAACACCATCCAAGAACCAAAGGT

Table 3-2: Primers used for microsatellite analysis of Cryptosporidium isolates

Secondary amplification products were visualised following electrophoresis on a 2% agarose gel stained with GelRedTM (Biotium, UK) on an AlphaImager 2000).

3.3.4.2 Microsatellite Analysis and Allele Identification

Labelled PCR products were subjected to fragment analysis using capillary electrophoresis (CE) via ABI 3730 (Applied Biosystems; University of Dundee), using size standard Genescan ROX500 (Applied Biosystems). A subset of samples was also re-sized using ROX400HD (Applied Biosystems). Trace files were then analysed using both Peak Scanner (Applied Biosystems) and STRand (www.vgl.ucdavis.edu/informatics/strand).

3.3.4.3 gp60 subtyping

Subtyping of *C. parvum* positive samples was carried out by amplification and sequencing of the gp60 gene (see sections 2.6.3 and 2.7). The PCR products were visualised on a 1.5% agarose stained with GelRed (Biotium, UK). Reactions were originally carried out in duplicate and if amplification was not successful the reactions were repeated. A sample was considered positive if a band corresponding with ~375 bp was seen. Positive, negative and extraction controls were included for each set of reactions. A positive result was only accepted when all negative controls were negative and all positive controls were positive.

gp60 positive samples were sequenced in the forward and reverse orientation by GATC Biotech (GATC, Germany) as described for "supremerun96" sequencing in section 2.7.3. The sequences were analysed using Chromas Lite (version 2.01 Technelysium) and the subtypes named as according to (Sulaiman et al., 2005)

3.3.5 Purification from faeces and preparation of inoculum

Oocysts were purified from faecal samples from calves from two farms (Farm BB & Farm LH) by selecting the samples with the highest oocyst counts in the smallest volume of faeces as determined by malachite green staining and microscopy as described in section 2.3.2.1. Using sucrose flotation (section 2.3.2.2) oocysts were purified from the *Cryptosporidium* positive faeces, the viability of the oocysts was tested by carrying out an excystation assay (section 2.3.3.1) which mimics the gut conditions and stimulates the oocysts (if viable) to excyst.

To prepare the oocysts for inoculation they were surface sterilised using 2 ml 70% EtOH for 30 mins before being resuspended in PBS. Oocysts were then counted (section 2.3.2.1) and the oocyst suspension diluted with PBS to give the desired infective dose.

3.3.5.1 Pre-amplification in lambs

To produce sufficient oocysts to inoculate all lambs in the experimental trial two of the *C. parvum* isolates (BB & LH) were amplified by orally infecting four neonatal lambs (in two groups) with 1×10^5 *C. parvum* oocysts in 1 ml PBS. Each

group of lambs were housed in separate plastic-sided lamb crates within individual concrete stables for ~13 days and were fed and monitored three times per day. Clinical assessments were made at each feeding/monitoring point using the criteria described in section 2.2.2. Total faecal output was collected from each lamb as described in section 2.2.4 using a harness and bag system. Faecal samples containing sufficient numbers of oocysts were processed as described in section 2.3.2 and processed to produce inoculum for the experimental trial.

3.3.6 Experimental Animals

Lambs used in the experimental trial were assigned to one of six groups. Each group of lambs was housed in concrete sided isolation boxes bedded with straw, the lambs were fed on UHT-treated milk three times per day via bottle and teat in a specially designed lamb feeder (Figure 3-2) to ensure that each lamb could access only its own milk. This enabled the handler to accurately record the feed intake of each individual animal; it also made the task of feeding 36 lambs slightly easier.



Figure 3-2: Lambs in specially designed feeder.

Lambs were fed UHT-treated milk three times per day in a specially designed feeder which allowed each lamb to be fed individually.

Lambs were fed exclusively on UHT-treated milk for the first two weeks of life with *ad libitum* access to hay and water, after this they were fed UHT-treated milk three times per day with *ad libitum* access to hay, water and pelleted lamb creep feed. Once the lambs reached six weeks of age they were fed UHT-treated milk twice per day and had *ad libitum* access to hay, water and pelleted lamb creep feed.

3.3.7 Collection of samples

3.3.7.1 Faecal Collection

Total faecal output was collected from each lamb using a specially designed harness and bag system (see section 2.2.4.1) which the lamb wore continuously (Figure 3-3). This enabled the lambs to be penned in groups rather than separately but still allowed collection of faeces from each lamb individually. This system also ensured that after inoculation the lambs were not exposed to any faecal matter and were not re-infected by oocysts shed in the faeces. Bags were taken off and replaced with a new one clearly marked with group/animal/sample when required. Samples were stored in the bags at 4°C until they were processed (see section 2.4.2).

3.3.7.2 Blood Sampling

Blood samples were taken weekly from each lamb by jugular venipuncture using a vacutainer needle (BD Vacutainer, USA). Blood samples were left overnight at 4° C to clot then centrifuged at 2000 × g for 10 minutes and the serum transferred to a 1.5 ml tube. Serum samples were stored at -20°C. Serum samples will be used to monitor the level of specific *Cryptosporidium* antibodies in an ELISA. Due to time constraints it was not possible to develop an ELISA during the course of this PhD.



Figure 3-3: Lambs with the harness and bag system to collect faeces Lambs wore a harness and bag at all times to ensure that total faecal output was collected and the animals were kept away from oocysts which were shed in the faeces

3.3.8 Clinical Assessment

Lambs were monitored a minimum of three times per day for the first six weeks and then a minimum of twice daily for the remainder of the study. Daily monitoring involved the recording of the volume of milk each lamb drank along with how well the animal drank, the demeanour of the animal and the appearance of the faeces produced, see section 2.2.2 for more details.

3.3.9 Sample Processing

Samples collected from the lambs were processed individually. The sample was weighed and the weight recorded, water was then added to the sample to make the faeces liquid enough to pipette. If the sample was already liquid then no additional water was added. The sample was then re-weighed and the new weight recorded. Using a food processor the sample was homogenised completely and a 2 ml sample aliquoted into a tube for storage for use in DNA extraction. These aliquots were stored at -20°C. A small subsample ~850 µl was

taken and used to quantify the oocysts present in the sample (see section 2.3.2.1). The remainder of the faecal material was discarded (Figure 3-4).



Figure 3-4: Sample processing

Entire faecal samples from each lamb were processed in the lab (A), the entire motion was homogenised in a food blender (B) and a sub-sample taken for counting of oocysts (C)

3.3.9.1 Oocysts counts

Oocysts present in the faeces were counted by staining the faecal suspension with 0.16% malachite green and examination under × 40 magnifications on a haemocytometer as described in section 2.3.2.1. A selection of samples from each lamb were counted; every sample produced by the lamb during the clinical stage of infection and 1-2 samples per week in the non-clinical stages were processed and counted.

3.3.10 Statistical Analysis

All statistical analyses were performed using Minitab17 (Version 17.1.0.0, Minitab Inc.). Outcome and explanatory variables are shown in (Table 3-3: Outcome and explanatory variables). Survival curves were produced using R statistical package.

Table 3-3: Outcome and explanatory variables

Codes used for definitions of value ranges and analytical approaches used for outcome and explanatory variables

	Code	Description	Value	Analysis
Diarrhoea D1 during the wk1 f D2 during the wk6 f FD2 during time FD1 during time FD2 during time Mumber of co DD1 diarrhoea or time period OC1 Total oocy time period	D1	Diarrhoea occurring during the time period wk1 to wk4	Yes/No	Chi-Squared test for univariate and logistic regression for multivariate
	D2	Diarrhoea occurring during the time period wk6 to wk8	Yes/No	Chi-Squared test for univariate and logistic regression for multivariate
	First day of diarrhoea during time period wk1 to wk4	1-28	Survival curve - Kaplan-Meier estimate	
	FD2	First day of diarrhoea during time period wk6 to wk8	42-56	Survival curve - Kaplan-Meier estimate
	DD1	Number of days on which diarrhoea occurred in the time period wk1 to wk4	0-28	One-way ANOVA
	DD2	Number of days on which diarrhoea occurred in the time period wk6 to wk8	0-28	One-way ANOVA
	OC1	Total oocyst count for time period wk1 to wk4	0-1.8E ⁹	Logn transformed then One-way ANOVA
	OC2	Total oocyst count for time period wk6 to wk8	0-1.8E ⁹	Logn transformed then One-way ANOVA

	Code	Description	Value
	Age	Age of primary infection	D7, D42
anatory iables	Isolate	Isolate used for challenge	BB, LH
	Dose	Challenge dose	Low, High
	Challenge	Previous exposure or not	Primary, Secondary
Expl Var	Туре	Secondary challenge with same isolate or not	Homologous, Heterologous
	Group	Group to which lamb was assigned	I - VI
	Animal	Unique ID within Group	A-F

3.3.10.1 Clinical Disease

The association between clinical disease (D1, D2, FD1, FD2, DD1 & DD2) for the dependant variables was analysed using chi-squared test, binary logistic regression and One-way ANOVA. The data for FD1 and FD2 was right-censored and survival curves were produced using R statistical package. The interval to morbidity (diarrhoea) was tested for each isolate using the "survdiff" function of R, using the G-rho family of Harrington and Fleming (1982). Weights on each morbidity are $S(t)^{rho}$, where S is the Kaplan-Meier estimate of survival, with rho = 0, this is the log-rank or Mantel-Haenszel test, and with rho = 1, it is equivalent to the Peto & Peto modification of the Gehan-Wilcoxon test.

3.3.10.2 Oocyst Shedding

Logn transformation was applied to all oocyst count data which resulted in approximately normal distributions, according to Anderson-Darling tests for normality, taken together with visual inspection of histograms of transformed data. One-way ANOVA with Tukey's post-hoc test was used to explored the association with explanatory variables.

3.4 Results

3.4.1 Cryptosporidium parvum isolates

3.4.1.1 Microsatellite analysis to identify suitable isolates

Microsatellite analysis of 6 markers including gp60 was carried out to ensure that the selected isolates were genetically distinct. The results of the analysis revealed that all three of the selected isolates were genetically distinct at at least one locus (Table 3-4). BB and LH could be distinguished at 3 markers (MM5, MM19 and gp60), BB and OV at 2 markers (MM5 and MM19). LH and OV could only be distinguished at 1 marker (gp60).

Table 3-4: Alleles assigned to each isolate

	Marker	Isolate BB	Isolate LH	Isolate OV
1	MM5	Allele 2	Allele 1	Allele 1
	MM18	Allele 1	Allele 1	Allele 1
	MM19	Allele 3	Allele 4	Allele 4
	TP14	Allele 2	Allele 2	Allele 2
	MS1	Allele 1	Allele 1	N/A
_	gp6 0	IlaA15G2R1	IIaA19G2R1	IIaA15G2R1

Red text indicates markers at which the isolates differed.

3.4.1.2 Faecal Consistency and Oocyst Shedding in Sampled Calves

These isolates presented different clinical signs in terms of oocysts shed and presence of diarrhoea in naturally infected calves (from personal observation and previous experience of *Cryptosporidium* infected calves on these two farms) (Table 3-5).

Table 3-5: Faecal consistancy and average oocysts counts associated with different (C.
parvum isolates in naturally infected calves	

Isolate	Faecal Consistency	Average Oocyst Count (per ml)	Number of Samples Collected
BB	No diarrhoea	1.53 × 10 ⁷	n=2
LH	Watery diarrhoea	1.72 × 10 ⁶	n=6
OV	Slight diarrhoea	3.24 × 10 ⁷	n=4

3.4.1.3 C. parvum isolates used in experimental trial

Isolates BB and LH were chosen as the ideal isolates to take forward to the trial as they differed in clinical manifestation as well as being genetically distinct at three loci. These isolates also came from calves on closed herd farms which decreased the likelihood that the infections were mixed. Isolate OV was excluded from the trial at this stage.

3.4.2 Production of inoculum

In total eight samples were obtained and were processed and the oocysts counted. The total numbers of oocysts obtained from each farm are shown in (Table 3-6). Samples containing high numbers of oocysts in a small volume of faeces were chosen for further processing to obtain oocysts for production of inoculum.

Sample	Type [†]	Volume (ml)	No. Oocysts	Total No. Oocysts
BB1*	3	194	1.84 × 10 ⁹	2.75×10^9
BB2	3	90	1.91 ×10 ⁹	3.75 × 10
LH1	3	108	1.35 × 10 ⁷	
LH2*	5	73	6.02 × 10 ⁸	
LH3*	5	65	6.91 × 10 ⁷	9 71 v 10 ⁸
LH4	4	189	0	0.71 × 10
LH5	3	294	1.10 × 10 ⁸	
LH6*	5	152	7.60 ×10 ⁷	

Table 3-6: Oocyst counts from faeces obtained from calves farms BB & LH

*these samples were processed by sucrose flotation to produce the inoculum for pre-amplification † See Table 2-1

The oocysts obtained from LH2, LH3, LH6 were pooled as a single sample did not contain sufficient oocysts. Before producing inoculum the viability of the oocysts were purified by sucrose flotation (section 2.3.2.2) prior to an excystation assay which mimics the gut conditions and stimulates the oocysts to excyst, releasing sporozoites (section 2.3.3.1). A sporozoite:shell ratio of >2 and an excystation percentage of >85% is desirable (Katzer F, Pers Comm.) (Table 3-7).

Isolate	Sporozoite:Shell ratio	Excystation %age
BB	2.8	97%
LH	2.1	88%

Table 3-7: Sporozoite: Shell ratios and excystation percentages for each isolate

3.4.3 Pre-amplification of Cryptosporidium parvum isolates

Four neonatal lambs were successfully inoculated with two different *C. parvum* isolates (n=2 lambs per group/isolate) by oral infection using a syringe and 1×10^5 oocysts in PBS. All lambs showed clinical signs of disease (diarrhoea, inappetence) approximately three DPI (days post-infection) and oocyst shedding occurred in all animals (Figure 3-5).





3.4.3.1 Sample Processing

Positive samples containing the highest number of oocysts in the smallest volume of faeces were processed by sucrose flotation to separate the oocysts from the faecal matter. The sucrose flotation step was repeated for samples from Group B as the faeces, in this case, was slightly thicker than ideal. In total, 6.73×10^9 oocyst in 430 g faeces from Group A (BB) and 1.01×10^{10} oocysts in 596 g faeces from Group B (LH) were processed. From Group A, 2.36×10^9 oocysts were recovered (35% recovery) and from Group B, 5.5×10^8 oocysts were recovered (5% recovery).

A minimum of 4.2×10^7 oocysts of each isolate were required to inoculate all 36 lambs in the trial.

3.4.3.2 Excystation Assay

Cleaned oocysts were subjected to an excystation assay prior to each inoculation (<1 week & 6 weeks) (section 2.3.3.1) to determine the excystation percentage and sporozoite:shell ratio (Table 3-8).

Table 3-8: Sporozoite:Shell ratios and excystation percentages for inoculum used in experimental trial

Challenge	Isolate	Sporozoite:Shell ratio	Excystation %age
<1 week	BB	2.21	95%
	LH	2.94	92%
6 weeks	BB	2.04	74%
	LH	1.31	82%

The excystation percentages and sporozoite:shell ratios were slightly lower than ideal (>2 sporozoite:shell ratio/>85% excystation) for the challenge at 6 weeks of age.

3.4.4 Experimental Animals – clinical observations

All lambs drank well even following infection, some animals showed mild signs of abdominal pain (lying down after feeding) one or two days after infection but they did not stop feeding. Feed intake between groups was similar (**Error! Reference source not found.**).



Figure 3-6: Average feed intake of each group. Lambs did not go off their feed following infection with *C. parvum.*

3.4.4.1 Age-related Susceptibility to Disease

To determine whether animals become less susceptible to disease as their age increases two groups (Group I and Group II) were infected with *C. parvum* at 6 weeks of age and the disease outcome and oocyst shedding was compared with lambs which were infected at <1 week of age (Group III, Group IV, Group V and Group VI). Lambs infected at 6 weeks of age were given with a higher dose (5×10^6) than animals infected at <1 week of age (1×10^6) (Figure 3-1)..

Isolate BB

Animals which were infected with BB for the first time at 6 weeks of age (Group I) showed milder clinical disease (number of diarrhoeic samples and duration of diarrhoea) compared with animals infected with this isolate for the first time at <1 week of age (Group III & Group IV). Overall, following primary infection at 6 weeks of age only one sample collected from one animal in Group I (BB wk6) was scored as diarrhoeic compared with 12 samples from four animals in Group III and 32 samples from six animals in Group IV (BB wk1). (Figure 3-7).

Isolate LH

Animals which were infected with isolate LH for the first time at 6 weeks of age (Group II)showed milder clinical disease compared with animals infected for the first time at <1 week of age (Group V & Group VI).Overall, following primary infection at 6 weeks of age five samples collected from one animal in Group II were scored as diarrhoeic. In contrast, 23 samples collected from six animals in Group V and 55 samples from six animals in Group VI were scored as diarrhoeic. (Figure 3-7).



Age at Primary Infection

Figure 3-7: Number of days lambs produced a diarrhoeic sample following primary-infection at <1 week (Group III, IV, V & VI) and 6 weeks (Group I & II) of age.

Lambs infected at 6 weeks of age produced fewer diarrhoeic samples than lambs infected at 6 weeks of age. Red bars show primary infection with BB isolate and blue bars show primary infection with LH isolate. (* indicates outliers, the horizontal line the median and the vertical line indicates that the median is zero). Animals challenged at 6 weeks (Group I & II) had a shorter duration of diarrhoea than animals challenged at <1 week of age (Group III, IV, V & VI) and produced fewer diarrhoeic samples (P<0.001).

3.4.4.2 Difference between isolates

To compare the pathogenicity of the two isolates one age-related susceptibility, one homologous and one heterologous challenge per isolate was carried out. Results show that the isolate used in the primary infection at <1 week of age does not significantly affect the probability of diarrhoea during the first three weeks post-infection. There was no significant effect of isolate on diarrhoea following challenge at 6 weeks of age.

Isolate had a significant effect on the interval to onset of diarrhoea (P < 0.05, Table 3-9a and Figure 3-8) such that isolate LH resulted in earlier onset diarrhoea following primary infection. There was no effect of isolate on days to diarrhoea after the second challenge (P= 0.434, Table 3-9b and Figure 3-8).





Lambs challenged with isolate LH at <1 week of age began scouring sooner than animals infected with isolate BB. There was no difference in time to onset of diarrhoea following challenge at 6 weeks of age.

Isolate 1	Ν	Observed(O)	Expected(E)	(O-E)^2/E	(O-E)^2/V	
BB	18	13	18.5	1.61	4.91	
LH	18	17	11.5	2.58	4.91	
Chisq= 4.9 on 1 degrees of freedom, p= 0.0266						

Table 3-9a: Log-rank test on Kaplan-Meier survival curves for onset of diarrhoea after challenge at <1 week of age.

Table 3-9b: Log-rank test on Kaplan-Meier survival curves for onset of diarrhoea after challenge at 6 weeks of age.

Isolate 1	Ν	Observed(O)	Expected(E)	(O-E)^2/E	(O-E)^2/V
BB	18	4	3.04	0.301	0.612
LH	18	2	2.96	0.310	0.612
Chisq= 0.6 on 1 degrees of freedom, p= 0.434					



Figure 3-9: Number of days lambs infected with BB (red) or LH (blue) isolate produced a diarrhoeic sample in the 3 weeks post-primary infection at <1 week of age.

3.4.4.3 Previous exposure (homologous and heterologous challenge)

To determine if previous exposure to *C. parvum* provides protection against a secondary challenge later in life two sequential challenges were carried out to test each isolate in a homologous and heterologous challenge.

All animals which received a primary infection or secondary challenge at 6 week of age showed a reduction in clinical disease compared with animals which received a primary infection at <1 week of age. Previous exposure did not provide significant protection (P>0.99) against re-infection and the occurrence of clinical disease. The effect which age had on the occurrence of diarrhoea masked any effect that previous exposure may have had. There was no significant difference in the occurrence (P>0.99) or days of diarrhoea (p=0.378) between groups with previous exposure (Group III, IV, V & V) and groups which had no previous exposure (Group I & II) (Figure 3-10 and Table 3-10).



Figure 3-10: Occurrences of diarrhoea following challenge at 6 weeks

There was no difference in the occurrence or days of diarrhoea following challenge at 6 weeks in lambs receiving (purple boxes) or not receiving (orange boxes) a primary infection. (* indicates outliers, the horizontal line the median and the vertical line indicates that the median is zero).

There was no difference (P=0.571) in clinical disease caused by a homologous or heterologous challenge (Table 3-10), again, any effect of the type of challenge (homologous or heterologous) may be masked by the effect of age.

Outcome Variable	Expla Var	natory iable	1	0	Odds Ratio	95% CI	P-value
	Isolate						0.086
		BB	13	5	1		
Dutcome Variable D1		LH	17	1	*	0.00 - *	
D1	Group						0.0837
		<u> </u>	3	3	1		
		II	5	1	5.00	0.55 – 45.39	
			4	2	2.00	0.31 – 13.06	
		IV	6	0	0	0.00 - *	
		V	6	0	0	0.00 - *	
		VI	6	0	0	0.00 - *	
	Isolate						1
		BB	4	14	1		
		LH	2	16	1	0.17 – 5.77	
	Group						0.0977
		<u> </u>	1	5	1		
		II	1	5	1	0.05 – 20.83	
			1	5	1	0.05 – 20.83	
		IV	1	5	1	0.05 – 20.83	
ר2		V	2	4	2.5	0.16 – 38.60	
DZ		VI	0	6	0	0.00 - *	
	Туре						0.571
		*	2	10	1		
		Homologous	1	11	1.67	0.22 – 12.37	
		Heterologous	3	9	0.45	0.04 – 5.81	
	Challenge						1
		Primary	1	10	1		
		Secondary	4	20	1	0.16 – 6.42	

Table 3-10: Multivariate logistic regression model for analysis of factors associated with occurrence of diarrhoea in wk1 – wk4 and wk6 – wk8.

3.4.5 Experimental Animals – oocyst shedding

Due to time constraints it was not possible to carry out oocyst counts for all groups, instead oocyst counts were carried out for four of the six groups. Oocyst

counts were carried out on samples from animals in Group I & II (age-related susceptibility to disease), Group III (homologous challenge with BB) and Group V (heterologous challenge).

3.4.5.1 Group I

Oocysts counts were carried out on 167 (52.35%) samples from six lambs in Group I. One sample from each lamb was counted each week for the first 6 weeks to ensure that the animal remained *Cryptosporidium*-free for this period. Following primary challenge every sample produced by each lamb was processed and counted. The number of samples processed from each lamb is summarised in Table 3-11.

Lamb	Samples Collected	Samples Processed and Counted	Percentage (%) of Samples Processed
А	59	30	50.84
В	60	22	36.60
С	46	26	52.38
D	54	29	53.70
E	57	32	56.14
F	43	28	65.11

Table 3-11: Samples collected and processed from each lamb in Group I

None of the lambs showed any signs of *Cryptosporidium* infection in the first 6 weeks of life and all samples tested were negative for oocysts. Following primary challenge at 6 weeks lambs started to shed oocysts at 1-2 DPI, shedding continued for 12-16 days before tailing off. Peaks in oocyst shedding occurred at 2, 4, 8, 11 & 15 DPI in individual animals (Figure 3-11). Oocyst shedding between lambs differed significantly (P=0.037) with one lamb (C) shedding higher numbers of oocysts than other animals (Figure 3-12).



Figure 3-11: Oocyst counts from lambs in Group I

Shedding profiles from lambs in Group I following primary challenge at 6 weeks of age with 5 × 10⁶ *C. parvum* oocysts isolate BB. Lambs began shedding oocysts very soon after inoculation with peaks in oocyst shedding at 2, 4, 8, 11 & 15 DPI in individual animals. Shedding began to tail off at 14 DPI. It was not possible to collect a sample from each animal on each day, days on which samples were collected are indicated by a data point.



Figure 3-12: Total number of oocysts shed by individual lambs in Group I (BB wk6) Oocyst shedding between individual animals within the same group varied with some animals shedding significantly more oocysts than others. (* indicates outliers).

3.4.6 Group II

Oocysts counts were carried out on 150 (48.85%) samples from six lambs in Group II. At least one sample from each lamb was counted each week for the first 6 weeks to ensure that the animal remained *Cryptosporidium*-free for this period. Following primary challenge every sample produced by each lamb was processed and counted. The number of samples processed from each lamb is summarised in (Table 3-12).

Lamb	Samples Collected	Samples Processed and Counted	Percentage (%) of Samples Processed
А	50	30	60.00
В	63	28	44.44
С	59	26	44.06
D	39	21	53.84
E	45	19	42.22
F	51	26	50.98

Table 3-12 [.] Sam	oles collected and	processed from	each lamb in Group	n II

None of the lambs showed any signs of *Cryptosporidium* infection in the first 6 weeks of life and all samples tested (n=52) were negative for oocysts. Following primary challenge at 6 weeks lambs started to shed oocysts at 1-2 DPI, shedding continued for 11 days before tailing off. Peaks in oocyst shedding occurred at 3, 5, 8 & 10 DPI in individual animals (Figure 3-13). Oocyst shedding between lambs differed but the difference in shedding was not significant (Figure 3-14).



Figure 3-13: Oocyst counts from lambs in Group II

Shedding profiles from lambs in Group II following primary challenge at 6 weeks of age with 5×10^6 *C. parvum* oocysts isolate LH. Lambs began shedding oocysts very soon after inoculation with a peak in shedding at approximately 3 DPI. Shedding began to tail off at 11 DPI. It was not possible to collect a sample from each animal on each day, days on which samples were collected are indicated by a data point.



Figure 3-14: Total number of oocysts shed by individual lambs in Group II. Oocyst shedding between lambs in the same group varied but not significantly. (* indicates outliers).

3.4.7 Group III

Oocysts counts were carried out on 335 (73.62%) samples from six lambs in Group III. Following primary infection (BB) every sample produced by each lamb was processed and counted for 3-4 weeks then 1-2 samples per week for two weeks and then every sample following secondary challenge (BB) for a further 19 days. The number of samples processed from each lamb is summarised in Table 3-13.

Lamb	Samples Collected	Samples Processed and Counted	Percentage (%) of Samples Processed
А	65	51	78.46
В	73	58	79.45
С	83	67	80.72
D	82	69	84.14
Е	74	60	81.08
F	68	50	73.53

Table 3-13: Samples collected an	d processed from each lamb in Group III
Table 5-15. Samples conected an	

None of the lambs showed any signs of *Cryptosporidium* infection in the days prior to primary challenge and all samples tested (n=26) were negative for Cryptosporidium oocysts. Following primary infection lambs began to shed oocysts at 14 DPI, shedding continued until 26-28 DPI when oocysts shedding tailed off for the majority of animals. Peaks in oocysts shedding occurred at 15, 18, 21 and 23 DPI in five of the six animals. Within this group animal F consistently shed higher numbers of oocysts than the others with peaks in shedding occurring at Days 24 and 31 PI. By 27 DPI oocyst shedding had decreased to $<2.5 \times 10^7$ in most of the animals (Figure 3-15 & Figure 3-16).

Animals in this group were re-challenged with the same isolate (BB) at 39 days of age, the dose of the secondary challenge was higher than the primary infection (primary infection 1×10^6 and secondary challenge 5×10^6).

Following secondary challenge animals began shedding oocysts at 2-4 DPI and continued to shed oocysts for 10-17 days. Peaks in shedding occurred at 5,7 and 11 DPI in the majority of animals, again animal F shed higher numbers of oocysts than the others and in this animal peaks in oocyst shedding occurred at 4, 6, 9, 12, 14 and 17 DPI.

All animals shed more oocysts following secondary challenge than following primary infection with the same isolate. In total the lambs in this group shed 1.80×10^{10} oocysts, 4.97×10^{9} following the primary infection and 1.31×10^{10} following the secondary challenge.

There was a significant difference (P<0.001) between the number of oocysts each lamb shed with some lambs, particularly animal F shedding higher numbers of oocysts than the others. This animal shed ~ 5.00×10^9 oocysts more than the other animals in this group following secondary challenge (Figure 3-17).



Figure 3-15: Oocyst counts from animals in Group III

Shedding profiles from lambs in Group III following primary infection at <1 week of age with 1×10^6 *C. parvum* oocysts isolate BB and challenge at 6 weeks of age with 5×10^6 *C. parvum* oocysts isolate BB. Shedding began to tail off at 14 DPI. It was not possible to collect a sample from each animal on each day, days on which samples were collected are indicated by a data point.



Figure 3-16: Oocyst counts from animals in Group III excluding animal F

Shedding profiles from lambs in Group III following primary challenge at <1 week of age with 1×10^6 *C. parvum* oocysts isolate BB and re-challenge at 6 weeks of age with 5×10^6 *C. parvum* oocysts isolate BB. It was not possible to collect a sample from each animal on each day, days on which samples were collected are indicated by a data point. Animal F has been excluded from this graph in order to see the shedding profile of the other 5 lambs more clearly. Animal F shed more oocysts than the other lambs in the study.



Figure 3-17: Total number of oocysts shed by individual lambs in Group III Oocyst shedding between individual animals within the same group varied with some animals shedding significantly more oocysts than others. (* indicates outliers).

3.4.8 Group V

Oocysts counts were carried out on 275 (75.55%) samples from 6 lambs in Group V. Following primary infection (LH) every sample produced by each lamb was processed and counted for 3-4 weeks then 1-2 sample per week for two weeks and then every sample following secondary challenge (BB) for a further 15 days. The number of samples processed from each lamb is summarised in Table 3-14.

Lamb	Samples Collected	Samples Processed and Counted	Percentage (%) of Samples Processed
А	64	43	67.19
В	67	53	79.10
С	67	48	71.64
D	69	51	73.91
E	51	38	74.51
F	46	42	91.30

None of the lambs showed any signs of *Cryptosporidium* infection in the days prior to primary challenge and all samples tested (n=26) were negative for *Cryptosporidium* oocysts. Following primary infection lambs began to shed oocysts at 11-18 DPI, shedding within animals in this group did not tail off as in other groups. In the majority of animals oocyst shedding continued at <1.00 × 10^8 until they were re-challenged at 38 days of age. Peaks in oocysts shedding occurred at 16, 21 and 30 DPI in most of the lambs although oocysts shedding within animals in Group V was more sporadic than in other groups.

Animals in this group were challenged with a different *C. parvum* isolate (BB) at 38 days of age, the dose of the secondary challenge was higher than the primary infection (primary infection 1×10^6 and secondary challenge 5×10^6).

Following secondary challenge animals continued shedding oocysts and all animals showed an increase in the number of oocysts shed compared with the previous challenge. Following secondary challenge oocyst shedding continued for 10-15 days. Peaks in shedding occurred at 2,5 and 11 DPI in the majority of animals (Figure 3-18 & Figure 3-19) with animals E & F shedding significantly (P<0.001) higher numbers of oocysts than the others (Figure 3-20).



Figure 3-18: Oocyst counts from Group V

Shedding profiles from lambs in Group V following primary challenge at <1 week of age with 1×10^6 *C. parvum* oocysts isolate LH and re-challenge at 6 weeks of age with 5×10^6 *C. parvum* oocysts isolate BB. It was not possible to collect a sample from each animal on each day, days on which samples were collected are indicated by a data point.



Figure 3-19: Oocyst counts from Group V excluding animals E & F

Shedding profiles from lambs in Group V following primary challenge at <1 week of age with 1×10^6 *C. parvum* oocysts isolate LH and re-challenge at 6 weeks of age with 5×10^6 *C. parvum* oocysts isolate BB. It was not possible to collect a sample from each animal on each day, days on which samples were collected are indicated by a data point. Animals E & F have been excluded from this graph in order to see the shedding profile of the other 4 lambs more clearly. Animals E & F shed more oocysts than the other lambs in the study.


Figure 3-20: Total number of oocysts shed by individual lambs in Group V Oocyst shedding between animals in group V varied significantly with lambs E and F shedding more oocysts than the other lambs. (* indicates outliers).

3.5 Effect of age and infective dose on shedding

When comparing the patterns of oocyst shedding between the groups, animals which were infected at <1 week of age with a low dose (1×10^6) began shedding oocysts after approximately 14 DPI whereas animals which were infected for the first time at 6 weeks of age with a higher dose (5×10^6) began shedding after a much shorter period of 1-2 DPI. Animals re-challenged with a higher dose (5×10^6) at 6 weeks of age (Group III & V) also began shedding oocysts sooner at 1-2 DPI. There was no significant difference (P=0.457) in number of oocysts shed between animals infected at <1 week of age and animals infected at 6 weeks of age.

3.5.1 Variation between isolates

Following primary infection at <1 week of age (Group III & Group V) there was no significant difference in number of oocysts shed and the isolate which the lambs were infected with (P=0.059). Following primary infection (Group I & Group II)

and secondary challenge (Group III & Group V) at 6 weeks of age infection with BB isolate was significantly associated with higher oocyst output (P<0.001).

3.5.2 Previous Exposure

Animals which had been previously exposed to *C. parvum* (Group III) & Group V) shed more oocysts following secondary challenge at 6 weeks of age than naïve animals (Group I & Group II) which received a primary infection at the same age (P=0.02)



Figure 3-21 Average oocysts counts (Group I, II, III & V) following primary infection. Lambs receiving a high (5×10^6) dose at primary infection shed more oocysts and had a shorter prepatent period than lambs receiving a low (1×10^6) dose.

3.6 Discussion

Infection with *Cryptosporidium* is very common in young ruminant livestock (de Graaf et al., 1999). As reported, in this thesis (Chapter 5 & Chapter 6) and other work, animals may become infected with *Cryptosporidium* at a very early age (Ortega-Mora et al., 1999; Tzipori et al., 1983) indicating that they have been exposed to the parasite almost immediately at birth. As a result it is difficult to examine the development of resistance to disease in a field setting. In this experiment the susceptibility of lambs to disease caused by two distinct *C*. *parvum* isolates was tested by a primary infection at <1 week and a homologous or heterologous challenge at 6 weeks of age as well as testing the susceptibility of lambs kept in isolation from the parasite and then experimentally challenged at 6 weeks of age.

Although the isolates used in this trial were originally derived from naturally infected calves it was decided to carry out the trial using lambs. From previous and ongoing work at the Moredun Research Institute it is known that lambs are susceptible to infection with *C. parvum* and exhibit similar clinical signs and patterns of oocyst shedding. Lambs are smaller and easier to handle than calves especially once they grow to 8 weeks of age. In addition, total faecal output from each individual animal was collected for the duration of the trial. Total faecal output from a calf infected with *C. parvum* could be as much as 3 kg per day compared to 600 g from a lamb (Bartley, Pers. Comm.). The results of this study suggest that lambs can be used as a model for disease outcome and shedding in calves as the clinical disease and shedding patterns of experimentally infected lambs, in general, were consistent with that of the naturally infected calves from whom the oocysts were originally derived.

3.6.1 Age-related clinical disease and oocyst shedding

The results of this study confirmed previous reports which demonstrated that livestock (calves and lambs) which were naïve to *Cryptosporidium* were still susceptible to the parasite at an older age (Harp et al., 1990; Ortega-Mora and Wright, 1994). In the present study the animals challenged at 6 weeks of age suffered from significantly less severe clinical disease than animals challenged at <1 week of age although they all still showed evidence of oocyst shedding. The same is true of the lambs challenged in a previous study (Ortega-Mora and Wright, 1994). The lambs in that study were infected at 6 days and then rechallenged (with the same isolate and dose (1×10^6)) at 4 weeks of age. In parallel, a further two groups of lambs were infected once at 4 and 8 weeks of age respectively. Lambs which were naïve to C. parvum at 4 weeks still showed clinical disease (diarrhoea) and shed oocysts whereas lambs with prior exposure (infected at 1 week of age) had no clinical disease or (detectable) oocyst shedding when re-challenged at 4 weeks of age. This demonstrates that lambs can develop specific immunity to infection with this particular C. parvum isolate if they have previous exposure to the same parasite. When naïve lambs were challenged at 8 weeks of age none of the lambs showed evidence of clinical disease though some lambs did shed oocysts although this was over a shorter duration and lower numbers of oocysts were shed. This indicates that sometime between 4 (naïve lambs still susceptible to disease) and 8 weeks (naïve lambs not susceptible to disease) of age lambs develop an age-related resistance to Cryptosporidium infection.

In the present study we demonstrate that lambs become less susceptible to disease at around 6 weeks of age with a reduction in the percentage of animals showing clinical signs of disease to 17-33% from 50-100% depending on isolate despite an increase in challenge dose. This shows that as the animals increase in age they are becoming less susceptible to cryptosporidiosis with a reduction in disease severity at 6 weeks but no change in oocyst shedding. In a similar study in calves, Harp et al (1990) reported that when naïve calves were challenged with C. parvum at 1 or 3 months of age they remained fully susceptible to disease and all challenged animals showed clinical signs (diarrhoea) as well as oocyst shedding. Two of the animals which were initially challenged at 3 months were re-challenged at 6 months of age and neither showed any evidence of clinical disease or oocyst shedding. There were no calves which were naïve to *Cryptosporidium* challenged at this age so it is not known if the calves developed resistance to disease because of previous exposure or because of an age-related immune response. Based on this study it seems that, in calves, a previous challenge can protect against re-infection indicating that exposure of calves to C. parvum results in specific acquired immunity but the age at which they develop age-related immunity to disease is not known. However, as will be

reported later in this thesis, it is possible for calves with previous exposure to one isolate of *C. parvum* to become infected with, and shed oocysts of, another *C. parvum* isolate later in life (Chapter 5).

It is possible that this age-related resistance to symptomatic disease in lambs (6-8 weeks of age) and calves (>6 months) is related to changes within the animal gut. These changes usually occur at weaning or when the diet of the animal is changed. In the present study lambs were fed on UHT-treated milk $3 \times per day$ for the first 2 weeks with *ad libitum* access to hay, pelleted lamb feed was made available from day 14 onwards. In the study by Ortega-Mora et al., (1994) lambs were fed on UHT-treated milk only for the first 20 days after which hay and lamb rations were made available. By day 45 of age the lambs had been weaned from milk completely. In the calf study (Harp et al., 1990) calves were fed a diet of milk replacer for 2.5 months with the addition of pelleted feed at 2 weeks of age and hay at 4 weeks of age. In addition, rumen fluid from five calves at 3.5 months of age was tested and showed some degree of function but not at the level of a fully mature adult bovine. The maturity of the rumen in cattle and sheep is affected by the age of weaning and diet (Quigley, III et al., 1985). This ties in with the development of age-related resistance to disease in both previous studies (resistant to disease at 6 months (calves) and 8 weeks (lambs)) and the current study (not resistant at 6 weeks but mild clinical disease only). It is not known if an animal with mature gut function which has no prior exposure to Cryptosporidium would be susceptible to disease. From other studies, carried out as part of this PhD we know that calves can still remain susceptible to infection and can shed oocysts at 6 months of age (Chapter 5 and Chapter 6).

In mice age-related resistance to disease caused by *Cryptosporidium* has been well documented (Sherwood et al., 1982; Tarazona et al., 1997; Upton and Gillock, 1996) and it has been demonstrated that even when the mice have had no previous exposure to the parasite they were not susceptible to disease after 3 weeks of age. This age-related resistance in mice coincides with the maturation of the intestinal gut flora (about 3 weeks in mice) (Harp et al., 1992). It has also been noted that pigs become more susceptible to infection with *C. scrofarum* at around 5-8 weeks of age (Kvac et al., 2014), the authors suggested that this may

be due to changes in the gut at weaning (~6 weeks of age) which may make the gut easier for the parasite to invade.

3.6.2 Challenge dose, Clinical Disease and oocyst shedding

It appeared that when lambs received a higher infective dose (5×10^6) they began shedding oocysts sooner than when they received a lower infective dose (1×10^6) . However, due to the age difference between the two dose groups it is not possible to attribute the shortened onset to oocyst shedding following infection at 6 weeks of age solely to the higher infective dose as age of the animal may also contribute to this effect. To clarify this a further study which focuses only on effect of age OR effect of infective dose on animals of the same age should be carried out. Lambs which received a secondary challenge at 6 weeks of age shed significantly more oocysts than lambs receiving a primary infection at this age. It is possible that this is because the lambs receiving a secondary challenge are still shedding oocysts from the primary infection they received at <1 week of age. It is also a possibility that the early onset of oocyst shedding which was seen in lambs receiving a higher dose may simply be due to some ingested oocysts passing through the animal.

Within each group in this study there were significant differences in individual lamb's responses to infection in terms of oocysts shed and clinical disease with the variation between some animals being considerable. Thirty-one percent of samples collected from lamb F (Group V) over the whole study were diarrhoeic compared with just 5.2% of samples from lamb C (Group V) and in Group III lamb F shed 1.01 × 10¹⁰ oocysts in total compared with lamb A (Group III) which shed 5.75×10^8 oocyts in total, a difference of 9.50×10^9 .

Differences, sometimes considerable, in oocyst shedding (number of oocysts shed, duration of shedding) as well as clinical manifestation between individuals within groups have been noted in other *Cryptosporidium* infection trials in humans (Baxby et al., 1985; Chappell et al., 1996; Teunis et al., 2002), calves (Moore et al., 2003) and mice (Petry et al., 1995). This may be due to the different immunocompetence status of individual animals; all lambs in this study received some colostrum, either from their dams or artificial colostrum. The quality of each individual ewes colostrum was not known, nor was the quantity

that the individual lamb took. It is known that in other livestock and in humans the severity of disease depends very much on the immune status of the individual. Immunocompromised patients (human and animal) will suffer from more severe and prolonged disease than immunocompetent individuals (Baxby et al., 1985; Current et al., 1983; DuPont et al., 1995; Harp et al., 1992; Harp et al., 1988; Hojlyng and Jensen, 1988; Shepherd et al., 1988; Snyder et al., 1978).

We already know that the prepatent period is dependent on host species, Cryptosporidium species and infective dose (Tzipori et al., 1983). In this study we noted that following primary challenge at <1 week of age the animals in both Groups III & V did not begin shedding oocysts until 3-14 DPI (Group III (mean = 7.2, mode = 4, median = 5.5) Group V (mean = 9.8, mode = 14, median = 10)) whereas, when these animals were re-challenged at 6 weeks of age, with a higher dose, they began shedding oocysts much sooner, almost immediately following the oral challenge. It is not possible to determine exact number of DPI following secondary challenge as most animals did not completely cease shedding oocysts before being re-challenged. Although, it appears that 1-3 DPI the animals began shedding much higher numbers of oocysts than previously. The difference in the prepatent period following higher or lower doses has also been noted in other studies. Zambriski et al., (2013) carried out an experimental challenge in calves using doses of 25, 50, 100, 500, 1×10^3 , 1×10^4 , 1×10^5 , and 1×10^{6} C. parvum oocysts. They reported that calves receiving lower doses (25) and 50 oocysts) shed later than calves receiving higher doses (100 - 1×10^5 and $1 \times 10^{\circ}$) occysts. Other studies report a mean prepatent period of 7.4 DPI in experimentally infected calves (challenged with 50, 100, 1000 and 10,000 C. parvum oocysts) with a longer prepatent period for animals receiving a lower dose and an increase in number of days of oocyst shedding in calves receiving a higher challenge dose and 16.3 DPI in naturally infected calves (Moore et al., 2003; O'Handley et al., 1999).

Not every sample produced from each animal was positive for oocysts indicating that oocyst production and excretion is not a constant process, similar results have been noted in experimental infections of human volunteers (Chappell et al., 1996). The observation in variation of oocysts shedding day to day could be as a result of the limitations of the detection method (62500 oocysts per gram of faeces) used in this study. This suggests that for diagnostic purposes a single sample from a potentially infected individual, even if with clinical disease, may not be sufficient to detect oocysts and diagnose infection. To minimise the chances of missing potential positive samples total faecal output was collected and processed from all lambs in this study (during the clinical stage). In addition, the whole faecal sample was homogenised completely before a sub-sample was taken for analysis.

As we can see from the results of this study the amplification rate of the parasite in a young naïve animal is phenomenal; an animal infected with 1×10^6 oocysts is able to shed up to 2.73×10^9 potentially infectious oocysts in a two week period. In theory, this is enough to infect a further 2730 susceptible hosts with the same number of oocysts as the initial challenge. It is the amplification rate which leads to the rapid spread of disease within a herd or flock. From this we can see why, in a field setting, it is possible that older lambs, or even adults, may become subclinically infected with *Cryptosporidium* and maintain the parasite between lambing or calving periods.

3.6.3 Cryptosporidium parvum isolates

Despite the similarities in management on these farms the differences in disease between calves on the two farms is marked with calves on farm LH becoming very sick and calves on farm BB showing very little illness. If we take only the clinical manifestation of disease on the farms into consideration we may be able to conclude that the better early calf management (good quality colostrum within 2 h of birth, calf jackets) on farm BB means that the calves on this farm are better protected against disease in general and are better able to fight off infection. The results of the study described in this chapter indicate that these differences which were noted in naturally infected calves are mimicked the experimental infection trial in lambs. We found that isolate LH which caused more severe disease in naturally infected calves also was associated with an increased probability of diarrhoea and number of days of diarrhoea as well as a quicker onset to diarrhoea in the experimentally infected lambs and that isolate BB which seemed to cause an increase in oocyst output in naturally infected calves was shown to significantly affect the oocyst shedding of experimentally infected lambs.

Other studies have also reported variations in the severity of infection in humans and animals depending on the specific Cryptosporidium species and isolate (Cama et al., 2008; Cama et al., 2007). An experimental trial which involved infection of calves with four different C. parvum isolates (three derived from human patients and one calf derived) showed that there was a difference in disease severity between the isolates examined (Pozio et al., 1992). Although, with the added knowledge that we have now we cannot be sure that these isolates were a) truly C. parvum and not another species i.e. C. hominis (which is not infectious in calves (Morgan-Ryan et al., 2002)) and b) different isolates, in this study the oocysts were assumed to be different isolates of C. parvum based on morphology and clinical presentation in naturally infected hosts. Another study (Okhuysen et al., 1999) investigated the virulence of three C. *parvum* isolates in human volunteers (two calf derived and one horse derived) and found differences in ID₅₀, prepatent period and duration of oocyst shedding between the three isolates. These studies indicate that though host factors are very important and can be responsible for differences in individual response to infection with C. parvum it is also clear that there can be a difference in virulence and pathogenicity of particular isolates.

3.7 Conclusion

In summary, it appears that from the work carried out in this study that when lambs are initially challenged with C. parvum at <1 week of age the resulting disease is more severe than when lambs are challenged at 6 weeks of age. The fact that lambs which are naïve to Cryptosporidium still show mild clinical disease when challenged at 6 weeks of age with a higher challenge dose demonstrates that lambs do not develop complete immunity to disease with increasing age. The results of this study also show that infection with C. parvum does not provide complete protection against a homologous or heterologous challenge later in life. These results do not confirm a previous study which showed that lambs challenged at <1 week of age will be resistant to a homologous challenge at 1 month of age (Ortega-Mora and Wright, 1994) though in the present study any protection provided by prior challenge is masked by the effect of age. The lambs in this study were also challenged with a higher dose at 6 weeks of age whereas the lambs in the study by Ortega-Mora & Wright were given exactly the same challenge dose at 1 month of age. It is possible that had the lambs in this study received an identical challenge dose at 6 weeks of age to the one they received at <1 week of age the results may have been different. There may also be differences in the isolates used in these studies, the study by Ortega-Mora and Wright (1994) used the Moredun isolate (first isolated from a deer calf and then passaged in calves and lambs $4 \times per$ year since 1987) and the present study used two different isolates (BB and LH) both isolated from calves and passaged once in neonatal lambs.

We have also demonstrated in this study that it is possible for different *C*. *parvum* isolates to behave differently in the host. From the results of this study we can report that one isolate (BB) resulted in fewer diarrhoeic samples in infected animals than another isolate (LH) which caused more severe and prolonged diarrhoea in infected animals. We also demonstrated that clinical disease does not directly correlate with oocyst output; animals challenged with isolate BB shed greater numbers of oocysts whilst showing milder clinical signs compared with animals infected with isolate LH. In some cases the oocyst shedding was completely asymptomatic and these asymptomatic animals could, in a field setting, be shedding several million potentially infective oocysts into the environment where susceptible human and animal hosts could then become infected.

Further research in this area could be focused on determining the host factors which effect resistance to disease and why individuals experience different clinical manifestations of disease even when challenged with the same isolate and dose. In this study, there were differences in clinical disease and oocyst shedding between individuals from the same challenge groups. Some animals shed much larger numbers of oocysts than others in the same challenge group and some suffered from diarrhoea while others simply exhibited asymptomatic shedding. Differences in clinical symptoms have been reported in other studies (Jeckel and Williams, 2009; Lindsay et al., 1986; Santin, 2013; Vitovec et al., 2006; Zambriski et al., 2013a) but the reason for these differences is unknown, it is likely that it is due to differences in the immune response of the infected individuals. Understanding more about how and when animals mount an immune response to infection with *Cryptosporidium* will help in the development of suitable control measures.

Papers in preparation and presentations at conferences

Can infection with one genotype of *C. parvum* protect against infection with another - 5th International *Cryptosporidium* and *Giardia* conference - Sweden, 2014

Can infection with one genotype of C. parvum protect against infection with another - Exchange Visit -Brazil, 2014

Chapter 4: Development of a molecular tool to differentiate between the most common species of *Cryptosporidium* in cattle

4.1 Introduction

For the work carried out for this thesis one of the first steps was to develop a quick, easy and reliable method for detection and speciation of *Cryptosporidium* species which may be found in cattle. This was to enable rapid identification of species present in faecal samples from calves and older cattle.

The most common methods for detection of *Cryptosporidium* in farm livestock are microscopy (in veterinary diagnostic laboratories) or by molecular detection (research laboratories); of the two methods molecular detection followed by sequencing or RFLP is much more sensitive and provides additional information (such as species or genotype) than microscopy which can only indicate the presence of the parasite (Chalmers and Katzer, 2013). One of the most common genes targeted for PCR detection of *Cryptosporidium* is the 18S rRNA gene which is located on a multi-copy DNA element meaning that the PCR is more sensitive than those which target a single copy gene for example actin, COWP or hsp70. The 18S rRNA gene is also one of the most commonly used genes for species discrimination and there are many 18S rRNA reference sequences available on the National Centre for Biotechnology Information website (www.ncbi.nlm.nih.gov). When using this gene it is necessary to carry out sequencing or PCR-RFLP to identify species in *Cryptosporidium* positive samples. These methods can be expensive, time consuming and it is very difficult to detect mixed infections as often only the most predominant species in a sample is detected (Cama et al., 2006).

A multiplex PCR test would enable the rapid speciation of a subset of *Cryptosporidium* species and previous attempts have been made to develop such a test (Lindergard et al., 2003; Patel et al., 1999; Santin and Zarlenga, 2009); the most recent of which can detect the four most commonly found *Cryptosporidium* species in cattle (Santin and Zarlenga, 2009). This test amplifies different sized fragments of the actin gene for the different species (*C. bovis* and *C. ryanae* both generate identically sized fragments of 300 bp and

further PCR tests and sequencing would be required to distinguish these species). This defeats the purpose of designing such a test and it would still be very difficult if not impossible to detect mixed infections if both of these species were present.

4.2 Aims

 Develop a multiplex PCR that enables the identification of the four most common *Cryptosporidium* species found in cattle in single and mixed infections as well as identify any non-cattle adapted species

4.3 Materials and Methods

4.3.1 Design of Cryptosporidium species specific primers

Alignments of multiple (between 50 and 100 for each species) 18S rRNA gene sequences for Cryptosporidium species (C. parvum, C. bovis, C. ryanae, C. andersoni, C. ubiquitum, C. hominis, C. muris, C. xiaoi and C. suis) were made using the BioEdit Software Program (www.mbio.ncsu.edu/bioedit/bioedit.html) to identify representative sequences for each species. ClustalW (www.ebi.ac.uk/Tools/msa/clustalw2) alignments of the representative sequences were made and regions for species specific primers were identified which would produce amplicons of different sizes for each species when used with the 18S rRNA reverse primer (AL3032) (Xiao et al., 2000a). The location of the primers used are shown in Figure 4-1; the first set of primers (AL1598 and AL3032) are those described by Xiao et al., 2000a and amplify a 840 bp genus specific fragment, the second primer set (CaF and AL3032) amplify a 625 bp fragment from C. andersoni; the third set (CrF and AL3032) amplify a 415 bp fragment from C. ryanae; the forward primer in this set contains a degenerate base (R) where the C. ryanae sequence contains either an 'A' or a 'G'; the fourth set (CphF and AL3032) amplify a 305 bp fragment from C. parvum, C. hominis and C. suis which can be later distinguished with an additional PCR if desired and the final set of primers (CbF and AL3032) amplify a 241 bp fragment from C. bovis. The primer sequences are shown in Table 4-1.

			10) 2	20	3()	40	5	0
		<u></u>	.		.	••••				
С.	parvum	GGAAGGG	ΤG	TATTTATTAC	G ATAAAG	AACC	AATATAATT	-	-GGTGACTCA	
с.	andersoni		••	•••••	• • • • • • •	• • • •	GAGC	_	T	
С.	bovis		••	•••••	• • • • • • •	••••	.GTT	—		
С.	ryanae		••	•••••	••••••	••••	T.TT	т	т	
С.	nominis		••	•••••	••••••	••••	•••••••	_		
С. С	suis									
с. С	wizoi						 mm	_		
с. С	muris			•••••	• • • • • • • •		GAGC	_		
с.	1111115		•••	•••••	• • • • • • •	1	GAGC		•••••	
			60) -	70	8()	90	10	0
			•		.	•••		1		-
C.	parvum	TAATAACI	TT	ACGGATCACA	A TTAA	ATGT	GACATATCA	T	TCAAGTTTCT	
C.	andersoni			G.	CTCTG	C				
C.	bovis		• •		•••					
C.	ryanae		•••	• • • • • • • • • •	. C	• • • •		•		
C.	hominis		•••	•••••	A.T	••••	••••••	•		
C.	suis		••	•••••	TTTA	••••		•	•••••	
C.	ubiquitum									
C.	xiaoi	•••••	••	•••••	• • •	••••	••••••	•	•••••	
C.	muris	•••••	••	•••••G••	CTCTG	C	••••••	•	••••••	
			1 1 0			1.0	` 1	4.0	1 -	~
		I	TT(20	130		.40	15	0
C	D 2 F 1717 m		• 'AC							
с. С	andersoni	GACCIAIC	AG	CITIAGACGO	J IAGGGI	AIIG	GCCIACCGI	G	GCAAIGACGG	
с. С	bowie	•••••	••	••••••	• • • • • • •	••••		•	π	
с. С	rvanae		••	••••••	• • • • • • •	••••	••••••	•	Ψ	
с.	hominis		•••		••••••	••••		•		
с.	suis									
с.	ubiquitum									
c.	xiaoi								T	
C.	muris									
			160) 17	70	180) 1	90	20	0
			•	••••	.	•••				
С.	parvum	GTAACGGG	GA	ATTAGGGTT	C GATTCC	GGAG	AGGGAGCCT	G	AGAAACGGCT	
С.	andersoni	•••••	•••	•••••	• • • • • • •	••••	•••••	•	••••••	
С.	DOVIS	•••••	•••	•••••	• • • • • • •	••••	•••••••	•	•••••••••	
С. С	ryanae	•••••	•••	•••••	• • • • • • •	••••	••••••	·	••••••	
с. С	nominis	•••••	•••	••••••	• • • • • • •	••••	•••••	•	•••••	
с. с	ubiquitum	•••••	•••	••••••	• • • • • • •	• • • •	••••••	•	••••••	
с.	xiaoi	•••••	••	•••••	• • • • • • •	••••		•		
с.	muris		•••					•		
•••			•••					•		
			210) 22	20	230) 2	40	25	0
			•		.					
C.	parvum	ACCACATC	TA	AGGAAGGCAG	G CAGGCG	CGCA	AATTACCCA	A	TCCTAATACA	
C.	andersoni		••	· · · · · · · · · · · ·	• • • • • • •	<u></u>			<u>G.C</u>	
C.	bovis		•••	••••••	• • • • • • •	••••		•	•••••	
C.	ryanae	•••••	••	•••••	• • • • • • •	••••		•	•••••	
C.	hominis	•••••	•••	••••••	• • • • • •	••••	•••••	•	•••••	
С.	suis	•••••	••	••••••	• • • • • • •	••••	••••••	•	•••••	
С.	ubiquitum	•••••	•••	•••••	• • • • • • •	••••	•••••	•	•••••	
C.	X1201	• • • • • • • •	•••	••••••	• • • • • • •	••••	•••••	•	•••••	
С.	murıs	•••••	••	•••••		••••	••••••	•	G.C	

		260) 270	28	0 290	300
C.	parvum	GGGAGGTAGT	GACAAGAAAT	AACAATACAG	GACTT-TT-T	GGTTTTGTAA
С.	- andersoni				.G.CAA-C	C
C	bovis				A C - A - C	
c.	rvanao				AC C - A - C	
с. С	bominic	••••••••••	•••••••••	••••••		•••••
с. С		••••••••••	•••••	•••••	· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·
C.	suis	•••••	•••••••••	•••••	•••••	A
С.	ubiquitum	••••••••••	••••••••••	•••••	TAAA.	A
С.	x1201	••••••	•••••••••	•••••	ACA-C	
C.	muris	••••••	•••••••••	••••••	.G.CAA-C	C
		31() 320) 33	0 340	350
		••••		••••	••••	
C.	parvum	TTGGAATGAG	TTAAGTATAA	ACCCCTTTAC	AAGTATCAAT	TGGAGGGCAA
C.	andersoni	•••••	•G		G	••••••••••••••••••••••••••••••••••••••
C.	bovis	••••	•••••	A		•••••
C.	ryanae			A		••••••••••••••••••••••••••••••••••••••
C.	hominis					••••••••••••••••••••••••••••••••••••••
с.	suis			C.		••••••••••••••••••••••••••••••••••••••
С.	ubiquitum					
с.	xiaoi	•••••		Д		••••••
с. С	muric	•••••	с	••••••••	с	•••••••
с.	muiis	•••••	•G••••••	•••••	G	••••••
		360) 370) 38	0 390	100
		500			0 590	400
~						
C.		GICIGGIGCC	AGCAGCCGCG	GIAAIICCAG	CICCAAIAGC	GIAIAIIAAA
С.	andersoni	•••••	•••••	•••••	•••••	•••••
С.	bovis	•••••	•••••	•••••	•••••	•••••
С.	ryanae	•••••	•••••	•••••	•••••	•••••
C.	hominis	•••••	•••••	•••••	•••••	••••
C.	suis	•••••	•••••	•••••	••••	•••••
C.	ubiquitum	· · · · · · · · · · ·		•••••	•••••	· · · · · · · · · · · ·
C.	xiaoi					
C.	muris					
		410	9 420) 43	0 440	450
C.	parvum	GTTGTTGCAG	TTAAAAAGCT	CGTAGTTGGA	TTTCTGTTAA	TAATTTATAT
C.	andersoni				GT	AT.AT
C.	bovis			AAT	C	T
C.	rvanae			AAT	–	T
с.	- hominis					
С.	suis			Т		
С.	ubiquitum					
с. С	viaoi		•••••	ΔΔΨ	с —	Т
с. с	murie	•••••••	•••••	••••••••	Ст.	
с.	marro	•••••	•••••	••••••	•••••••••	
		460) <u>47</u> (1 48	0 490	500
		-100		, <u></u> ,		
c	naritim	•••• ••••		•••• ••••	•••• •••• <u>۵</u> קרק קרק און	•••• •••• <u> </u> <u> </u>
с. С	Parvuii		CNACC 7			
с. с	anuerson1	AC	CAAGG.A	••••T•		T
С. С	DOVIS	.TA.A.	CACG	••••	•••••	•••••
С.	ryanae	.TC.A.GC	TACGG	••••	•••••	•••••
C.	nominis	•••••	•••••••••	••••	•••••	•••••
C.	suis	.T	T	••••	•••••	•••••
C.	ubiquitum	.T	ATT	•••••G•	••••	••••
C.	xiaoi	.TC.A.A.	CACG	••••		•••••
C.	muris	AC	TAAGG	ATT.	CCC	T

		51(520) 530	540) 550
C.	parvum	TATATATTTT	AGTATAT	GAAATTTTAC	TTTGAGAAAA	TTAGAGTGCT
C.	andersoni	ATCTAAA.	AG			
с.	bovis	.T.TG		C		
C	rvanae	<u>т</u> т G		С		
с. С	hominis	• т т	••••• ጥጥጥ	•••••••••••	••••	
с. С	nominitis cui c	۰۰۰±۰±۰۰۰ ۸۳	<u>λ</u> ΨΨ	••••••	••••	••••••••••
с. а	suis	••••AI••••	AII	•••••	••••	
С.	ubiquitum		A1	•••••	••••	•••••
С.	X1201	.T.TG			••••	•••••
С.	muris	AT.TCTAAA.	AG		••••	•••••
		560) 570) 580	590) 600
C.	parvum	T AAAGCAGGC	ATATGCCTTG	AATACTCCAG	CATGGAATAA	TAT-TAAAGA
C.	andersoni	• • • • • • • • • •	.AC		••••	
C.	bovis		TAT			G
С.	rvanae		ТАТ			G
c.	hominis					_
с. С	enie				••••••••	_ 7
с. с	suis ubiquitum	· · · · · · · · · · · · · · · · · · ·		•••••	•••••	- 7
с. а	ubiquitum	••••••	••••••••••	•••••	•••••	· · · ⁻ A · · · ·
С.	x1a01	•••••	TAT	•••••	•••••••	•••-•••G••
С.	muris	••••••	.AC	••••••	•••••	
		610	0 620) 630	0 640) 650
C.	parvum	TTTTTATCTT	TCTTATTGGT	TCTAAGATAA	GAATAATGAT	TAATAGGGAC
C.	andersoni	CG		GC	A.GG.	
C.	bovis	TC.		GA	AG	••••
C.	rvanae	TC.		GA	A	
С.	hominis		. Т			
С.	suis		. Т		Α	
c.	ubiquitum		Ψ		Δ	
с. С	vizoi	 тс	• • • • • • • • • •	сл	Λ	•••••
с. С	x1a01					•••••A••
с.	muiis	CG	•••••	••••G••C••	A.G	•••••
		CC) <i>с</i> ло			0 700
		660	5 670		5 690	J /00
-						
С.	parvum	AGTTGGGGGGC	ATTTGTATTT	AACAGTCAGA	GGTGAAATTC	TTAGATTTGT
С.	andersoni	••••••••	C	C	•••••	•••••
C.	bovis	· · · · · · · · · · ·	• • • • • • • • • • •	•••••	••••	• • • • • • • • • • •
C.	ryanae	· · · · · · · · · · ·	• • • • • • • • • • •		•••••	•••••
C.	hominis	· · · · · · · · · · ·			••••••	•••••
C.	suis	· · · · · · · · · · ·			· · · · · · · · · · · · · · · · · · ·	•••••
C.	ubiquitum					
C.	xiaoi	· · · · · · · · · · ·			· · · · · · · · · · ·	
C.	muris		C	C		
		71() 720) 73() 74(0 750
		1 1				
С	parvum	TAAAGACAAA	CTAATGCGAA	AGCATTTCCC	AAGGATGTT	ͲϹΑͲͲΑΑͲϹϫ
с.	andersoni	C	C			
с. С	hovie	••••••••	с	•••••	••••••	•••••
с. С	20123	•••••		•••••	••••••	•••••
с. С	ryande bominis	•••••	••••	•••••	•••••	•••••
с. С	nominis	•••••	•••••	•••••	••••••	•••••
С.	suis	•••••	••••G	•••••	•••••	•••••
C.	ubiquitum	••••••	G	•••••••	•••••••••	•••••
C.	xiaoi	•••••	C	•••••	••••••	••••
C.	muris	. G	C		· · · · · · · · · · · ·	

		760) 77() 780) 790	008 0
С	narvum					
с. С	parvan	1101110011110	11100000110	0///0//0//10	10111100010	0111010111111
с. с		•••••	•••••	••••••	•••••	•••••
C.	DOVIS	•••••	•••••	••••••	•••••	•••••
С.	ryanae	••••	•••••	••••••••	••••••••	•••••
C.	hominis	•••••	•••••	••••••	•••••	••••
C.	suis	· · · · · · · · · · ·	•••••	•••••••	•••••	•••••
C.	ubiquitum	· · · · · · · · · · ·	•••••	••••••••	· · · · · · · · · · -	
C.	xiaoi		•••••	· · · · · · · · · · · · · · · · · · ·		
C.	muris					
		810) 820) 83() 840	0
		810) 82() 83(<u> </u>) 840	0
c.	parvum	81(CCATAAACTA) 820 TGCCAACTAG) 83(AGAT <mark>T</mark> GGAGG) 84(TTGTTCCTTA	O CTCCT
с. с.	parvum andersoni	810 CCATAAACTA) 820 TGCCAACTAG G) 830 AGATTGGAGG) 84(TTGTTCCTTA	0 CTCCT
с. с. с.	parvum andersoni bovis	810 CCATAAACTA) 820 TGCCAACTAG G) 830 AGATTGGAGG) 84(TTGTTCCTTA	0 CTCCT
с. с. с.	parvum andersoni bovis ryanae	810 CCATAAACTA) 820 TGCCAACTAG G) 830 AGATTGGAGG) 84(TTGTTCCTTA	0 CTCCT
с. с. с. с.	parvum andersoni bovis ryanae hominis	810 CCATAAACTA) 820 TGCCAACTAG G) 83(AGATTGGAGG) 84(TTGTTCCTTA	
с. с. с. с. с.	parvum andersoni bovis ryanae hominis suis	810) 820 TGCCAACTAG G) 83(AGATTGGAGG) 840 TTGTTCCTTA	
с. с. с. с. с.	parvum andersoni bovis ryanae hominis suis ubiquitum	810) 820 TGCCAACTAG G) 83(AGATTGGAGG) 840 TTGTTCCTTA	O CTCCT ····· ·····
с. с. с. с. с. с.	parvum andersoni bovis ryanae hominis suis ubiquitum xiaoi	810) 820 TGCCAACTAG G) 83(AGAT TGGAGG) 84(TTGTTCCTTA	

Figure 4-1: Positions of species specific primers on the 18S rRNA gene plus the location of the 18S rRNA internal primers.

The primers designed amplify a 625bp fragment for <u>C. andersoni</u>, 415bp fragment for <u>C. ryanae</u>, 305bp fragment for <u>C. parvum</u>, <u>C. hominis</u> and <u>C. suis</u> and a 241bp fragment for <u>C. bovis</u>. Accession numbers for each species were: C. andersoni (FJ463187), C. ryanae (JN400880), C. parvum (AF164102), C. bovis (HQ179572), C. hominis (DQ286403), C. suis (AB694726), C. xiaoi (GU014553.1), C. muris (AF381167.1) and C. ubiquitum (KC608029.1).

Drimor		Fragment	Species
Primer	Filliel Sequence 5-5		Detected
AL1687 (EF)	TTCTAGAGCTAATACATGCG	1270	Genus
AL1691 (ER)	CCCATTTCTTCGAAACAGGA	1370	Specific
AL1598 (IF)	GAAGGGTTGTATTTATTAGATAAAG	940	Genus
AL3032 (IR)	AAGGAGTAAGGAACAACCTCCA	040	Specific
CaF	GCAAATTACCCAATCCTGAC	625	C andersoni
AL3032 (IR)	AAGGAGTAAGGAACAACCTCCA	025	C. andersom
CrF	TGTTAATTTTTATATACAAT <u>R</u> *CTACGG	415	C ryanae
AL3032 (IR)	AAGGAGTAAGGAACAACCTCCA	415	O. Iyanac
CphF	AGAGTGCTTAAAGCAGGCATA	305	C parvum
AL3032 (IR)	AAGGAGTAAGGAACAACCTCCA	303	C. parvum
CbF	CTTCTTATTCCTTCTAGAATAAAAATG	2/1	C bovis
AL3032 (IR)	AAGGAGTAAGGAACAACCTCCA	241	C. DOVIS

Table 4-1: Primer sequences and size of amplicon produced with each primer pair

*The degenerate base in the C. ryanae primer is shown in bold and underlined

4.3.2 Faecal Sample Collection and DNA extraction

Faecal samples (n=498) were collected from 308 adult cattle and 190 calves from 21 beef farms in Aberdeenshire and Caithness, Northeast Scotland in Spring 2011 as part of another study. The faecal samples were collected by the farmers and sent to the Moredun Research Institute by their vet practices.

DNA was extracted from the calf faeces processing with a QIAamp DNA Stool Minikit (Qiagen) according to the manufacturer's instructions except that $10 \times$ freeze-thaw cycles in liquid nitrogen were carried out after adding 200 µg sample to lysis buffer (provided in the kit).

Adult faeces were subjected to a sedimentation step prior to DNA extraction. Briefly, 3 g of faeces was added to a 50 ml tube (A) and resuspended in 25 ml dH_2O before vortexing vigorously. The solution was left to settle for 5 mins and the supernatant removed and added to a second tube (B). The pellet in tube A was resuspended with 25 ml dH_2O and the settlement step repeated, the supernatant was removed from tube A and added to tube B. The tube containing the supernatant was centrifuged at 1000 x g for 5 mins and the supernatant discarded, 200 µg of the resultant pellet was added to lysis buffer and subjected to 10 × freeze-thaw cycles in liquid nitrogen followed by processing by QIAamp DNA Stool Minikit (Qiagen) according to the manufacturer's instructions.

All 498 samples were tested for the presence of *Cryptosporidium* parasite DNA using the 18S genus specific PCR, 4.13% (n=13) of the samples from adult cattle and 60.5% (n=115) of the samples from calves tested positive for *Cryptosporidium*. These samples were used to test the nested species specific multiplex - Polymerase Chain Reaction (nssm-PCR).

4.3.2.1 Testing of individual species specific primers

First round reactions were carried out as described in section 2.6.1 using DNA from known positive samples for each of the four species. Secondary reactions were carried out to test each individual species specific forward primer with the reverse 18S rRNA primer (AL3032). First round PCR products were diluted with 50 µl dH₂O and 1 µl of the dilution used as a template in a reaction containing 2.5 µl 10× PCR buffer (45 mM Tris-HCL pH 8.8, 11 mM (NH₄)₂SO₄, 4.5 mM MgCl₂, 4.4 µM EDTA, 113 µg ml-1 BSA, 1 mM of each of the four deoxyribonucleotide triphosphates) (MacLeod et al., 1999), 0.5 units BioTaq (Bioline, UK) and 10 µM forward and reverse primers. The final volume of each reaction was made up to 25 µl using dH₂O.

Cycling conditions were 3 min at 94°C, followed by 35 cycles of 45 s at 94°C, 45 s on a temperature gradient of 49 - 59°C and 1 min at 72°C. The final extension was 7 min at 72°C. Second round PCR products were visualised following electrophoresis on a 1.5% agarose gel stained with GelRedTM (Biotium, UK) on an Alphalmager 2200.

4.3.2.2 Field Samples

Primers were tested on a panel of DNA samples which included: known positive samples for *C. parvum*, *C. andersoni*, *C. bovis* and *C. ryanae* from naturally infected cattle (n=128). Results from the nssm-PCR were compared with sequence data for some of these samples.

4.3.2.3 Optimisation of PCR cycling conditions

The first round of the nssm-PCR was carried out in a 25 µl reaction as described in section 4.3.2.1 using 10 µM of each primer and 3 µl of template DNA in single and multiple species combinations. The first round products were diluted with 50 µl dH₂O and 1 µl of the dilution used as template in the 2nd round. Second round reactions were carried out in a 25 µl reaction as described in section 4.3.2.1. Primers were tested on a temperature gradient to determine the optimum annealing temperature. Cycling conditions were 3 minutes at 94°C, followed by 35 cycles of 45 s at 94°C, 45 s on a temperature gradient of 49 -59°C and 1 minute at 72°C. The final extension was 7 minutes at 72°C. Second round PCR products were visualised following electrophoresis on a 1.5% agarose gel stained with GelRedTM (Biotium, UK) on an Alphalmager 2200.

4.3.3 Sequence Analysis

To confirm the results of the nssm-PCR when testing field samples, sequence analysis of the 18S rRNA amplicon was carried out. Sequencing of PCR products was carried out by GATC (GATC, Germany) using their "Light-run" service. Five microlitres of purified PCR product was sent at a concentration of between 20-80 ng/µl with 5 µl of the appropriate primer at a concentration of 5 pmol/µl. In some circumstances, instead of the light-run service, unpurified PCR products were sent to GATC for sequencing using their "Supremerun96" sequencing option which includes PCR purification. With this service 20 µl PCR product was sent with a separate 1.5 ml tube containing 50 µl of the appropriate primer at a concentration of 10 pmol/µl. Sequences were analysed using BioEdit Sequence Alignment Software (www.mbio.ncsu.edu/bioedit/bioedit.html) and Chromas Lite (www.technelysium.com).

4.4 Results

4.4.1 Primer Design

The primer designed for *C. parvum* will also amplify *C. hominis* and *C. suis*, the *C. andersoni* primer will also amplify *C. muris* and the *C. bovis* primer will amplify *C. xiaoi* as the 18S region of these species are very similar. *Cryptosporidium suis* and *C. hominis* have only been found very rarely in cattle (Fayer et al., 2006; Smith et al., 2005) and *C. muris* and *C. xiaoi* have not yet been identified in cattle.

4.4.2 Individual Species PCR

Amplicons of the expected size for each species were obtained when tested in a single species PCR with the species specific forward primer and the 18S rRNA reverse primer.

4.4.3 Optimisation of PCR cycling conditions

An annealing temperature of 56°C was selected as the optimum temperature for all primers in the mix.

4.4.4 Nested Species Specific Multiplex PCR

Amplicons of the expected size for each of the four species (*C. andersoni*, *C. ryanae*, *C. parvum* and *C. bovis* - individually and mixed combinations) were obtained using the nssm-PCR assay (Figure 4-2) from a variety of cattle samples.



Figure 4-2: Detection of species specific amplicons on a 2% agarose gel stained with GelRedTM (Biotium, UK).

Lane 1, 100bp marker XIV (Roche); lane 2, 18S rRNA genus specific amplicon (840 bp); lane 3, *C. andersoni* (625 bp); lane 4, *C. ryanae* (415 bp); lane 5, *C. parvum* (305 bp); lane 6, *C. bovis* (241 bp); lane 7, *C. andersoni* and *C. bovis*; lane 8, *C. andersoni*, *C. ryanae* and *C. bovis*; lane 9, *C. ryanae* and *C. parvum*; lane 10, nssm-PCR molecular marker; lane 11, 100bp marker XIV (Roche). Amplicons in lanes 2-6 were produced using cloned DNA and lanes 7-9 were from naturally infected cattle.

From the 128 samples from naturally infected cattle (1 day - adult) from farms in Aberdeenshire and Caithness the expected amplicons for all four cattle species (C. parvum (n=92), C. andersoni (n=1), C. rvanae (only in mixed infections) and C. bovis (n=13)) were observed as were various combinations of mixed infections (Figure 4-2). Occasionally an additional non-specific band appeared with the amplicon for *C. parvum* but of a different size to the species specific bands. The results of the nssm-PCR were compared with sequence data for the 18S rRNA gene of these samples. In all cases the single species of *Cryptosporidium* identified by nssm-PCR matched the species identified by sequencing. In addition, the nssm-PCR identified mixed infections in 14 of the samples examined. The mixed infections included; C. bovis, C. parvum and C. ryanae (n=1), C. ryanae and C. bovis (n=5), C. parvum and C. bovis (n=1), C. parvum and C. ryanae (n=1) and C. andersoni and C. ryanae (n=6). These mixed infections were not identified by sequencing. Eight of the samples failed to amplify using the nssm-PCR and three produced unreadable trace files when sequenced (Table 4-2).

Sample Type	105 Sequence Result	105 WINDPIEX Result
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. bovis	C. bovis
Calf (7-14 days)	C. bovis	C. bovis
Calf (7-14 days)	C. bovis	C. bovis
Calf (7-14 days)	C. bovis	C. bovis
Calf (7-14 days)	failed sequencing	C. bovis
Calf (7-14 days)	C. bovis	did not amplify
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	did not amplify
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum

 Table 4-2: Comparison of results of nssm-PCR with sequencing of 18S rRNA gene.

 Sample Type 18S Sequence Result

 18S Multiplex Result

Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. bovis	C. bovis
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. bovis	C. bovis
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. bovis	C. bovis
Calf (7-14 days)	C. bovis	C. bovis
Calf (7-14 days)	C. bovis	C. bovis, C. parvum + C. ryanae
Calf (7-14 days) Calf (7-14 days)	C. bovis C. parvum	C. bovis, C. parvum + C. ryanae C. parvum
Calf (7-14 days) Calf (7-14 days) Calf (7-14 days)	C. bovis C. parvum C. parvum	C. bovis, C. parvum + C. ryanae C. parvum C. parvum
Calf (7-14 days) Calf (7-14 days) Calf (7-14 days) Calf (7-14 days)	C. bovis C. parvum C. parvum C. parvum	C. bovis, C. parvum + C. ryanae C. parvum C. parvum C. parvum
Calf (7-14 days) Calf (7-14 days) Calf (7-14 days) Calf (7-14 days) Calf (7-14 days)	C. bovis C. parvum C. parvum C. parvum C. parvum	C. bovis, C. parvum + C. ryanae C. parvum C. parvum C. parvum C. parvum
Calf (7-14 days) Calf (7-14 days) Calf (7-14 days) Calf (7-14 days) Calf (7-14 days) Calf (7-14 days)	C. bovis C. parvum C. parvum C. parvum C. parvum C. parvum	C. bovis, C. parvum + C. ryanae C. parvum C. parvum C. parvum C. parvum C. parvum
Calf (7-14 days) Calf (7-14 days) Calf (7-14 days) Calf (7-14 days) Calf (7-14 days) Calf (7-14 days) Calf (7-14 days)	C. bovis C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum	C. bovis, C. parvum + C. ryanae C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum
Calf (7-14 days) Calf (7-14 days)	C. bovis C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum	C. bovis, C. parvum + C. ryanae C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum
Calf (7-14 days) Calf (7-14 days)	C. bovis C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum	C. bovis, C. parvum + C. ryanae C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum
Calf (7-14 days) Calf (7-14 days)	C. bovis C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum	C. bovis, C. parvum + C. ryanae C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum
Calf (7-14 days)	C. bovis C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum	C. bovis, C. parvum + C. ryanae C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum
Calf (7-14 days)	C. bovis C. parvum C. bovis C. bovis	C. bovis, C. parvum + C. ryanae C. parvum C. parvum
Calf (7-14 days)	C. bovis C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. bovis C. bovis C. bovis	C. bovis, C. parvum + C. ryanae C. parvum C. parvum
Calf (7-14 days)	C. bovis C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. bovis C. bovis C. bovis C. bovis	C. bovis, C. parvum + C. ryanae C. parvum C. parvum
Calf (7-14 days)	C. bovis C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. bovis C. bovis C. bovis C. bovis C. bovis	C. bovis, C. parvum + C. ryanae C. parvum C. parvum
Calf (7-14 days)	C. bovis C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. bovis C. bovis C. bovis C. bovis C. bovis C. bovis	C. bovis, C. parvum + C. ryanae C. parvum C. ryanae + C. bovis C. ryanae + C. bovis C. ryanae + C. bovis
Calf (7-14 days) Calf (7-14 days)	C. bovis C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. parvum C. bovis C. bovis C. bovis C. bovis C. bovis C. bovis C. bovis	C. bovis, C. parvum + C. ryanae C. parvum C. ryanae + C. bovis C. ryanae + C. bovis C. ryanae + C. bovis C. ryanae + C. bovis C. bovis

Calf (7-14 days)	C. suis-like	C. parvum + C. ryanae
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	failed sequencing	did not amplify
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. bovis	C. bovis + C. ryanae
Calf (7-14 days)	C. bovis	C. bovis
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	failed sequencing	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum

Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Calf (7-14 days)	C. parvum	C. parvum
Adult	C. parvum	C. parvum
Adult	C. andersoni	C. andersoni
Adult	C. parvum	did not amplify
Adult	C. parvum	did not amplify
Adult	C. suis-like	did not amplify
Adult	C. andersoni	did not amplify
Adult	C. andersoni	C. andersoni + C. ryanae
Adult	C. andersoni	C. andersoni + C. ryanae
Adult	C. andersoni	C. andersoni + C. ryanae
Adult	C. andersoni	C. andersoni + C. ryanae
Adult	C. andersoni	C. andersoni + C. ryanae
Adult	C. andersoni	did not amplify
Adult	C. andersoni	C. andersoni + C. rvanae

4.5 Discussion

In this chapter the development of a multiplex PCR test which allows the discrimination of four of the most commonly detected *Cryptosporidium* species of cattle (some of which are or may pose a risk to human health) is described. The test allows the differentiation of these species by generating distinct amplicons of the 18S rRNA gene: C. andersoni (625 bp), C. ryanae (415 bp), C. parvum (305 bp) and C. bovis (241 bp) in addition to a Cryptosporidium genus specific amplicon (840 bp). Currently the most popular method of species identification in Cryptosporidium positive samples is gene sequencing of the 18S rRNA gene. This method is very sensitive and can detect a large range of Cryptosporidium species and genotypes but is limited when mixed infections occur as it has been shown that in a mixed sample either the dominant *Cryptosporidium* species is preferentially amplified by PCR meaning that only one species will be identifiable by sequencing or the sequence is unreadable (Cama et al., 2006; Santin et al., 2004; Silverlas et al., 2010). The multiplex PCR described here allows the detection of mixed infections of up to four different common cattle species of *Cryptosporidium* and is an excellent method for detecting underlying infections of zoonotic *C. parvum* in calves which may have higher burdens of the less-pathogenic *Cryptosporidium* species.

In addition to the easy and quick visual differentiation of the four common cattle species other species of *Cryptosporidium* may be identified by sequencing of the 18S rRNA gene when a fragment responding to the *Cryptosporidium* genus specific amplicon (840 bp) is amplified with no species specific amplicons. At present, using this method, it is not possible to distinguish between *C. parvum*, *C. hominis* and *C. suis* however, as *C. hominis* and *C. suis* have rarely been detected in cattle (Park et al., 2006; Smith et al., 2005) there is, at this stage, no real need to distinguish between these two species. If necessary these species could be differentiated using a further PCR or by sequence analysis.

The advantages of the PCR protocol described here over other methods used for species discrimination (PCR-RFLP, sequencing) are: (1) the ability to detect mixed infections which may be missed by other methods; (2) the ability to identify other *Cryptosporidium* species by further sequencing when none of the cattle adapted species are detected; (3) the ability to test large numbers of

samples quickly and easily, and; (4) the ability to detect subclinical infections of the zoonotic species *C. parvum* in animals shedding high numbers of non-zoonotic species.

As well as being used extensively throughout this PhD thesis this molecular tool has also been used to identify the species of *Cryptosporidium* present in faeces from adult cattle and their calves as part of a whole catchment study in the North East of Scotland (Wells et al., 2015).

Papers in preparation and presentations at conferences

1. Thomson, S., Innes, E.A., Jonsson, N., Katzer, F., Development of a Nested Species Specific Multiplex PCR for the detection of bovine adapted *Cryptosporidium* species. (Submitted)

Chapter 5: Cryptosporidium species and subtypes found in calves from birth to one year

5.1 Introduction

Cryptosporidium parvum is one of the most important causes of neonatal enteritis in calves not only in the UK but worldwide (Mosier and Oberst, 2000). It is known that cattle are commonly infected with four distinct species of *Cryptosporidium* and that infection with these species tends to follow an agerelated distribution with *C. parvum* being predominantly found in pre-weaned calves, *C. bovis* and *C. ryanae* in older calves (<2 months) and *C. andersoni* in adult cattle (Fayer et al., 2006; Santin et al., 2004). There are some exceptions to this with *C. bovis* and *C. ryanae* being found in pre-weaned calves in some countries (France and Sweden) (Rieux et al., 2013; Silverlas et al., 2010) although these species have not been associated with clinical disease (Fayer et al., 2008; Fayer et al., 2005).

Recent advances in typing and subtyping methodologies has allowed more detailed studies into the dynamics of *C. parvum* infection in cattle and humans. The most common subtyping method is sequencing of the 60-kDa glycoprotein (*gp60*) gene, so far, at least 14 subtype families for *C. parvum* have been identified (IIa-IIo (there is no IIj)) (Wang et al., 2014a). Some of the subtype families appear to be exclusively anthroponotic (IIc and IIe) (Cama et al., 2008; Hijjawi et al., 2010; Sulaiman et al., 2005) and others are major zoonotic subtypes (IIa and IId) (Alves et al., 2001; Alves et al., 2003). The remaining subtype families have been found less commonly in animals and humans throughout the world. It has been demonstrated previously that most farms have one predominant *gp60* subtype within their calves which persists over time (Brook et al., 2009).

At present there is limited knowledge of the species and subtypes of *Cryptosporidium* found in cattle on Scottish farms. Particularly little is known about the shedding profiles (age and duration) of the different species found in cattle. Until more is understood about the species and subtypes which calves shed at different times throughout their lives it is difficult to develop effective strategies to control the parasite. Knowing when an animal is likely to pick up

the infection, and when they will be infective to others is hugely important as this will allow targeted control measures to help reduce the impact of cryptosporidiosis in young animals. This part of the thesis aims to address some of these questions by carrying out a detailed longitudinal study in calves.

5.2 Aims

- 1) Identify the species of *Cryptosporidium* that are shed by calves over the initial six weeks of life and then again at set points over one year
- 2) Determine the lengths of the intervals over which each of these species are shed
- 3) Determine how the prevalence of each species changes over time
- 4) Identify any periods during which the numbers of shed oocysts are significantly higher than average for the entire shedding period

5.3 Materials and Methods

5.3.1 Sample Collection

Faecal samples were collected from 25 calves on a dairy farm in Midlothian from the day of birth until six weeks of age, three times per week (Monday, Wednesday and Friday) as described in section 2.1.1. The calves were kept under the normal working conditions of this farm with regards to housing (kept in groups of -20 in straw bedded pens), feeding (fed milk replacer with *ad libitum* access to pelleted calf feed, hay and water) and veterinary treatment. During this time none of the calves were treated specifically for scour or *Cryptosporidium*. The calves were kept with their mothers for one or two days then moved to a small group (5-8 calves) for one week then moved to a larger indoor straw bedded pen with approximately 20 calves in each pen for the remaining five weeks. After this time the calves were moved, and groups mixed, to a different, indoor straw bedded pen. Samples were collected from the same calves again at three, six, nine and twelve months of age where possible. The total number of samples collected was 426. The samples were mixed thoroughly and aliquoted into bijoux tubes and stored at -20°C for further processing.

5.3.2 Sample processing and DNA extraction

Faecal samples from calves up to, and including, 9 months were processed as described in section 2.5.2 and samples from calves aged 12 months as described in sections 2.4.1 and 2.5.1 to obtain DNA for PCR amplification.

5.3.3 PCR amplification of the 18S rRNA gene

All DNA samples were amplified using the 18S PCR described in section 2.6.1 and visualised on a 1.5% agarose gel stained with GelRed (Biotium, UK). All reactions were carried out in triplicate and a sample was considered positive if a band corresponding with ~840 bp was seen in 1/3 triplicates. Several controls, both negative (no template and extraction controls) and positive (*C. andersoni, C. parvum, C. bovis* and *C. ryanae* DNA) were run with each set of reactions. A positive result was only accepted when all negative controls were negative and positive controls were positive.

5.3.4 Species identification by nssm-PCR (nested species specific multiplex-PCR)

For the differentiation of *Cryptosporidium* species an nssm-PCR was carried out using the first round PCR product from the 18S PCR (see section 2.6.1) as a template. The PCR products were visualised on a 1.5% agarose gel stained with GelRed (Biotium, UK). All reactions were carried out in triplicate and a sample was considered positive if a band corresponding with 840 bp and any combination of 625 bp, 415 bp, 305 bp or 241 bp was seen in at least 1/3 lanes. Positive controls (obtained from cloned DNA from oocysts collected from naturally infected cattle as part of another study) for each *Cryptosporidium* species tested for were included as well as extraction controls and no template controls. A positive result was only accepted when all negative controls were negative and all positive controls were positive.

5.3.5 PCR amplification and sequencing of the *gp60* gene for *C. parvum* subtyping

Subtyping of *C. parvum* positive samples was carried out by amplification and sequencing of the *gp60* gene (see sections 2.6.3 and 2.7). The PCR products were visualised on a 1.5% agarose stained with GelRed (Biotium, UK). Reactions were originally carried out in duplicate and if amplification was not successful the reactions were repeated. A sample was considered positive if a band corresponding with ~375 bp was seen. Positive, negative and extraction controls were included for each set of reactions. A positive result was only accepted when all negative controls were negative and all positive controls were positive.

gp60 positive samples were sequenced in the forward and reverse orientation by GATC Biotech (GATC, Germany) as described for "supremerun96" sequencing in section 2.7.3. The sequences were analysed using Chromas Lite (version 2.01 Technelysium) and the subtypes named as according to (Sulaiman et al., 2005).
5.4 Results

5.4.1 Prevalence of *Cryptosporidium* in calves from birth to twelve months

In total 426 faecal samples were collected from calves on the study farm (Table 5-1).

I able J			iples coll	ecteu at		eponii			
Week	Week	Week	Week	Week	Week	3	6	9	12
1	2	3	4	5	6	Months	Months	Months	Months
64	62	63	56	57	57	17	16	17	17

Table 5-1: Number of samples collected at each time point

All 426 samples were tested for the presence of *Cryptosporidium* parasite DNA using the 18S genus specific PCR and the overall percentage of positive samples for the whole study was 59.7% (n=255). In the first six weeks of life each calf tested positive for *Cryptosporidium* on at least six occasions, giving a period prevalence of 100% and the overall percentage of samples positive in calves <6 weeks was 64% (n=232). The earliest observation of oocyst shedding occurred at 2 d, which was observed in three calves. In several of the calves the first observation of oocyst shedding occurred at day 4 (4 animals) or day 6 (4 animals). Peak shedding of *Cryptosporidium* oocysts occurred in week 3 with 100% of the calves tested shedding oocysts. Oocyst shedding tailed off towards the end of week 6. Samples from the calves at 3, 6, 9 and 12 months showed a prevalence of 24%, 63%, 41% and 18.65% respectively (Figure 5-1). Five of the animals did not test positive for *Cryptosporidium* at any point after the initial 6 week period and no animals tested positive at all age groups though two tested positive at <6 weeks, 3, 6 and 9 months.

The percentage of positive samples collected from each calf ranged from 38.9% to 87.5% (Figure 5-2).



Figure 5-1: Shedding profile of *Cryptosporidium* oocysts in calves from birth to 12 months. Percentage of positive samples from calves in the first six weeks of life and at 3, 6, 9 and 12 months of age. The percentage of positive samples at each sampling point (i, ii, iii) is indicated by a red bar.

	Calf Age (Weeks)														Ca	alf Age	(Month	ıs)					
Calf ID		Week 1			Week 2		۱ ۱	Week 3			Week 4	•		Week 5			Week 6	5	3 M	6 M	9 M	12 M	% Positive
2384	N	N	Р	Р	Р	Р	Р	Р	Р	Ν	Р	0	0	Р	Ν	0	Ν	N	Р	0	Р	N	61.1
2385	N	Р	Р	Р	Р	Ν	0	Р	Р	Р	Р	0	0	Р	Р	0	Р	N	N	Ν	Ν	N	61.1
2386	0	N	Р	Р	Р	Ν	Р	N	Р	Ν	Р	Р	0	Р	Р	0	Ν	0	0	0	0		64.3
2387	0	N	Р	Р	Р	Ν	0	Р	Р	Р	Р	Р	0	Р	Р	0	Ν	0	0	0	0		76.9
2388	0	Р	Р	Р	Р	Р	Р	Ν	Р	Р	Р	Р	0	Р	Ν	0	Ν	N	Р	Р	Ν	N	68.4
2389	N	Ν	Р	Ν	N	Р	0	Р	0	Р	Р	0	Ν	Ν	0	Р	0	0	0	0	0		50.0
2390	N	N	Р	N	Р	0	Р	Р	N	Р	Р	0	Ν	Ν	0	0	Ν	N	N	Р	Ν	N	38.9
2393	N	Р	Ν	Р	Р	0	Р	Р	Р	Р	0	0	Ν	0	N	N	Р	N	N	Р	Р	N	55.6
2394	N	Ν	Р	Ν	Р	0	0	Р	Р	Р	0	0	Р	0	Р	N	Р	N	0	0	0		61.5
2395	N	Ν	Ν	Р	0	Р	Р	Р	Р	Р	0	Ν	Ν	0	Ν	Р	Р	N	0	0	0		53.3
2396	N	N	Ν	Р	Р	Р	Р	Р	Р	Р	0	Ν	Ν	0	N	N	Р	N	0	0	0		50.0
2397	0	Р	Р	Р	Р	Р	Р	Р	Р	0	0	Ν	0	Р	0	Р	Р	Р	N	Р	Р	0	87.5
2398	N	Ν	Р	Р	0	Р	Р	Р	Р	0	Р	Ν	0	0	Ν	0	Р	N	Р	Р	Р	N	64.7
2399	0	0	Ν	Р	Р	Р	Р	0	N	0	N	Р	0	Р	Ν	Р	Ν	N	0	0	0		53.8
2402	Р	Р	Р	Р	Р	Р	Р	0	N	Ν	0	Ν	Ν	N	Ν	N	Ν	0	N	N	Р	N	42.1
2403	Р	Р	Р	Р	Р	Р	Р	0	Р	Ν	0	0	Ν	Р	Ν	0	Ν	N	Р	Р	Р	N	66.7
2404	0	Р	Ν	Р	Р	0	0	0	Р	0	Р	Р	Р	0	Ν	N	Ν	Р	N	Р	Ν	N	56.3
2405	0	Ν	Ν	Р	Р	Р	Р	0	Р	Р	Р	Р	Р	Р	Ν	Р	Ν	N	N	Р	Ν	N	60.0
2407	0	N	Р	Р	Р	Р	0	Р	N	0	Р	Р	Р	Р	Р	Р	Р	N	N	Р	Ν	N	68.4
2409	0	Р	0	0	Р	0	Р	Р	Р	Р	Ν	Ν	Ν	N	Ν	N	Ν	N	0	0	0	0	42.9
2411	0	Р	Р	Р	Р	0	Р	Р	Р	Ν	Р	Ν	Р	Р	Ν	N	Р	N	N	N	Ν	N	55.0
2412	Р	N	Р	Р	0	Р	Р	Р	N	Р	Ν	Р	Р	Р	0	Р	Р	0	N	Р	Р	N	73.7
2413	0	Р	Р	Р	0	Р	Р	Р	Р	Р	Р	Ν	Ν	Ν	Ν	Ν	Ν	0	Ν	Ν	Ν	N	47.4
2415	Ν	Ν	Р	Р	0	Р	Р	Р	Р	Ν	Р	Р	Ν	Ν	Ν	0	Ν	0	Ν	Ν	Ν	Р	47.4
2417	0	Ν	Р	Р	0	Р	Р	Р	Ν	Р	Ν	Р	Р	Ν	Р	Р	Р	0	Ν	Ν	Ν	Р	63.2
% Positive	23.1	41.7	75.0	87.5	94.7	84.2	100.0	90.0	75.0	70.0	77.8	55.6	41.2	63.2	28.6	50.0	45.8	11.8	23.5	62.5	41.2	12.5	

Figure 5-2: Positive and negative samples collected from each calf. Samples collected from each calf at each time point. A positive sample is denoted by a red "P" while a negative sample is denoted by a blue "N", "0" indicates that a sample was not collected from that calf at on that day. The percentage of positive samples at each time point are shown at the bottom of each column while percentage of positive samples from each individual is shown in the right-hand column.

5.4.2 Speciation of *Cryptosporidium*-positive samples by nssm-PCR

All *Cryptosporidium*-positive samples were speciated using a nssm-PCR (Chapter 4). Three species were identified (*C. parvum*, *C. bovis* and *C. ryanae*) in 228, 21 and 12 samples respectively. In four samples the species present could not be determined. Fourteen mixed infections were identified: these were *C. parvum* and *C. bovis* (n=7) and *C. bovis* and *C. ryanae* (n=7). In this study *C. parvum* was the predominant species detected in calves <6 weeks old with seven calves showing evidence of mixed infections. Mixed infections of *C. parvum* and *C. bovis* were detected between three and five weeks and *C. parvum* and *C. ryanae* were seen only in week 6. The species shed over the initial six week period and at the other age groups are shown in Figure 5-3.



Figure 5-3: Prevalence of *Cryptosporidium* **species in each age group of cattle.** Percentage of each species of *Cryptosporidium* at each time point.

To confirm the results of the nssm-PCR a selection of samples were sent for sequencing by GATC (see section 2.7) and all sequenced samples matched the result of the nssm-PCR.

5.4.3 Subtyping of C. parvum by gp60 PCR

To determine the *gp*60 subtypes *C. parvum* positive samples were subjected to a nested PCR to amplify a fragment of the gp60 gene. Two hundred and eight of these were positive, a sub-set of which, selected from 5 calves which shed *C. parvum* throughout the initial 6 weeks and at 6 months as well as others which were only *C. parvum* positive in the first 6 weeks, were sequenced (n=30). Two different *gp*60 subtypes were identified. In all of the samples from the calves <6 weeks the genotype identified was IIaA19G2R1 and in the samples from the same calves at 6 months the genotype identified was IIaA15G2R1 (Table 5-2).

	Jenetypee deteeted .	all called at all of of	n ugeei	
Calf ID	< 1 Weeks	2-3 Weeks	6 Weeks	6 Months
2384		IIaA19G2R1		
2385		IIaA19G2R1	IIaA19G2R1	
2388	IIaA19G2R1	IIaA19G2R1	IIaA19G2R1	IIAa15G2R1
2393	IIaA19G2R1	IIaA19G2R1		IIAa15G2R1
2395	IIaA19G2R1	IIaA19G2R1		
2397		IIaA19G2R1		IIaA15G2R1
2398	IIaA19G2R1	IIaA19G2R1		
2404	IIaA19G2R1	IIaA19G2R1	IIaA19G2R1	IIaA15G2R1
2405	IIaA19G2R1	IIaA19G2R1	IIaA19G2R1	IIaA15G2R1
2407	IIaA19G2R1	IIaA19G2R1	IIaA19G2R1	
2417	IIaA19G2R1	IIaA19G2R1	IIaA19G2R1	

Table 5-2: gp60 genotypes detected in calves at different ages.

5.5 Discussion

Although many prevalence studies (longitudinal and point-prevalence) investigating *Cryptosporidium* in cattle have been carried out worldwide this study is the most in-depth study carried out in the UK and provides detailed information on the species and subtypes of *Cryptosporidium* shed in a group of calves over a 12 month period. The findings of this study are comparable with those found by others investigating species of *Cryptosporidium* in calves of a similar age; Santin et al. (2004) found that *C. parvum* was the predominant species detected in calves <2 months with *C. bovis* and *C. ryanae* being detected in some samples from week 3 onwards, in another study *C. parvum* was the only species detected in calves <4 weeks with *C. bovis* being detected in two samples at week 4 and one at week 5. *Cryptosporidium ryanae* was not detected until week 10 (Santin et al., 2008). A more recent study from France (Rieux et al., 2013) also showed that *C. parvum* was the most prevalent species of *Cryptosporidium* in young calves (7-17 days old) with *C. bovis* first being detected at around week 2 and *C. ryanae* from week 3.

In the present study, when the same calves were examined at 3 months, only two species were detected and there were no mixed infections with 50% of the positive samples being identified as C. bovis and 50% as C. ryanae. These species are the most commonly detected in calves of this age group in other studies including longitudinal and point prevalence studies (Follet et al., 2011; Langkjaer et al., 2007; Ng et al., 2011; Santin et al., 2008; Santin et al., 2004; Silverlas et al., 2010). Few studies have examined calves older than three months and most show that the most commonly detected species in this age group (~6 months) are C. bovis and C. ryanae (Follet et al., 2011; Langkjaer et al., 2007; Santin et al., 2008; Santin et al., 2004; Silverlas et al., 2010) and occasionally C. andersoni (Santin et al., 2004), C. ubiquitum (Follet et al., 2011) or C. suis-like (Langkjaer et al., 2007). In the group of calves studied here we report a more unusual distribution of Cryptosporidium species for the age of cattle, at three months the calves shed only C. bovis and C. ryanae which is as expected however, in the same calves at 6 months old 70% of the positive samples were identified as C. parvum (by both nssm-PCR and sequencing of a fragment of the 18S rRNA gene), 20% as C. bovis and 10% as a mixed infection of C. bovis and C. ryanae. This is very unusual as C. parvum in 6 month old cattle

has only been reported a handful of times, and usually as a small percentage of positive samples.

In one study, which examined faeces from older calves *C. parvum* was detected in 5% of calves between the age of 31 days and 6 months (n=368) but it is not stated at exactly which age the calves were shedding *C. parvum* (Follet et al., 2011) and in another study from the USA (Santin et al., 2008) *C. parvum* was found in one calf at 6 months. In the present study, when the same calves were sampled at 9 months of age, no *C. parvum* positive samples were detected. Forty-three percent of the positive samples were identified as *C. bovis*, 14% *C. ryanae* and 29% as a mixed infection of *C. bovis* and *C. ryanae*. Again this ties in with other studies done on this age group of cattle (Langkjaer et al., 2007; Santin et al., 2008; Santin et al., 2004).

The first detection of *Cryptosporidium* positive samples occurred at day two of age in three calves with the majority of calves producing a positive sample between day four and six of life. This early detection indicates that calves pick up the parasite very soon after birth and while it might seem unlikely that calves can become infected and be shedding oocysts in such a short period of time the prepatent period of *Cryptosporidium* in calves is known to be 2-7 days (Tzipori et al., 1983). The first born calves started to shed oocysts at 5-7 days while later born calves began shedding between 2-4 days. This may be an indication that as calving progresses there is a build-up of environmental contamination which means later born calves receive a higher infective dose and therefore begin to shed oocysts sooner. It is known that the prepatent period for *Cryptosporidium* is also dependant on infective dose (Tzipori et al., 1983).

The previously reported age distribution of *Cryptosporidium* species in cattle has been confirmed again in this study with the majority of calves under 6 weeks of age being infected with *C. parvum*. However, in this study we also report the occurrence of *C. bovis* and *C. ryanae* in this age group, which is less common. We also show evidence of mixed infections within this age group and others (6, 9 and 12 months). It has been previously shown that even when using sensitive molecular methods to detect *Cryptosporidium* species in faecal samples, mixed infections can be missed because the predominant species is preferentially amplified (Cama et al., 2006). The detection method used in this study (nssmPCR) is much more effective at detecting mixed infections than standard 18S PCR, this accounts for the increased detection of mixed infections in these calves compared to similar studies. This is of particular use in terms of public health, when a low level of infection with the zoonotic species *C. parvum* is missed because of a predominant infection with a less pathogenic species such as *C. bovis* or *C. ryanae*.

Recent advances in molecular techniques have enabled further characterisation of some *Cryptosporidium* species, this can help provide further information and aid in the understanding of parasite transmission and pathogenicity. In the present study subtyping of C. parvum delivered some interesting and unusual findings. It has been demonstrated previously that most farms tend to have to have only a single gp60 subtype present (Broglia et al., 2008; Brook et al., 2009) (Santin et al., 2008) particularly where there is little movement of cattle. On the study farm replacement heifers are reared on the farm and the only animal bought in is a bull every couple of years. In contrast, on farms where there is greater cattle movement (buying in stock) more than one subtype may be found (Brook et al., 2009; Follet et al., 2011; Silverlas and Blanco-Penedo, 2013). On the farm in this study it appears that there are two distinct populations (one in the young calves and one in the older calves) despite the fact that very few animals are bought in. The subtype identified from the 6 month old calves is the most commonly detected gp60 subtype in cattle and humans worldwide and has been found in cattle in many different countries including; Belgium (Geurden et al., 2007), Great Britain (Brook et al., 2009; Thompson et al., 2007), Germany (Broglia et al., 2008), Italy (Duranti et al., 2009), The Netherlands (Wielinga et al., 2008), Portugal (Alves et al., 2006; Alves et al., 2003), Romania (Imre et al., 2011) and Spain (Quilez et al., 2008b). This subtype is extremely common and one review article reports that 55.3% of *C. parvum* subtypes in cattle were identified as IIAa15G2R1 (Imre and Darabus, 2011). In contrast, the subtype identified in the samples from the <6 week old calves (IIaA19G2R1) is much less common than IIAa15G2R1 in calves (Couto et al., 2013; Wielinga et al., 2008; Xiao et al., 2007) and humans (Gait et al., 2008; Hadfield et al., 2011). However, it is has been present on this particular farm for many years and is known to be zoonotic (Gait et al., 2008).

The management of cattle on this farm is very good: calves are kept in straw bedded pen in groups of ~20 from birth until they reach 3 months then the groups are mixed and moved to a new straw bedded shed (group sizes of ~40). Once the calves are older (5/6 months) they are moved again and all calves in the same age range are housed in a concrete floored pen with cubicles or turned out depending on the time of year. This could explain the different subtypes being found in different areas of the farm as different age groups of cattle are never mixed or housed in the same pens. It is also possible that there are two subtypes circulating in the calves but that one (IIaA19G2R1) is more pathogenic than the other (IIaA15G2R1) and that infection with this subtype overwhelms the gut and the second subtype is not seen until later in life. It has been demonstrated that in humans clinical disease and shedding of oocysts varies depending of species and subtype of *Cryptosporidium* (Cama et al., 2008) and the same may be true in cattle.

Although some longitudinal studies have been carried out in cattle, most have focused on pre-weaned animals or have been point prevalence studies only looking at a group of animals at one particular time point, this study has investigated the prevalence of *Cryptosporidium* species in a group of animals over a longer period with more frequent sample collections (3/week initially) providing more detailed information on how the prevalence of *Cryptosporidium* in a group of calves changes as the animals get older. Of the few studies which have looked at older calves many report that as the calves' age increases the prevalence of *Cryptosporidium* decreases (Follet et al., 2011; Rieux et al., 2013; Santin et al., 2008; Silverlas et al., 2010). This study reports that prevalence of Cryptosporidium decreases after 6 weeks with peak shedding at 3 weeks with a secondary peak in prevalence at 6 months. This pattern has been observed in two other studies, one in the USA and the other in The Netherlands (Huetink et al., 2001; Santin et al., 2004), the secondary peaks in shedding observed in the other studies were lower than reported here (~25% and 3.3% respectively), this is possibly due to the increased sensitivity of the techniques used in this study compared with previous studies.

5.6 Conclusion

It has been shown from this work that young calves become infected with, and begin shedding, Cryptosporidium very soon (minimum 2 days) after birth and that shedding of oocysts (with and without clinical disease) occurs for at least 6 weeks. It has also been demonstrated that calves can be infected with multiple species of Cryptosporidium simultaneously in the first 6 weeks of life. It was previously assumed that older post-weaned calves and adult cattle were not a significant risk for zoonotic transmission of C. parvum as most studies indicated that once calves were older they were not usually infected with this species. This study shows that even if calves are infected with C. parvum before weaning they can become infected with this species again, although a different subtype, when they are older. This is an important factor when considering the zoonotic implications, it also demonstrates the importance of molecular characterisation and speciation of all *Cryptosporidium* positive samples. Many older studies report "C. parvum-like" oocysts, since there are no morphological differences between C. parvum, C. bovis and C. ryanae this molecular characterisation is very important to make sure that the species is correctly and accurately assigned.

Further subtyping which was carried out in this study showed evidence that multiple subtypes of *C. parvum* may be present on a single farm and that these different subtypes may have different clinical manifestations, this is important from both an animal and public health perspective.

Finally, the techniques used in this study demonstrate the need to use the most sensitive methods available to detect oocysts and assign species particularly in the case of older animals where a predominant infection with a non-pathogenic species may be hiding a lesser infection with a pathogenic species such as *C*. *parvum*.

Papers in preparation and presentations at conferences

- 1. Thomson, S., Wells, B., Harvey, C., Jonsson, N., Innes, E.A., Katzer, F., Species and genotypes of *Cryptosporidium* in calves from birth to 1 year. (In. Prep)
- 2. Species and genotypes of *Cryptosporidium* found in calves from birth to one year 2nd
- International apiCOWplexa Conference Turkey, 2013

^{3.} Shedding of zoonotic Cryptosporidium parvum by dairy calves from birth to six months - CoZee Conference - Scotland, 2012

Chapter 6: Transmission of *Cryptosporidium* from dam to calf

6.1 Introduction

Many different routes have been proposed for the transmission of C. parvum to calves (Figure 6-1) and one suggestion is that adult cattle may be the source of infection for neonatal calves. Calves tend to become infected with Cryptosporidium shortly after birth, within 24-72 h (Ramirez et al., 2004; Xiao and Herd, 1994) therefore a likely source for infection would be their dams as the calves are in close contact with them immediately after birth. A few studies have been carried out to test this hypothesis where the presence of Cryptosporidium in the adult faecal samples was detected using microscopy or ELISA (to detect Cryptosporidium antigens in faecal supernatant using an anti-Cryptosporidium antibody) and from this it was assumed that the *Cryptosporidium* in the adult animals were the source of infection for the neonatal animals. However, the methods used in these studies were unable to identify the species of *Cryptosporidium* only that there were oocysts present. Now that it is known that there are many different species of Cryptosporidium and that cattle may be infected with four main species, the question of the source of *C. parvum* infection for neonatal animals needs to be clarified.



Figure 6-1: Possible routes for transmission of *Cryptosporidium* **to calves.** Many different routes for transmission of *Cryptosporidium* to young neonatal livestock have been suggested, the most common of these are transmission from dam to calf at or shortly after birth, from the environment (cattle shed or contaminated pasture), direct contact with contaminated faeces of another host or from contaminated drinking water supplies.

In adult cattle excretion of oocysts tends to be low and intermittent and there are studies to demonstrate this (Table 6-4). One reason for this may be that the methods used are not sensitive enough to detect oocysts in adult cattle faeces. Studies on sheep suggested that dams were the likely source of infection to their young due to a periparturient rise in oocyst excretion (Ortega-Mora et al., 1999; Xiao et al., 1994) leading to the suggestion that the same may be true of cattle. However this evidence was based on the assumption that the oocysts shed were *C. parvum* and at this time many of the other species which we now know infect farm livestock had not been identified. More recent studies show no evidence for a periparturient rise in oocyst shedding in cattle (Atwill and Pereira, 2003; Sturdee et al., 2003) and dams as a source of infection for calves were suggested to be of minor importance (De Waele et al., 2012) except for the transmission of *C. bovis* in older calves (Kvac et al., 2011). Another study (Faubert and Litvinsky, 2000) showed that a periparturient rise in oocyst shedding in cattle does occur and that cows are a likely source of infection for their calves.

A more probable source of infection for young calves is environmental contamination (from other calves or hosts); Cryptosporidium oocysts are highly resistant to heat, cold, disinfectants and are very difficult to eliminate (Robertson et al., 1992). It is thought that differences in calf management practices such as housing can have an effect on infection rates but some of the results are conflicting. In one study of 2024 calves from 248 farms over a two year period, it was shown that housing calves with their dams decreased the risk of *C. parvum* infection while calves removed from their dam and kept in individual pens were at higher risk of infection (Duranti et al., 2009) hinting that dam to calf transmission is less important than calf to calf or via environmental transmission. In contrast, other studies have shown that the opposite is true (Quigley, III et al., 1994; Quigley, III et al., 1995) or that calf housing (individual or group penning), has no effect on Cryptosporidium infection (Castro-Hermida et al., 2002) though calves housed on straw rather than concrete were at higher risk of infection regardless of whether or not they were housed individually or in a group. It is also possible for infection to be passed between calves in a particular pen (O'Handley et al., 1999).

One reason for the apparent low level or lack of shedding of *Cryptosporidium* oocysts by adult animals (cattle and sheep) may be due to the lack of sensitivity of the methods used in processing the faecal samples. Faecal samples from adult animals, particularly cattle are much larger in volume than those from calves and contain more fibrous material meaning that detection of oocysts is much harder. In many studies when negative samples have been detected it is possible that these samples are actually positive but have been missed by the techniques used. In the first part of this study several techniques were assessed to determine the most effective method of processing adult cattle faeces.

6.2 Aims

- 1. Develop a more sensitive method for oocyst concentration and extraction from adult cattle faeces
- 2. Assess the shedding profile of different *Cryptosporidium* species found in dams throughout gestation and after calving
- 3. Determine the *Cryptosporidium* species found in the calves born to cows in the study
- 4. Determine the subtypes of *C*. *parvum* in the same dams and calves

6.3 Materials and Methods

6.3.1 Optimisation of oocyst concentration and DNA extraction from adult cattle faeces

6.3.1.1 Spiking Experiment

A *Cryptosporidium* negative faecal sample (as determined by 10 × DNA extractions and PCR) was obtained from a one year old Charolais bull from the University of Glasgow, School of Veterinary Medicine. The animal was born at the veterinary school and had never had access to grazing and had limited access to other cattle and livestock. To test each oocyst concentration method (direct DNA extraction, acid flocculation (AF), salt flotation (SF) and combined acid flocculation and salt flotation) 0, 5, 10, 100 and 1000 *C. parvum* oocysts (obtained from experimentally infected calves) per gram were added to each sample as shown in Table 6-1.

Table 6-1: Methods tested for optimisation of oocyst concentration before DNA extraction from cattle faeces.

			Method	
No. of oocysts	Acid	Salt	Acid Flocculation and	No Prior
per gram	Flocculation	Flotation	Salt Flotation	Concentration
0	50 g	3 g	50 g	250 mg
5	50 g	3 g	50 g	250 mg
10	50 g	3 g	50 g	250 mg
100	50 g	3 g	50 g	250 mg
1000	50 g	3 g	50 g	250 mg

Number of oocysts spiked per gram of faeces and volume of faeces used for each method.

6.3.1.2 Field Samples

Using faecal samples (n=205) collected from adult dairy cattle as part of this study, AF followed by DNA extraction was compared with AF/SF followed by DNA extraction.

6.3.1.3 Acid Flocculation

The faecal sample was well mixed and 50 g added to a 1 L cylinder with 600 ml water and 7 ml 2% sulphuric acid. The sample was put on a magnetic stirrer and thoroughly mixed for 5 mins then left to settle for approximately 5 mins or until

a clear line was visible between the sediment and supernatant, after which the supernatant was removed by pipette, collected in two 250 ml bottles and centrifuged for 20 mins at 1000 x g. The supernatant was discarded and 6 ml H₂0 was added to the pellet, mixed and put into a 15 ml tube, which was centrifuged for 5 minutes at 3000 x g. The supernatant was discarded and the pellet retained for further processing by either DNA extraction (6.3.1.5) or salt flotation (6.3.1.4).

6.3.1.4 Salt Flotation

A 3 g faecal sample or pellet from acid flocculation was added to 8 ml saturated salt solution and thoroughly mixed by vortexing and 2 ml dH₂O trickled on top of the salt solution before centrifugation at 1000 x g for 8 mins. Following centrifugation the water layer was gently swirled using a Pasteur pipette to create a vortex which draws the oocysts from the layer between the dH₂O and salt into the dH₂O, the layer was removed and added to 6 ml dH₂O, the total volume was made up to 10 ml, mixed by inverting and centrifuged at 5000 x g for 5 mins. The supernatant was poured off and discarded and the pellet retained for DNA extraction (Ryley et al., 1976).

6.3.1.5 DNA Extraction

DNA was extracted as described in section 2.5.1.

6.3.1.6 PCR amplification of the 18S rRNA gene

DNA samples were amplified using the 18S PCR described in section 2.6.1 and visualised on a 1.5% agarose gel stained with GelRed (Biotium, UK). All reactions were carried out in triplicate and a sample was considered positive if a band corresponding with ~840 bp was seen. Several controls, both negative (no template) and positive (*Cryptosporidium* DNA) were run with each set of reactions. A positive result was only accepted when all negative controls were negative and positive controls were positive.

6.3.2 Sample Collection

6.3.2.1 Adults

The study farm used for this study was the same farm which was used for the study described in Chapter 5. Faecal samples were collected from 29 in-calf adult dairy cattle (heifers and cows) three times per week for up to ten weeks pre-calving, in total 205 samples were collected. The cows were kept under the normal working conditions of this farm and as such were housed in cubicle sheds before calving, in straw courts at calving and in slatted cubicle sheds after calving when they returned to the milking herd. Cows were observed in the cattle shed until they defecated and then the entire motion was collected in a plastic bag. Due to the method of collection it was not always possible to collect a sample from each cow at each sampling point. After collection the samples were transferred to the laboratory where the entire motion was mixed and a sub-sample aliquoted into a 125 ml tub and stored at 4°C.

6.3.2.2 Calves

Samples from calves (n=30) born to these cows were collected from the day of birth until 6-9 weeks of age three times per week (Monday, Wednesday and Friday). The samples were collected from the rectum of calves using a gloved finger and bag for collection, where possible the entire motion was collected. It was not always possible to collect a sample from each calf on each sampling day; therefore the number of samples collected per calf did vary. The calves were kept under the normal working conditions of this farm with regards to housing, feeding and veterinary treatment. The calves were kept with their mothers for one/two days then moved to a small group (5-8 calves) for one week then moved to a larger indoor straw bedded pen with approximately 20 calves in each pen for the remaining five weeks. After this time the calves were moved, and groups mixed, to a different, indoor straw bedded pen. Samples were collected from the same calves again at six months of age where possible. The total number of samples collected was 380. The samples were mixed thoroughly and aliquoted into bijoux tubes and stored at -20°C for further processing

6.3.3 Sample processing and DNA isolation

6.3.3.1 Adults

Samples were processed to concentrate *Cryptosporidium* oocysts from a 50 g starting sample by acid flocculation and salt flotation as described in section 2.4.1. DNA was isolated as described in section 2.5.1.

6.3.3.2 Calves

Faecal samples from calves were processed and DNA isolated as described in section 2.5.2.

6.3.4 Species identification by nssm-PCR (nested species specific multiplex-PCR)

For the differentiation of *Cryptosporidium* species an nssm-PCR was carried out using the first round PCR product from the 18S PCR (see section 2.6.1) as a template. The PCR products were visualised on a 1.5% agarose gel stained with GelRed (Biotium, UK). All reactions were carried out in triplicate and a sample was considered positive if a band corresponding with 840 bp and any combination of 625 bp, 415 bp, 305 bp or 241 bp was seen. Positive controls for each *Cryptosporidium* species tested for were included as well as extraction controls and no template controls. A positive result was only accepted when all negative controls were negative and all positive controls were positive.

6.3.5 C. parvum subtyping

Sequencing of *C. parvum* positive samples was carried out by amplification and sequencing of the *gp60* gene (see sections 2.6.3 and 2.7). The PCR products were visualised on a 1.5% agarose gel stained with GelRed (Biotium, UK). Reactions were originally carried out in duplicate and if amplification was not successful the reactions were repeated. A sample was considered positive if a band corresponding with ~375 bp was seen. Positive, negative and extraction controls were included for each set of reactions. A positive result was only accepted when all negative controls were negative and all positive controls were positive.

gp60 positive samples were sequenced in the forward and reverse orientation by GATC Biotech (GATC, Germany) as described for "supremerun96" sequencing in section 2.7.3. The sequences were analysed using Chromas Lite (version 2.01 Technelysium) and the subtypes named as according to (Sulaiman et al., 2005).

6.4 Results

6.4.1 Optimisation of oocyst concentration and DNA extraction from adult cattle faeces

6.4.1.1 Spiking Experiment

The results of the spiking experiment comparing the four methods of oocyst extraction from adult cattle faeces are given in Table 6-2 and Figure 6-2. Acid flocculation alone failed to concentrate oocysts; SF alone was sensitive to 100 oocysts/g; the combined AF/SF method to 5 oocysts/g and direct DNA extraction was sensitive to 100 oocysts/g of faeces. The PCR reactions were carried out in triplicate.

			Method	
No. of oocysts	Acid	Salt	Acid Flocculation and	No Prior
per gram	Flocculation	Flotation	Salt Flotation	Concentration
0	-	-	-	-
5	-	-	++	-
10	-	-	+++	-
100	-	++	+++	+
1000	-	+++	+++	+++

Table 6-2: Detection limits of each of the four concentration methods.

- = negative, + = positive in 1 of 3 reactions, ++ = positive in 2 of 3, +++ = positive in 3 of 3 PCR reactions



Figure 6-2: Spiking experiment to compare AF, SF, AF/SF and no concentration methods of oocyst detection from adult cattle faeces.

Top image: lane 1= DNA Molecular Weight Marker XIV 100 base pair ladder (Roche); lanes 2-13= direct method: 2-4=5 oocysts/g; 5-7=10 oocysts/g; $8-10=10^2$ oocysts/g; $11-13=10^3$ oocysts/g. Lanes 14-25=salt flotation only: 14-16=5 oocysts/g; 17-19=10 oocysts/g; 20-22= 10^2 oocysts/g; 23-25= 10^3 oocysts/g. Lanes 26-27= positive controls.

Bottom image: lane 1= marker (Roche 100bp); lanes 2-13= acid flocculation only: 2-4=5 oocysts/g; 5-7=10 oocysts/g; $8-10=10^2$ oocysts/g; $11-13=10^3$ oocysts/g. Lanes 14-25=acid flocculation and salt flotation: 14-16=5 oocysts/g; 17-19=10 oocysts/g; $20-22=10^2$ oocysts/g; $23-25=10^3$ oocysts/g. Lanes 26-27= negative controls.

6.4.1.2 Evaluation of AF and AF/SF

Two hundred and five faecal samples were tested for the presence of *Cryptosporidium* DNA by either AF followed by DNA extraction or AF/SF followed by DNA extraction. Of the 205, 11 tested positive using the acid flocculation method only and 57 tested positive using the AF/SF.

6.4.2 Prevalence of Cryptosporidium

6.4.2.1 Adult Cattle

In total 205 faecal samples from 29 adult dairy cattle (cows and heifers) were collected. On average seven samples from each animal were collected over the pre-calving period (range 3-13, median 7, mode 7). Due to the nature of

collection it was not possible to collect a sample from each animal every day, it was hoped that a minimum of one sample per cow per week would be obtained. The majority of samples were collected during the four week period immediately prior to calving (Figure 6-3).

	Pre-Calving																Post-	Calvir	ng																
I.D	١	Week	9	١	Neek	8	١	Neek	7	١	Week	6	١	Neek	5		Week	4	١	Week	3	١	Week	2	١	Week	1	Calving	١	Neek	1		Week	2	Samples Per Cow
254																		0	0	0	0	Х	0	Х	0	0	Х	Х							4
220													0	0	Х	0	0	0	х	0	Х	Х	Х	0	Х	0	0	0	0	Х	0	0	0	Х	8
136														0	0	0	0	Х	0	0	Х	Х	Х	Х	0	0	Х	0	Х						7
278													0	0	0	0	0	0	0	0	Х	Х	0	Х	0	0	0	0							3
350													0	0	0	0	0	0	0	Х	Х	0	Х	0	0	Х	0	0							4
26												0	0	0	0	0	0	Х	0	Х	Х	Х	Х	Х	0	0	0	0	х						7
354												0	0	0	0	Х	0	0	0	0	Х	Х	0	0	0	0	Х	0							4
52											0	0	0	0	0	0	0	Х	0	0	Х	0	0	0	Х	0	0	0	0	0	0	0	Х		4
15											0	0	Х	0	0	0	Х	0	Х	Х	Х	0	0	0	0	0	0	0	х						6
256											0	0	0	0	0	0	0	Х	Х	Х	0	0	0	0	Х	0	0	Х	Х						6
243										0	0	0	0	0	0	Х	Х	0	Х	Х	Х	Х	0	0	Х	0	0	0							7
96										0	0	0	0	0	0	0	Х	Х	0	Х	0	0	0	Х	0	0	Х	0							5
178										0	0	Х	0	Х	0	0	0	Х	0	Х	0	Х	Х	0	Х	0	0	Х							8
51										0	0	Х	0	Х	0	0	Х	0	Х	Х	0	0	Х	Х	0	0	0	0							7
64										0	0	0	0	0	0	0	Х	Х	Х	Х	0	Х	Х	0	0	0	0	Х							7
20								0	0	0	0	0	0	0	0	Х	Х	0	0	0	Х	0	Х	0	Х	0	0	Х	Х						7
130								0	0	0	0	0	0	0	0	Х	0	Х	0	Х	Х	Х	Х	0	Х	0	0	0	Х		Х				9
105								0	0	0	0	0	0	0	Х	Х	0	0	0	Х	0	Х	Х	0	Х	0	0	0							6
108							0	0	0	0	0	0	Х	Х	Х	0	Х	0	Х	0	Х	0	Х	Х	Х	0	0	0	Х	Х					11
62						0	0	0	0	0	0	0	0	0	0	0	0	0	Х	0	0	0	Х	0	Х	0	Х	0							4
176						0	0	0	0	0	0	Х	0	0	0	0	Х	0	0	0	0	0	Х	0	0	0	Х	0	Х						5
138					0	0	0	0	0	0	0	0	0	0	0	Х	Х	0	0	Х	0	Х	0	0	0	0	0	0	Х	Х					6
50					0	0	0	0	Х	0	0	Х	0	Х	0	0	Х	0	0	Х	Х	Х	Х	Х	0	Х	Х	0	Х	Х					13
54				0	0	0	0	0	0	0	0	Х	Х	0	Х	Х	Х	Х	0	0	0	Х	0	0	0	0	0	0	Х						8
357		0	0	0	0	0	0	Х	0	0	Х	0	0	0	Х	0	0	0	0	Х	Х	0	0	Х	Х	Х	Х	0	Х						10
149	0	0	0	0	Х	0	0	Х	0	Х	Х	Х	0	0	Х	0	0	Х	Х	0	0	0	Х	0	0	0	Х	Х							11
57	0	0	0	Х	0	0	Х	0	0	0	0	0	0	Х	0	0	0	0	0	0	Х	0	Х	Х	0	Х	0	0							7
82	0	Х	0	0	0	0	Х	Х	0	0	0	0	Х	0	0	0	Х	Х	Х	0	0	0	X	Х	Х	Х	0	0							11
233	Х	0	0	0	0	Х	0	0	0	0	0	Х	0	0	0	0	Х	Х	Х	0	Х	0	0	0	Х	0	Х	0	Х						10
Total	1	1	0	1	1	1	2	3	1	1	2	7	4	5	6	7	13	12	11	14	16	14	17	11	13	5	10	6	13	4	1	0	1	1	205

Figure 6-3: Samples collected from each cow pre and post calving. Faecal samples collected over the study period from 29 adult dairy cattle (cows and heifers). A blue "X" denotes a sample collected on that day while "0" shows that no sample was collected from that cow on that particular day. The total number of samples collected at each collection point are shown at the bottom of the figure and total number of samples collected from each cow are shown in the right-hand column.

All samples were tested for the presence of *Cryptosporidium* DNA by 18S nssm-PCR after oocyst concentration by acid flocculation and salt flotation. Overall 27.9% (n=57) of samples from adult cows tested positive for *Cryptosporidium*. Four cows did not test positive for *Cryptosporidium* at any point throughout the study, the remaining 25 cows tested positive on at least one occasion and two cows that tested negative for *Cryptosporidium* pre-calving tested positive postcalving. The percentage of *Cryptosporidium* positive samples remained fairly stable throughout the study period (Figure 6-4) at ~30%. There were more fluctuations when looking at each individual time point when the percentage of positive samples ranged from 0-100%.



Figure 6-4: Positive samples detected in adult cows pre- and post-calving.

Percentage of positive samples detected in cows in the weeks pre- and post-calving. The percentage of positive samples at each sampling point (i, ii, iii) is indicated by a red bar. Weeks 8 and 9 pre-calving and week 2 post-calving have been omitted from this graph as only two samples were collected in each of these week. "C" indicates the day of calving, it was not possible to collect a sample from every cow on the day it calved.

6.4.2.2 Calves

In total 380 samples were collected, 364 from the initial collection period then 16 from the same calves at 6 months of age. On average 13 samples were collected from each calf (range 5-20, median 13, mode 14). Due to the nature of collection it was not possible to collect a sample from each animal every day, (Figure 6-5).

															alf A	ge													
Calf ID	V	Veek	1	v	Veek	2	v	Veek	3	V	Veek	4	V	Veek	5	v	Veek	6	v	Veek	7	V	Veek	8	v	Veek	9	6 Months	Total
2596	0	0	Х	0	Х	0	Х	0	Х	0	Х	Х	Х	0	Х	0	0	Х	0	Х	0	Х	0	Х	0	Х	Х	0	14
2604	0	0	0	Х	Х	0	0	0	Х	Х	0	Х	Х	Х	Х	0	0	0	Х	Х	Х	0	Х	Х	0	0	0	0	13
2605	0	Х	0	Х	Х	Х	Х	Х	Х	Х	0	Х	Х	Х	Х	0	Х	0	Х	Х	Х	0	Х	Х	Х	Х	0	0	20
2606	0	0	Х	Х	Х	0	Х	Х	Х	0	Х	Х	0	0	0	Х	0	Х	Х	Х	0	Х	Х	Х	Х	Х	0	Х	18
2607	0	0	Х	Х	Х	Х	Х	Х	Х	0	Х	Х	Х	Х	0	Х	0	0	Х	Х	0	0	Х	Х	Х	0	0	0	17
2608	0	Х	Х	Х	Х	Х	Х	Х	0	Х	Х	0	х	0	0	0	Х	Х	0	0	Х	0	0	Х	Х		0	Х	16
2609	0	0	0	Х	Х	Х	Х	Х	0	Х	Х	Х	Х	0	0	0	Х	Х	0	0	Х	0	Х	Х	0		0	Х	15
2610	Х	0	Х	Х	0	Х	Х	0	Х	Х	Х	Х	0	0	0	0	0	Х	0	0	Х	Х	Х	0				Х	14
2611	0	0	0	Х	Х	0	Х	Х	0	Х	Х	0	Х	0	Х	0	Х	Х	0	0	Х	Х	Х	Х	Х			0	15
2612	0	0	Х	0	Х	Х	0	Х	0	Х	Х	0	Х	0	0	0	Х	Х	0	0	Х	0	0	Х	0			Х	12
2613	0	0	Х	Х	0	Х	Х	0	Х	Х	Х	Х	0	Х	0	Х	Х	Х	0	Х	0	0	0	0				0	13
2614	0	0	Х	Х	Х	Х	0	Х	Х	Х	0	0	Х	0	Х	Х	Х	0	Х	Х	Х	0	Х					Х	16
2615	0	0	Х	Х	Х	Х	0	Х	0	Х	0	0	Х	0	Х	Х	Х	0	Х	0	Х	Х	0					Х	14
2616	0	0	Х	0	Х	Х	Х	Х	Х	Х	Х	0	Х	0	Х	0	Х	0	Х	0	0	Х	0					Х	14
2617	0	0	Х	Х	Х	Х	Х	Х	Х	Х	Х	0	Х	0	Х	Х	Х	0	0	0	Х	Х	Х					Х	17
2618	0	0	0	Х	Х	Х	х	Х	Х	Х	Х	0	Х	0	Х	0	Х	0	Х	Х	Х	0	0					Х	15
2619	0	0	0	Х	Х	Х	Х	Х	Х	Х	Х	0	Х	0	0	Х	Х	0	0	0	Х	0	0					Х	13
2620	0	0	0	0	Х	Х	Х	Х	Х	Х	Х	0	Х	0	0	0	Х	0	Х	0	0	Х	0					Х	12
2621	0	0	Х	Х	Х	Х	х	Х	0	Х	0	Х	Х	Х	0	Х	0	0	0	0								0	11
2622	0	0	Х	0	Х	Х	Х	Х	0	0	0	Х	Х	0	0	Х	Х	Х	Х	Х								0	12
2623	0	Х	0	Х	Х	Х	Х	Х	0	Х	0	Х	Х	Х	0	Х	0	Х	Х	Х								0	14
2624	0	0	0	Х	Х	Х	Х	0	0	0	0	Х	Х	0	0	0	Х	Х	Х									0	9
2625	0	0	0	Х	Х	Х	Х	0	Х	0	Х	Х	0	0	0	Х	Х	Х	0									Х	11
2626	0	0	Х	Х	Х	Х	0	X	0	Х	0	Х	0	Х	0	Х	X	0										Х	11
2627	0	0	Х	0	0	Х	0	Х	Х	Х	0	Х	X	Х	Х	0												Х	10
2628	0	0	Х	0	0	0	Х	Х	Х	0	0	Х	X	Х	0													0	7
2629	0	0	0	X	0	X	X	X	0	X	X	X	X	Х														0	9
2630	0	0	0	0	X	X	X	0	0	0	X	0	X															0	5
2631	0	0	X	0	X	0	X	X	X	X	0	0	0	0														X	/
2633	0	0	Х	0	Х	0	Х	Х	Х	Х	0	0	0	0								-				-		0	6
Total	1	3	18	20	25	23	24	23	18	22	17	17	23	10	10	12	17	12	12	10	12	8	9	9	5	3	1	16	380

Figure 6-5: Number of samples collected from each calf. Faecal samples collected over the study period from 30 dairy calves from birth. A blue "X" denotes a sample collected on that day while "0" shows that no sample was collected from that calf at that particular collection point. The total number of samples collected at each collection point are shown at the bottom of the figure and total number of samples collected from each calf are shown in the right-hand column

All 380 samples were tested for the presence of *Cryptosporidium* parasite DNA using the 18S genus specific PCR and overall percentage of positive samples over the whole study was 51.0% (n=194). In the first nine weeks of life each calf tested positive for *Cryptosporidium* on at least 2 occasions, giving a period prevalence of 100% and the overall percentage of samples positive in calves <9 weeks was 51.9% (n=189). The earliest evidence of oocyst shedding occurred in one calf one day after birth, the majority of calves began shedding oocysts at day 4 (7 calves) or day 5 (6 calves) after birth. Peak shedding of *Cryptosporidium* oocysts occurred in week two with 95% of calves tested shedding oocysts (Figure 6-6). The percentage of positive samples per calf ranged from 22.2% to 80.0% although the calf with 80.0% positive samples did have fewer samples collected overall (Figure 6-7).



Figure 6-6: Shedding profile of *Cryptosporidium* oocysts in calves <9 weeks. Percentage of positive samples detected in calves in the first nine weeks of life. The percentage of positive samples at each sampling point (i, ii, iii) is indicated by a red bar. Peak shedding of oocysts occurred in week 2 of life and tailed off towards the end of week 8/9.

										-				Ca	alf Age				-		-					-			
Calf ID		Week 1	L		Week 2	2		Week	3		Week	4		Week 5	5	1	Week 6	5		Week	7		Week 8	3		Week	9	6 Months	% Positive
2596	0	0	Ν	0	Р	0	N	0	N	0	Р	Ν	Р	0	Ν	0	0	Р	0	Ν	0	Ν	0	Ν	0	Ν	Ν	0	28.6
2604	0	0	0	Р	Р	0	0	0	Р	Ν	0	Р	Р	Ν	Ν	0	0	0	N	N	Ν	0	Ν	Ν	0	0	Ν	0	35.7
2605	0	Ν	0	Р	Р	Р	Р	Ν	Ν	Ν	0	N	Р	Ν	Р	0	Р	0	N	Р	Ν	0	Ν	Ν	Ν	Ν	Ν	0	38.1
2606	0	0	Р	Р	Р	0	Р	Р	N	0	Р	Р	0	0	0	Р	0	N	Р	Ν	0	Ν	Ν	Ν	Ν	Ν		N	50.0
2607	0	0	Ν	Р	Р	Р	Ν	Р	Ν	0	Ν	Р	N	Ν	0	Ν	0	0	Р	Ν	0	0	Р	Ν	Ν	0		0	41.2
2608	0	Ν	Р	Р	Р	Р	Р	Ν	0	Р	Р	0	Р	0	0	0	Ν	N	0	0	Р	0	0	Ν	N			N	56.3
2609	0	0	0	Р	Р	Р	Р	Р	0	Р	Р	Р	Ν	0	0	0	Ν	Р	0	0	Ν	0	Р	Ν	0			Р	73.3
2610	N	0	Р	Р	0	Ν	Р	0	Р	Ν	Ν	Р	0	0	0	0	0	N	0	0	Р	Ν	Ν	0				N	42.9
2611	0	0	0	Р	Р	0	Р	Р	0	Ν	Р	0	Р	0	Ν	0	Ν	Р	0	0	Ν	Ν	Ν	Ν	N			0	46.7
2612	0	0	Р	0	Ν	Р	0	Р	0	Ν	Р	0	Р	0	0	0	Ν	Ν	0	0	Ν	0	0	Ν	0			Р	50.0
2613	0	0	Р	Р	0	Р	Р	0	Р	Р	Р	Ν	0	Р	0	Ν	Р	Ν	0	N	0	0	0	0				0	69.2
2614	0	0	Р	Р	Р	Р	0	Ν	Р	Ν	0	0	Ν	0	Ν	Р	Ν	0	Ν	Р	Ν	0	Ν					N	43.8
2615	0	0	Р	Р	Р	Р	0	Р	0	Ν	0	0	N	0	Ν	Ν	Ν	0	Ν	0	Ν	Ν	0					N	35.7
2616	0	0	Ν	0	Р	Р	Р	Ν	Р	Ν	Р	0	Р	0	Ν	0	Ν	0	Ν	0	0	Ν	0					N	42.9
2617	0	0	Р	Р	Р	Р	Р	Ν	Р	Ν	Р	0	Р	0	Ν	Ν	Ν	0	0	0	Ν	Ν	Ν					N	47.1
2618	0	0	0	Р	Р	Ν	Р	Р	Р	Р	Ν	0	N	0	Ν	0	Ν	0	Р	Ν	Ν	0	0					Р	53.3
2619	0	0	0	Р	Р	Р	Р	Р	Р	Р	Р	0	Р	0	0	Ν	Ν	0	0	0	Ν	0	0					N	69.2
2620	0	0	0	0	Р	Р	Р	Р	Р	Р	Р	0	Ν	0	0	0	Ν	0	Р	0	0	Ν	0					Р	75.0
2621	0	0	Р	Р	Р	Ν	Р	Р	0	Р	0	Ν	Ν	Ν	0	Ν	0	0	0	0								0	54.5
2622	0	0	Ν	0	Р	Р	Р	Р	0	0	0	Ν	N	0	0	Р	Р	Ν	Ν	Ν								0	50.0
2623	0	Р	0	Р	Р	Р	Р	Р	0	Ν	0	Ν	Р	Ν	0	Ν	0	Ν	Ν	Ν								0	50.0
2624	0	0	0	Р	Р	Р	Р	0	0	0	0	Р	N	0	0	0	Р	Ν	Р									0	77.8
2625	0	0	0	Р	Ν	Р	Р	0	Р	0	Ν	Р	0	0	0	Ν	Р	Р	0									N	63.6
2626	0	0	Ν	Р	Р	Р	0	Р	0	Ν	0	Ν	0	Ν	0	Ν	Р	0										Р	54.5
2627	0	0	Р	0	0	Р	0	Р	Р	Ν	0	Р	Ν	Ν	Ν	0												N	50.0
2628	0	0	Ν	0	0	0	Р	Р	Ν	0	0	Р	Ν	Ν	0													0	42.9
2629	0	0	0	N	0	Р	Р	Ν	0	Ν	Ν	Ν	Ν	Ν														0	22.2
2630	0	0	0	0	Р	Р	Р	0	0	0	Р	0	Ν															0	80.0
2631	0	0	Р	0	Р	0	Р	Ν	Ν	Р	0	0	0	0														Ν	57.1
2633	0	0	Р	0	Р	0	Р	N	N	Р	0	0	0	0														0	66.7
% Positive	0.0	33.3	66.7	95.0	92.0	87.0	91.7	65.2	61.1	40.9	70.6	52.9	43.5	10.0	10.0	25.0	35.3	33.3	41.7	20.0	16.7	0.0	22.2	0.0	0.0	0.0	0.0	31.3	

Figure 6-7: Positive and negative samples collected from each calf. Samples collected from each calf at each time point. A positive sample is denoted by a red "P" while a negative sample is denoted by a blue "N", "0" indicates that a sample was not collected from that calf on that day. The percentage of positive samples at each time point are shown at the bottom of each column while percentage of positive samples from each individual is shown in the right-hand column.

6.4.3 Speciation of *Cryptosporidium* positive samples by nssm-PCR

All *Cryptosporidium* positive samples were speciated using a nssm-PCR described in section 2.6.2.

6.4.3.1 Adults

Of the 57 *Cryptosporidium* positive samples from the adult cattle, 91.2% (n=52) of the samples were identified as *C. parvum*, 3.5% (n=2) as a mixed infection of *C. parvum* and *C. ryanae*, 1.75% (n=1) as a mixed infection of *C. ryanae* and *C. bovis*, 1.75% (n=1) as a mixed infection of *C. parvum* and *C. andersoni* and 1.75% (n=1) as a mixed infection of *C. parvum*, *C. ryanae* and *C. bovis* (Figure 6-8).



Figure 6-8: Species detected in individual cows.

Percentage of different species detected in positive samples from each individual cow. C. parvum was the most frequently detected species in these samples. Four of the cows produced no Cryptosporidium positive samples.

To confirm the results of the nssm-PCR a selection of samples (n=19) were sent for sequencing by GATC (see section 2.7), 12 of these failed sequencing (unreadable trace files) the remaining 7 matched the result of the nssm-PCR.

6.4.3.2 Calves

All *Cryptosporidium* positive samples were speciated using the nssm-PCR described in section 2.6.2. All positive samples from calves <9 weeks were *C. parvum* positive. Of the five positive samples from the calves at 6 months old, 20% (n=1) were *C. parvum* positive and 80% (n=4) were mixed infections of *C. parvum* and *C. bovis*.

6.4.4 Subtyping of C. parvum by gp60 PCR

To determine the *gp60* subtypes of the *C. parvum* positive samples collected from both adult cattle and calves a subset of samples were amplified using the *gp60* nested PCR described in section 2.6.3 and sequenced as described in section 2.7.

6.4.4.1 Adults

A small selection (n=32) of *C. parvum* positive samples were subjected to a nested PCR to amplify a fragment of the *gp60* gene. Fifteen of these failed to amplify in the PCR. The remaining seventeen were positive and were sequenced successfully. Four different *gp*60 subtypes were identified; these were IIaA15G2 (n=2), IIaA15G2R1 (n=11), IIaA18G2R1 (n=3) and IIaA19G2R1 (n=1).

6.4.4.2 Calves

A larger selection of *C. parvum* positive samples (n=106) from calves were further analysed using the *gp60* nested PCR, 12 of these failed to amplify. Of the 89 which were amplified successfully, 83 of these were sent for sequencing. Eighty of these returned good trace sequences and two subtypes were identified IIaA19G2R1 (n=75) in the samples from animals <9 weeks of age and IIaA15G2R1 (n=5) in samples from the same animals at 6 months of age (Table 6-3).

Table 6-3: *gp*60 subtypes detected in calves at different ages.

Calf ID	< 1 Weeks	2-4 Weeks	5-8 Weeks	6 Months
2604	IIaA19G2R1	IIaA19G2R1		
2609		IIaA19G2R1	IIaA19G2R1	IIAa15G2R1
2612	IIaA19G2R1			IIAa15G2R1
2613	IIaA19G2R1	IIaA19G2R1	IIaA19G2R1	
2617	IIaA19G2R1	IIaA19G2R1	IIaA19G2R1	
2619		IIaA19G2R1		IIaA15G2R1
2620		IIaA19G2R1		IIaA15G2R1
2626		IIaA19G2R1		IIaA15G2R1
2633	IIaA19G2R1	IIaA19G2R1		

6.5 Discussion

6.5.1 Adults

In adult cattle the reported prevalence of *Cryptosporidium* varies markedly between studies (0-71%) (Table 6-4), the most likely reason for this variation is the differing sensitivities of the techniques used to sample and test the faeces. The new and improved techniques developed in this study enabled detection of lower numbers of Cryptosporidium oocysts in faeces collected from adult cattle than other methods. Using these techniques it was possible to detect as few as 5 oocysts per gram of faeces which is more sensitive than other methods such as acid fast staining (1000 opg), fluorescent staining (66 opg), although not as sensitive as the gold-standard qPCR which can detect as little as 40 fg of DNA (one oocyst) (Hadfield et al., 2011). The increased sensitivity with this method is probably due to a combination of the larger starting sample (50 g) combined with the two methods of concentrating the oocysts. The acid flocculation step removes the majority of fibrous material from the faeces and when followed with salt flotation there is very little debris left in the final pellet. This reduces the amount of inhibitors present in the pellet for DNA extraction and PCR. In this study the samples from the adult cattle were tested by both acid flocculation followed by DNA extraction, and acid flocculation and salt flotation followed by DNA extraction. Using the acid flocculation 5.4% (n=11) samples tested positive compared with 26.8% (n=55) when using both acid flocculation and salt flotation. This indicates that the initial acid flocculation concentration step is not enough to increase the chances of detecting oocysts.
Table 6-4: Summary of studies carried out investigating Cryptosporidium in adult cattle.Table shows the studies carried out to determine prevalence and species of Cryptosporidium in adult cattle. The number of animals and samples, methods and results are summarised.

Author(s)	Year	Number of Animals	Number of Samples	Methods Used	%age Positive	Species
(Villacorta et al.)	1991	131	131	Microscopy (Heine's Technique)	1.4%	C. parvum (n=2)
(Castro-Hermida et al.)	2006	63	63	Microscopy (Heine's Technique)	8%	No Speciation
(Gow and Waldner)	2006	560	560	Sucrose flotation, Microscopy (DFA)	1%	No Speciation
(Singh et al.)	2006	67	67	Sucrose flotation, Microscopy (mZN)	13.4%	No Speciation
(Lorenzo Lorenzo et al.)	1993	131	131	Microscopy (Heine's Technique)	71.8%	<i>C. muri</i> s* (n=2) <i>C. parvum</i> (n=92)
(Scott et al.)	1995	553	553	Two smears of each sample, Microscopy (mZN and FITC to confirm) 50x 5g samples - sucrose flotation, Microscopy (mZN)	62.4%	<i>C. parvum</i> (n=345)
(Quilez et al.)	1996	225	225	Formalin-ethyl acetate sedimentation, Microscopy (mZN)	17.8%	<i>C. parvum</i> (n=40)
(Bukhari and Smith)	1996	109	109	Water-ether sedimentation, Microscopy (FITC, DAPI)	23.0%	<i>C. parvum</i> (n=21) <i>C. muris</i> * (n=4)
(Esteban and Anderson)	1995	Not Reported	1746	Microscopy (mZN)	8.0%	C. muris*
(Esteban & Anderson)	1996	1240 [†]	1240	Microscopy (mZN)	0.6-2.6%	C. muris*
(Atwill et al.)	1998	154	384	Direct fluorescent antibody (DFA) and acid-fast (AF) assays	0.0%	N/A
(Kaneta and Nakai)	1998	512	512	Sucrose flotation, Microscopy (modified Kinyoun acid fast staining), Mouse Bioassay	4.7%	<i>C. muris</i> * (n=24)
(Atwill et al.)	1999	484	484	Immunofluorescent Microscopy	0.6%	C. parvum
(Ares-Mazas)	1999	41	41	Immunofluorescent Microscopy	0.0%	N/A
(Faubert and Litvinsky)	2000	148	Not Reported	500g samples, Sucrose flotation, Microscopy (ZN)	Not Reported	Not Reported
(Fayer et al.)	2000	184	184	10g faecal sample, Filtered, Caesium Chloride Flotation, Fluorescence Microscopy	22.3%	<i>C. parvum</i> (n=38) <i>C. andersoni</i> (n=3)
(Wade et al.)	2000	998	998	1g Faecal Samples, Sucrose Flotation, Microscopy (bright field and phase contrast)	1.5%	<i>C. muris</i> * (n=15)
(Enemark et al.)	2002	117	117	Smears, Microscopy (mZN, FITC), one sample was speciated by PCR-RFLP, Mouse Bioassay	40.2%	C. parvum (n=12) C. andersoni (n=27)

						C. parvum & C. andersoni (n=8)
(Atwill and Pereira)	2003	43	86	2g faecal sample, Microscopy (IMS–DFA)	0.0%	N/A
(Kvac and Vitovec)	2003	96	96	Sucrose Flotation, Microscopy (light), PCR (18S), Sequencing	43.8%	C. andersoni (n=42)
(Sturdee et al.)	2003	736	736	Modified Formal-Ether Sedimentation, Microscopy (FITC, mZN)	3.5%	<i>C. parvum</i> (n=26)
(McAllister et al.)	2005	669	669	2-3g faeces, Filtered through sponge, Sucrose Flotation, Microscopy (FITC)	29.0%	C. parvum C. muris*
(Fayer et al.)	2006	571	571	15g Faecal Sample, Filtered, Caesium Chloride Flotation, PCR (18S), Sequencing	11.9%	C. suis (n=1) C. parvum (n=4) Deer-like genotype [‡] (n=10) C. bovis (n=24) C. andersoni (n=29)
(Fayer et al., 2007)	2007	514	514	15g Faecal Sample, Filtered, Caesium Chloride Flotation, PCR (18S), Sequencing	5.7%	C. parvum (n=2) C. bovis (n=9) C. andersoni (n=20)
(Santin et al., 2008)	2008	30	360	15g Faecal Sample, Filtered, Caesium Chloride Flotation, Microscopy (IFA), PCR (18S), Sequencing	2.2%	Deer-like genotype [‡] (n=3) <i>C. andersoni</i> (n=2)
(Ralston et al., 2010)	2010	567	567	1-5g Faecal Sample, Filtered, Sucrose Flotation, Microscopy (FITC), PCR (18S), Sequencing	3.5%	C. andersoni (n=20)
(Wang et al.)	2011	295	295	20g Faecal Sample, Sucrose Flotation, Microscopy (bright field), PCR (18S), Sequencing	1.0%	C. andersoni (n=3)
(De Waele et al.)	2012	42	126	2g Faecal Sample, Filtered, Centrifuged, qPCR,	33.3%	C. parvum (n=6) C. bovis (n=2) C. andersoni (n=2)
(Silverlas and Blanco-Penedo)	2013	259	259	1g Faecal Sample, Sodium Chloride flotation, Microscopy (Fluorescence), PCR (18S & GP60)	3.5%	C. parvum (n=9)§
(Smith et al.)	2014	2669	2669	10g Faecal Sample, Microscopy (IFAT), PCR (RFLP, GP60), Sequencing	1.9%	C. andersoni (n=36)

* Now recognised as C. andersoni

† From 5 different herds - no indication is given of the number of cows from each herd

‡ Now recognised as *C. ryanae*

[§] Reported in paper as *C. parvum*-like but the single sample which was sequenced was identified as *C. bovis*

It has been reported that the most common species of *Cryptosporidium* to be detected in adult cattle is C. andersoni (Fayer et al., 2010a; Feng et al., 2007; Santin et al., 2008; Xiao, 2010). This species is only rarely detected in preweaned animals (Amer et al., 2010a; Kvac et al., 2011; Wang et al., 2011b) indicating that adult cattle are not a risk for infection for neonatal calves or that C. andersoni is not infective to neonatal calves. Some studies have reported a higher prevalence of *C. andersoni* in young calves, between 11.1-92.9% in calves <35 weeks old on a farm in Bohemia, Czech Republic (Kvac and Vitovec, 2003) and in China C. andersoni was found to be the most prevalent species in calves aged 3-11 months (8.70% of samples were C. andersoni positive and 2.6% C. bovis) (Wang et al., 2011a). Some studies investigating the risk factors associated with cryptosporidiosis in young calves found that calves kept for <72 h with their mothers increased the risk that the calf would suffer from cryptosporidiosis (Quigley, III et al., 1994). Other studies reported the opposite and show that being kept with their dams decreased the likelihood that calves would suffer from cryptosporidiosis (Duranti et al., 2009; Kvac et al., 2006). Most of these risk factor studies did not speciate or subtype the parasites detected in the dams, if, in fact, they even tested the adults for the presence Cryptosporidium. Although the present study was not looking at risk factors it did determine the species and subtypes of *Cryptosporidium* being shed by adult cattle and their offspring. We found that the predominant species of *Cryptosporidium* shed by adult cattle on this farm was *C. parvum*. This is unusual though C. parvum has been previously reported in adult cattle (Table 6-4). In this study 98.2% (n=56) of the positive samples (n=57) were identified as C. parvum, either as a single species or in conjunction with other species. This is a more unusual result when compared with other studies investigating species in adults as most studies show either a low percentage of samples (<20%) from adult cattle are C. parvum (Atwill et al., 1999a; De Waele et al., 2012; Enemark et al., 2002; Fayer et al., 2007; Fayer et al., 2006; Quilez et al., 1996; Silverlas and Blanco-Penedo, 2013; Sturdee et al., 2003; Villacorta et al., 1991) or that only a small number of adult cattle (<20%) are positive for any species of Cryptosporidium (Atwill et al., 1999a; Atwill and Pereira, 2003; Bukhari and Smith, 1996; Esteban and Anderson, 1995; Kaneta and Nakai, 1998; Quilez et al., 1996; Ralston et al., 2010; Santin et al., 2008; Silverlas and Blanco-Penedo, 2013; Smith et al., 2014; Sturdee et al., 2003; Wang et al., 2011a). Very few

studies show a higher percentage of positive samples from adult cattle (Enemark et al., 2002; Kvac and Vitovec, 2003; Lorenzo Lorenzo et al., 1993; Scott et al., 1995). The results from the present study sit somewhere in the middle at ~30% positive samples. This study was a longitudinal study and it should be noted that if a point prevalence study had been carried out on the same cattle then the percentage of positive samples would have ranged from 19.0% to 66.7% depending on the sampling day. This highlights the underestimation in parasite burden when looking at a single time point.

In this study 86.2% (n=25) of the cows were positive for *Cryptosporidium* at some point throughout the sampling period, again if this had been a point prevalence study a huge underestimation in the percentage of positive animals (0-50%) may have been made. In adult cattle, unlike calves, the prevalence of *Cryptosporidium* seems to vary hugely making it difficult or impossible to apply findings from one study to another. Even within one country or region the results from each study differ. The variation in results may be due to different sensitivities of the techniques used and highlights the need for standardised methodologies for detection of *Cryptosporidium* in adult cattle faeces.

From the results of this study when just taking into account the species of *Cryptosporidium* detected in both the adults and the calves the conclusion may be reached that adult cattle were in fact the source of infection for calves. The majority of positive samples (98.2% (n=56)) from adults were either C. parvum or C. parvum mixed infections as were all positive samples from calves <9 weeks. In previous studies this is the conclusion that many have made, several of these studies were carried out before we had the knowledge and ability to determine subtypes of these parasites or even before some common species of cattle adapted Cryptosporidium had been characterised (Faubert and Litvinsky, 2000; Scott et al., 1995). In this study a single subtype (IIaA19G2R1) was detected in all tested calf samples (n=75) from calves <9 weeks old and four different subtypes (IIaA15G2, IIaA15G2R1, IIaA18G2R1, IIaA19G2R1) were identified in the adults. The most common subtype in the adult cattle was IIaA15G2R1 which was identified in 11 samples from nine cows, this is also the subtype which has been identified in six month old calves in this, and a previous study on the same farm (Chapter 5). This subtype is the most common gp60 subtype identified worldwide

in both animal and human hosts (Broglia et al., 2008; Imre and Darabus, 2011). The other subtypes were identified in fewer samples, IIaA15G2 in two samples from two cows, IIaA18G2R1 in three samples from three cows and finally IIaA19G2R1 in a single sample from one cow which was also the most common subtype in calves <9 weeks of age.

One explanation for the single subtype in the young calves is that the one cow which shed IIaA19G2R1 calved early on in the calving season (Figure 6-9). This meant that this subtype was one of the first in the calving shed and could have been picked up by naïve calves. Once a calf is infected with Cryptosporidium it amplifies rapidly in the gut (Borowski et al., 2010) and the infected calf can then shed many infective oocysts into the environment where other calves can pick them up and become infected themselves (Nydam et al., 2001). The amplification rate and shedding of huge numbers of oocysts increases the likelihood that other calves will become infected with this subtype rather than another. It is also possible that some calves were infected with more than one subtype but that the calf gut preferentially amplifies one subtype over another or that the techniques used in this study were not sensitive enough to detect another subtype which may be present at a lower level. This is a recognised issue with Sanger sequencing and some researchers are now moving towards a Next Generation Sequencing (NGS) or Multi-Locus Genotyping (MLG) approach. One study which used NGS to study intra-host genetic diversity of *C. parvum* identified 10 different gp60 alleles in samples which Sanger sequencing had identified only one (Grinberg et al., 2013). In the present study samples from five cows were tested twice and three of these cows were shedding different gp60 subtypes at each sample point, the remaining two cows shed the same subtype each time. This demonstrates that testing animals once does not give a true and clear picture and also that cattle (and other hosts) are likely to be infected with, or shedding, multiple Cryptosporidium species and subtypes at any one time.



Figure 6-9: Diagrammatic representation of cows calving dates and *C. parvum* subtypes they shed.

Each individual cow is represented by a cow figure and each *gp60* subtype detected is represented by a colour. This diagram represents the subtypes identified in the cows throughout the study, the sample collected at calving may not have been tested.

6.5.2 Calves

6.5.2.1 Comparison with previous calf study

This study was carried out on the same study farm as the study described in Chapter 5. Overall the study design was similar in that a group of calves were sampled and tested for *Cryptosporidium* on a regular basis from birth. The first study (Chapter 5) was carried out between December 2011 and January 2012 and this study approximately a year later from November 2012 until February 2013. The percentage of positive samples in each age group of calves was lower in the second study (Figure 6-10). Generally shedding followed a similar pattern, but with a slightly earlier peak in shedding at week 2 in the second study before tailing off towards the end of week 5. In this study where animals were followed for longer, until 9 weeks of life compared with 6 weeks of life in the first study, there was a secondary peak in positive samples at week 7 (not shown). In both studies the same animals were sampled at 6 months of age, a similar pattern occurs here with the percentage of positive samples in the second study being lower than the percentage in the first study. The overall percentage of positive samples was lower in study two (mean 37.6%) compared to study one (mean 62.9%).



Figure 6-10: Comparison between studies.

Throughout the whole study the prevalence of *Cryptosporidium* was lower in study two compared with study one. The blue bars represent the percentage of positive samples each collection day from calves in study 1 and the orange bars represent the percentage of positive samples collected from calves in study two on each collection day.

Only *C. parvum* was detected in calves in this study whereas *C. bovis*, *C. ryanae* and mixed infections of *C. parvum* and *C. bovis* and *C. ryanae* and *C. bovis* were seen in study one from week three onwards. It is likely that other species or mixed infections were present in study two but that they were missed by both the nssm-PCR and sequencing analysis. As stated previously one failing of PCR amplification when mixed species are present is that the predominant species is often preferentially amplified. Although the nssm-PCR is very effective for detecting mixed infections it is possible that any mixed infections were below the level of detection for this method.

In both studies similar results were found when animals were aged 6 months. In the first study *C. parvum*, *C. bovis* and one mixed infection of *C. bovis* and *C. ryanae* were detected and in the second study *C. parvum* and *C. parvum* in older calves but in both studies on this farm *C. parvum* has been the predominant species detected in calves of this age group. Seventy percent (n=7) of the positive

samples for 6 month old calves were *C*. *parvum* in study one and all positive samples were *C*. *parvum* positive in study two (80% (n=4) mixed with *C*. *bovis*).

In both studies a number of samples were subtyped at the gp60 locus (n=30, study one and n=106, study two) and findings from both studies were similar. All samples tested from calves <9 weeks of age were IIaA19G2R1 and all samples tested from calves 6 months of age were IIaA15G2R1. These results confirm that, on this farm, there are at least two distinct *C. parvum* populations which infect cattle at different times throughout their lives. It is not known if these different subtypes are present in the different age groups as a result of: a local geographical distribution of subtypes on the farm; if something changes in the animal gut to enable the secondary subtype to establish; if the secondary subtype is always present in the animal but at low levels (which the detection methods used here cannot detect) or whether the secondary subtype is, in fact, another species all together. Many of the *Cryptosporidium* species are very similar and only vary genetically by one or two base-pairs at a single locus. Some examples of this would be the two very closely related (genetically 98.50 -99.70% at certain loci) species which infect humans; C. parvum, C. hominis but which differ in their epidemiology (Morgan and Thompson, 1999; Morgan-Ryan et al., 2002; Sulaiman et al., 2000). In livestock the species C. bovis and C. xiaoi are genetically very similar but one is predominantly found in cattle (C. bovis) while the other infects sheep (C. xiaoi) (Fayer and Santin, 2009; Fayer et al., 2005).

6.6 Conclusion

This study shows that contrary to popular belief adult cattle are a potential source of *Cryptosporidium* infection in calves. Calves are in close contact with faeces from their mother directly at birth. It is very likely that the infection in calves on this farm came from this source as calves show signs of infection and shed oocysts very soon after birth (as few as one day following birth). Further subtyping which was carried out on faecal samples from both calves and dams showed evidence that multiple subtypes of *C. parvum* may be present on a single farm and that these different subtypes may have different clinical manifestations, this is important from both an animal and public health perspective. Although only one adult cow was shown to shed the *C. parvum* subtype which commonly infected calves it is possible that the neonatal calf gut preferentially amplified one subtype over another and therefore the oocysts shed are predominantly the dominant subtype.

Papers in preparation and presentations at conferences

- 1. Wells, B., Thomson, S., Katzer, F., Development of a sensitive method to extract *Cryptosporidium* oocysts from adult cattle faecal samples. (Submitted)
- 2. Thomson, S., Jonsson, N., Innes, E.A., Katzer, F., Role of adult cattle in the transmission of *Cryptosporidium*. (In. Prep)
- 3. The role of adult cattle in the transmission and persistence of *Cryptosporidium* 5th International *Cryptosporidium* and *Giardia* Conference Sweden, 2014
- 4. Cryptosporidiosis in cattle Exchange Visit Brazil 2014

Chapter 7: General Discussion

The protozoan parasite Cryptosporidium is a common cause of diarrhoea in both human and animal hosts. In livestock hosts the disease caused by Cryptosporidium has been associated with reduced weight gain in infected animals (Ralston et al., 2010; Sweeny et al., 2011b) as well as mortality (Blanchard, 2012; Featherstone et al., 2010; Pritchard et al., 2008). There are many different species and genotypes of the parasite (~27 species and >60 genotypes currently recognised) which infect different host species (Chalmers and Katzer, 2013; Fayer, 2010). Some of the species and genotypes are believed to be host specific while others can infect several host species (Fayer, 2010; Ryan et al., 2014). In farm livestock (sheep and cattle) one of the most common causes of neonatal enteritis is C. parvum (de Graaf et al., 1999; Mosier and Oberst, 2000), this species of Cryptosporidium is also zoonotic and is responsible for ~50% of human cases of cryptosporidiosis in the UK (Chalmers et al., 2009a). Unlike several other pathogens associated with neonatal enteritis in sheep and cattle there is no available vaccine to prevent disease caused by C. parvum and treatment options are extremely limited and generally rely on rehydration therapy.

As previously discussed (Chapter 1), cattle may be infected with four different species of *Cryptosporidium* (*C. andersoni*, *C. ryanae*, *C. bovis* and *C. parvum*), infection with these species tends to follow an age-related distribution with young calves being infected with *C. parvum*, older calves with *C. bovis* and *C. ryanae* and adult cattle with *C. andersoni* (Fayer et al., 2007; Fayer et al., 2006; Santin et al., 2008; Santin et al., 2004). To date *C. bovis* and *C. ryanae* have not been associated with clinical disease (Fayer et al., 2008; Fayer et al., 2005) and *C. andersoni* has been associated with mild clinical disease (reduction in milk yield and weight-loss) (Kvac and Vitovec, 2003; Ralston et al., 2010). Morphologically these species look very similar, *C. andersoni* is slightly larger and more ovoid than the other three species (Fayer, 2010) which can make diagnosis and species identification difficult when relying on microscopy alone. More advanced methods for species identification generally involve sequence analysis of gene fragments or PCR-RFLP (as discussed in Chapter 1) these

methods can be costly and time consuming and it is also difficult or impossible to identify mixed infections.

The development of immunity to *Cryptosporidium* is poorly understood, it is known that humans and animal hosts can be infected with *Cryptosporidium* more than once demonstrating that the immune response to infection is not particularly strong or long lasting (Okhuysen et al., 1998). Many of the studies into development of immunity and immune responses to *Cryptosporidium* infection have been carried out in murine models. This has given a basic understanding into the possible factors which might play a role in the immune response. However the immune responses in an inbred model may not always be applicable to outbred populations such as humans and livestock hosts. We know that in certain livestock species it appears that as the animals increase in age they become less susceptible to disease (Harp et al., 1990; Kvac et al., 2014; Ortega-Mora and Wright, 1994), it is also known that prior infection can reduce the signs of clinical disease but asymptomatic infection (oocyst shedding) can still occur (Current and Snyder, 1988; Harp et al., 1990; Miller et al., 1990; Ortega-Mora and Wright, 1994). This demonstrates that previous exposure does not provide compete immunity to *Cryptosporidium* but that some level of protection does occur. It is not known if this partial protection can provide protection to infection with another *Cryptosporidium* species or isolate or if the protection is species specific.

The main aims of this PhD were to: assess the development of host resistance to *C. parvum* in experimentally infected lambs; develop a new typing tool to enable easy differentiation of the *Cryptosporidium* species commonly found in cattle; to determine which species and genotypes of *Cryptosporidium* calves became infected with in the first year of life and finally, to assess the role that adult cattle play in the transmission of *Cryptosporidium* (specifically *C. parvum*).

These aims were all achieved with a variety of studies including experimental infections, longitudinal studies and molecular tool design.

A novel technique (Chapter 4) for detection and differentiation of *Cryptosporidium* species commonly found in cattle (*C. andersoni*, *C. ryanae*, *C.*

parvum and *C. bovis*) was developed as part of this PhD. This enabled in-depth detailed longitudinal on farm studies to be carried out to investigate the transmission and persistence of *Cryptosporidium* in young calves and adult cattle. The nssm-PCR technique was used extensively throughout this PhD to identify the species of *Cryptosporidium* being shed in the faeces of naturally infected cattle.

The longitudinal study described in Chapter 5 (*Cryptosporidium* species and genotypes found in calves from birth to one year) and the longitudinal study described in Chapter 6 (Transmission of *Cryptosporidium* from dam to calf) confirmed that young calves (<9 weeks) are most commonly infected with *C. parvum* which supports the results of many other studies carried out worldwide (Abu et al., 2013; Amer et al., 2010b; Broglia et al., 2008; Diaz et al., 2010; Follet et al., 2011; Geurden et al., 2007; Huetink et al., 2001; Kvac et al., 2006; Lefay et al., 2000; Rieux et al., 2013; Santin et al., 2004; Silverlas and Blanco-Penedo, 2013; Thompson et al., 2007). The results of Chapter 5 also showed that neonatal calves become infected with *Cryptosporidium* at a very young age, almost immediately at birth, with some calves shedding oocysts at only 2 days of age. This suggested that adult cattle may be the source of infection for newborn calves and lead to the study described in Chapter 6.

The results of both longitudinal studies (Chapters 5 & 6) showed that all calves <9 weeks old on the study farm became infected with *C. parvum* within 2 weeks of life. Infection with *C. parvum* caused severe diarrhoea, inappetence and lethargy in several of the affected animals, others suffered from mild disease only and some animals showed only asymptomatic shedding of oocysts. Shedding of *C. parvum* oocysts occurred for at least 7 weeks PI with peaks in oocysts shedding at 2-3 weeks of age and again at 7 weeks of age (Chapter 5). Clinical signs tended to occur in animals <4 weeks of age and by 6/7 weeks of age many of the infected animals showed no signs of clinical infection though they were still shedding oocysts. In both studies any calves which were still present on the farm at 6 months of age were sampled and their faeces tested again. The results showed that the majority of calves in Chapter 5 (62%) and 32% of calves in Chapter 6 were positive for *Cryptosporidium* and that the most common species detected was *C. parvum*. As discussed previously, this is unusual as it is

generally expected that after initial exposure to the parasite, at a young age, cattle will develop immunity to infection. None of the 6 month old calves were showing any clinical disease at the time of sampling. Subtyping results revealed that the subtype of *C. parvum* identified in the young calves in both studies was IIaA19G2R1 which is an uncommon subtype identified in cattle only a small number of times previously (Couto et al., 2013; Ng et al., 2012; Wielinga et al., 2008; Xiao et al., 2007) while the subtype identified in faeces from 6 month old calves was the "common" subtype IIaA15G2R1 (Broglia et al., 2008; Imre and Darabus, 2011). Perhaps it is possible for cattle to develop immunity to one subtype of *C. parvum* following infection but this immunity may not necessarily protect against further infections with a different subtype. This possibility was tested in the experimental trial described in Chapter 3.

Following the results of Chapter 5 where it was shown that calves can become infected with Cryptosporidium almost immediately at birth the second longitudinal study (Chapter 6) focused on the possible transmission of C. parvum from dam to calf. Previous studies investigating *Cryptosporidium* infection in adult cattle reported that adult cattle were not believed to be a potential source from which young calves became infected (Atwill and Pereira, 2003; De Waele et al., 2012; Sturdee et al., 2003) although there are two studies which disputed these results (Faubert and Litvinsky, 2000; Kvac et al., 2011). These studies found that adult cattle either shed a species other than C. parvum or that they shed such low numbers of oocysts that the authors believed that they were not a significant risk. It is likely that the apparent low levels of Cryptosporidium found in adult cattle in previous studies (Table 6-4) could be due to the limitations of the detection methods applied or the number of samples taken. It is known that oocyst shedding and production is not a continuous process and that by taking only a single sample from any individual the overall percentage positivity is underestimated. In previous studies microscopy (with a variety of different staining techniques) was a commonly used method to detect *Cryptosporidium* oocysts. Microscopy is generally a much less sensitive method for the detection of oocysts compared with PCR or gPCR (Chalmers et al., 2011). Only a few of the studies (n=9) used either PCR or qPCR to detect Cryptosporidium in adult cattle faeces and many of the studies did not use any prior concentration step before examining the faeces. Faeces from adult

174

cattle are very different to faeces from young milk fed animals. Faeces from adult animals contain a large amount of fibrous materials and the volume produced is also much greater (~43.1 kg faeces per cow per day) (Weiss, 2015) and so the oocysts may be harder to detect without a concentration step. In the study described in this PhD (Chapter 6) an optimised technique was developed for the concentration and extraction of oocysts from adult cattle faeces which involved an acid flocculation and salt flotation steps (Wells et al., 2015 (to be submitted)).

In this study (Chapter 6) a group of in-calf cattle were sampled $3 \times per$ week from 10 weeks pre-calving until calving (increasing the number of samples collected from each cow) and then the calves born to these cows were sampled $3 \times per$ week for 6-10 weeks. We found that over 80% of the adult cattle involved in this study were shedding *Cryptosporidium* oocysts in their faeces, the majority of positive samples (98.2% (n=56)) were identified as *C. parvum* using the nssm-PCR described in Chapter 4. This finding is unusual as most studies investigating *Cryptosporidium* in adult cattle find only low numbers of positive samples, or animals and the species most commonly detected is *C. andersoni* not *C. parvum*. This is likely to be due to the additional concentration step employed during this study.

Subtyping results of *C. parvum* positive samples from adult cattle showed evidence of four different *C. parvum gp60* subtypes within this farm. One of the subtypes was IIaA19G2R1 which was the only subtype identified in young (<9 week old) calves and another (IIaA15G2R1) which was the only subtype found in older (6 months) calves, these and a further two subtypes (IIaA15G2 and IIaA18G2R1) were also detected in adult cattle on this farm. This age-related distribution of *C. parvum* subtypes in calves on this farm is very interesting; similar results were found in two separate studies in subsequent years on this farm indicating that this age-related distribution of *C. parvum* subtypes occurs frequently on this farm. From other work carried out on this particular farm it is known that the IIaA19G2R1 *C. parvum* subtype has been present and stable since at least 2008 (Gait et al., 2008). None of the other subtypes have been previously detected although other studies have only examined young calves.

As discussed in Chapters 5 & 6 there are several reasons for this apparent distribution of isolates including the possibility that the *C. parvum* isolate detected in 6 month old animals is another species of *Cryptosporidium*. To confirm this it would be necessary to carry out further typing of different loci such as actin, HSP70 and COWP or microsatellite analysis (Hotchkiss et al, 2015 (to be submitted)). Many species of *Cryptosporidium* are very similar to each other at one locus but different at others. Attempts were made to amplify DNA from the parasites from both calves and adults at different loci but these attempts were unsuccessful.

Another reason for the difference in isolates found in calves of different age groups on this farm could be related to changes in the host themselves. It has been reported in previous experimental studies that some host species (pigs and chickens) become more susceptible to certain *Cryptosporidium* species as they increase in age. The authors of these papers (Kvac et al., 2014; Lindsay et al., 1988; Yui et al., 2014) have suggested that this may be due to physiological changes in the gut or the gut flora of the host. It could be that the different isolate found in older cattle is somehow better adapted to replicate in the gut conditions of a fully mature bovine than in the undeveloped gut of a neonatal animal. Other studies examining the subtypes of *C. parvum* in different age groups of cattle on the same farms also found that adult cattle may be carrying different subtypes compared to their calves (Wells et al., 2015) indicating that adult cattle may not be an important source of infection for young calves.

Future work would expand on the findings from these studies and investigate further the possible transmission of *Cryptosporidium* from dam to calf using multiple farms (beef and dairy) with a known history of cryptosporidiosis. Groups of in-calf cows would be sampled regularly, daily if possible, to maximise the number of samples collected. In the study carried out here sample collection occurred 3 × per week but it is not possible to guarantee a faecal sample from each cow at each sampling time. More regular sampling would increase the chances of collecting sufficient numbers of samples from each cow to allow a more complete picture of the shedding dynamics in adult cattle to be formed. In addition to the techniques used in the study described in this thesis it would be beneficial to use a qPCR to quantify the shedding by adult cattle to determine if

shedding of oocysts increases in the approach to calving as suggested by studies on other livestock species (Atwill et al., 1998; Ortega-Mora et al., 1999; Xiao et al., 1994; Ye et al., 2013). This more in-depth approach would give us a more detailed insight into the persistence of the parasite on farms and within animals of differing ages.

Relatively little is understood about the possible differences between subtypes of the same *Cryptosporidium* species. It is known that different subtypes of the same species can vary in their clinical manifestation in infected individuals with some causing more severe disease than others. Some subtypes of C. hominis (the human adapted species of *Cryptosporidium*) cause vomiting and diarrhoea in infected people whereas other subtypes cause only diarrhoea (Cama et al., 2008; Cama et al., 2007). It has also been reported that some C. parvum isolates appear to be more common in humans (Alves et al., 2003; Cama et al., 2008; Hijjawi et al., 2010) whereas others more commonly infect cattle, sheep or goats (Quilez et al., 2008a; Quilez et al., 2008b). It is not known if these isolates can infect other host species or whether they are host-adapted as experimental studies to test this have not been carried out. It does however, demonstrate that even within a species of Cryptosporidium, different subtypes can behave differently in different host species. In this thesis a large scale experimental trial was carried out to assess the development of host resistance to distinct C. parvum isolates as both homologous and heterologous challenges were performed. Both isolates were known to cause clinical disease in naturally infected calves from two dairy farms in Scotland. The results of this study showed that distinct isolates of C. parvum can have different clinical manifestations in infected individuals even when those individuals have received an identical infective dose. The study also demonstrated that host factors play a role in the severity of disease and oocyst shedding as even within groups when animals received the same infective dose of the same isolate there was variation in the outcome of disease with some individuals experiencing more severe or prolonged diarrhoea or increased shedding of oocysts compared to others in the same group. This variation in response has been noted in several other experimental trials involving *Cryptosporidium* and other pathogens (Chappell et al., 1996; Chappell et al., 1999; DuPont et al., 1995; Harp et al., 1990; Moore et al., 2003; Okhuysen et al., 1999; Ortega-Mora and Wright, 1994; Petry et al.,

1995). Although it is clear that there is a difference in the virulence/pathogenicity of isolates there are also host factors which should be investigated further.

In the experimental infection study the results showed that while lambs do not develop complete resistance to disease by 6 weeks of age they do become less susceptible to clinical infection. If lambs are naïve to Cryptosporidium and are infected for the first time at 6 weeks of age then they show significantly fewer clinical signs than when lambs are infected at <1 week of age. Another study carried out in 1994, which was similar, demonstrated that lambs become resistant to disease at 8 weeks of age. The lambs in that trial showed no signs of clinical disease though some did shed oocysts when infected at 8 weeks of age. In the trial described in Chapter 3 it was shown that while older lambs exhibited fewer clinical signs of disease than younger lambs they still shed huge numbers of potentially infective oocysts in their faeces. This may be due to the dose of oocysts the animals were given; animals infected at <1 week of age were inoculated with 1×10^{6} oocysts while animals infected at 6 weeks of age were inoculated with 5×10^6 oocysts. This increase in infective dose may be the reason why the animals showed asymptomatic shedding of oocysts. Ortega-Mora & Wright (1994) orally infected the lambs in their trial with the same infective dose (1×10^6) at each age group. This may be why, in their study, the older lambs did not show any oocyst shedding. It is possible that if an 8 week old lamb had been challenged with a higher dose as in this trial, that asymptomatic shedding of oocysts would have occurred.

It has been previously demonstrated (Zambriski et al., 2013a; Zambriski et al., 2013b) that an increase in infective dose can lead to an increase in numbers of oocysts shed as well as duration of shedding. In this trial when animals were inoculated with a higher dose, they began shedding oocysts in their faeces much quicker than with the lower dose regardless of any previous exposure. They also shed greater numbers of oocysts following challenge with the higher dose confirming that the infective dose affects the prepatent period and influences oocyst shedding. This means that in a field setting it would be possible for an older animal to become infected with a large dose of *C. parvum* from a young animal and for the older animal to shed huge numbers of oocysts into the

environment while showing no clinical signs of infection itself. This could lead to the infection of a large number of susceptible young animals.

Work which could follow on from this study would involve more detailed investigation into the host factors involved in the development of resistance to disease rather than the differences in the parasite itself. While it is clear that there are differences within subtypes of *Cryptosporidium* parasites of the same species the development of resistance to infection is multifactorial and host factors are very important. Key points to address in future work would be the difference in the innate and adaptive immune responses in young <1 week old animals compared with naïve older animals when infected with Cryptosporidium as well as the effect of colostrum. In the experimental trial carried out for this PhD lambs in Group III (BB wk1, BB wk6), IV (BB wk1, LH wk6), V (LH wk1, BB wk6) & VI (LH wk1, LH wk6) all received colostrum from their dams whereas lambs in Group I (BB wk6) & II (LH wk6) received artificial colostrum. Within these two groups (Group I & II) there were still differences in the disease outcome even when given the same colostrum and infective dose. This indicates that other factors are involved which affect the outcome of disease in individuals.

Overall, work carried out for this PhD has enhanced our understanding of the transmission dynamics of the parasite within cattle on the study farm and has indicated that adult cattle may act as a source of infection for young calves although further work is required to confirm this. Improved techniques developed throughout this thesis also advance our ability to quickly and accurately determine the species of *Cryptosporidium* being shed by individual animals as well as increasing the sensitivity of detection methods from adult cattle faeces. These improved techniques have also been used to examine the faeces of other host species such as deer, sheep and rabbits with great success (Wells et al., 2015). The experimental study looking at host resistance has contributed greatly to the field and has answered several questions about the development of immunity as well as the possibility of variation in pathogenicity between isolates of the same *Cryptosporidium* species. This work can be used as a basis to improve our understanding of disease pathogenesis, transmission,

development of host resistance and variation in pathogenicity of different *C*. *parvum* subtypes.

7.1 PhD Outcomes

The main outcomes from this thesis are:

- 1) New molecular tool developed to enable speciation and detection of mixed infections of *Cryptosporidium* in cattle faeces (Chapter 4).
- 2) Adapting (oocyst concentration and purification steps) and improving techniques (DNA extraction) to deal with different sample types can help to ensure accurate results (Chapter 3 & 5).
- 3) Adult cattle shed more *C. parvum* than previously thought (Chapter 6). The majority of adult cattle in this study shed *C. parvum* at some point, they have potential to be an important source of infection for other susceptible hosts.
- 4) Adult cattle, though shedding *C. parvum*, may not be an important source of infection for calves (Chapter 6) as they seem to shed different subtypes of *C. parvum*.
- 5) Cattle can be infected with and shed multiple species and subtypes of *Cryptosporidium* at any one time (Chapter 4, 5 & 6).
- 6) Clinical disease does not correlate with oocyst shedding (Chapter 3, 5 & 6). Animals can be shedding oocysts without diarrhoea.
- 7) Cattle do not develop complete sterile immunity to infection with *Cryptosporidium* (Chapter 5 & 6).
- Lambs do not develop immunity to infection with *Cryptosporidium* in the first six weeks of life although there is a reduction in disease severity at 6 weeks of age (Chapter 3).

- 9) Infection with one isolate does not provide protection against challenge with the same or a different isolate (Chapter 3).
- 10) Different *C. parvum* isolates can manifest differently in the host in terms of oocyst output and clinical disease (Chapter 3).
- Evidence that there is extensive variation in the disease outcome in individual animals in terms of oocyst output and disease manifestation (Chapter 3) indicating that host factors are very important in the outcome of disease.
- 12) A higher infective dose of *C. parvum* leads to a shorter prepatent period, even in animals with previous exposure (Chapter 3).

List of References

- Abe, N., Iseki, M., 2004. Identification of *Cryptosporidium* isolates from cockatiels by direct sequencing of the PCR-amplified small subunit ribosomal RNA gene. Parasitol Res 92, 523-526.
- Abu, S.N., Jori, F., Xiao, L., Rikhotso, O., Thompson, P.N., 2013. Molecular characterization of *Cryptosporidium* species at the wildlife/livestock interface of the Kruger National Park, South Africa. Comp Immunol Microbiol Infect. Dis 36, 295-302.
- Adamu, H., Petros, B., Zhang, G., Kassa, H., Amer, S., Ye, J., Feng, Y., Xiao, L., 2014. Distribution and clinical manifestations of *Cryptosporidium* species and subtypes in HIV/AIDS patients in Ethiopia. PLoS Negl. Trop. Dis 8, e2831.
- Alvarez-Pellitero, P., Quiroga, M.I., Sitja-Bobadilla, A., Redondo, M.J., Palenzuela, O., Padros, F., Vazquez, S., Nieto, J.M., 2004. *Cryptosporidium scophthalmi* n. sp. (Apicomplexa: Cryptosporidiidae) from cultured turbot *Scophthalmus maximus*. Light and electron microscope description and histopathological study. Dis Aquat. Organ 62, 133-145.
- Alvarez-Pellitero, P., Sitja-Bobadilla, A., 2002. Cryptosporidium molnari n. sp. (Apicomplexa: Cryptosporidiidae) infecting two marine fish species, Sparus aurata L. and Dicentrarchus labrax L. Int J Parasitol 32, 1007-1021.
- Alves, M., Matos, O., Pereira, D.F., I, Delgado, E., Lourenco, A.M., Antunes, F., 2001. Multilocus genotyping of *Cryptosporidium* isolates from human HIVinfected and animal hosts. J Eukaryot. Microbiol Suppl, 17S-18S.
- Alves, M., Xiao, L., Antunes, F., Matos, O., 2006. Distribution of *Cryptosporidium* subtypes in humans and domestic and wild ruminants in Portugal. Parasitol Res 99, 287-292.
- Alves, M., Xiao, L., Sulaiman, I., Lal, A.A., Matos, O., Antunes, F., 2003. Subgenotype analysis of *Cryptosporidium* isolates from humans, cattle, and zoo ruminants in Portugal. J Clin. Microbiol 41, 2744-2747.
- Amadi, B., Mwiya, M., Musuku, J., Watuka, A., Sianongo, S., Ayoub, A., Kelly, P., 2002. Effect of nitazoxanide on morbidity and mortality in Zambian children with cryptosporidiosis: a randomised controlled trial. Lancet 360, 1375-1380.
- Amer, S., Harfoush, M., He, H., 2010a. Molecular and phylogenetic analyses of *Cryptosporidium* SPP from dairy cattle in Egypt. J Egypt. Soc. Parasitol 40, 349-366.

- Amer, S., Honma, H., Ikarashi, M., Tada, C., Fukuda, Y., Suyama, Y., Nakai, Y., 2010b. *Cryptosporidium* genotypes and subtypes in dairy calves in Egypt. Vet Parasitol 169, 382-386.
- Anderson, B.C., 1987. Abomasal cryptosporidiosis in cattle. Vet Pathol 24, 235-238.
- Appelbee, A.J., Thompson, R.C., Olson, M.E., 2005. *Giardia* and *Cryptosporidium* in mammalian wildlife--current status and future needs. Trends Parasitol 21, 370-376.
- Ares-Mazas, E., Lorenzo, M.J., Casal, J.A., Fernandez da, P.B., Castro, J.A., Freire, F., 1997. Effect of a commercial disinfectant ('Virkon') on mouse experimental infection by *Cryptosporidium parvum*. J Hosp. Infect. 36, 141-145.
- Atwill, E.R., Harp, J.A., Jones, T., Jardon, P.W., Checel, S., Zylstra, M., 1998. Evaluation of periparturient dairy cows and contact surfaces as a reservoir of *Cryptosporidium parvum* for calfhood infection. Am. J Vet Res 59, 1116-1121.
- Atwill, E.R., Johnson, E., Klingborg, D.J., Veserat, G.M., Markegard, G., Jensen, W.A., Pratt, D.W., Delmas, R.E., George, H.A., Forero, L.C., Philips, R.L., Barry, S.J., McDougald, N.K., Gildersleeve, R.R., Frost, W.E., 1999a. Age, geographic, and temporal distribution of fecal shedding of *Cryptosporidium parvum* oocysts in cow-calf herds. Am. J Vet Res 60, 420-425.
- Atwill, E.R., Johnson, E.M., Pereira, M.G., 1999b. Association of herd composition, stocking rate, and duration of calving season with fecal shedding of *Cryptosporidium parvum* oocysts in beef herds. J Am. Vet Med Assoc. 215, 1833-1838.
- Atwill, E.R., Pereira, M.G., 2003. Lack of detectable shedding of *Cryptosporidium parvum* oocysts by periparturient dairy cattle. J Parasitol 89, 1234-1236.
- Barker, I.K., Carbonell, P.L., 1974. Cryptosporidium agni sp.n. from lambs, and cryptosporidium bovis sp.n. from a calf, with observations on the oocyst.
 Z. Parasitenkd. 44, 289-298.
- Barnes, D.A., Bonnin, A., Huang, J.X., Gousset, L., Wu, J., Gut, J., Doyle, P., Dubremetz, J.F., Ward, H., Petersen, C., 1998. A novel multi-domain mucin-like glycoprotein of *Cryptosporidium parvum* mediates invasion. Mol. Biochem. Parasitol 96, 93-110.
- Baxby, D., Hart, C.A., Blundell, N., 1985. Shedding of oocysts by immunocompetent individuals with cryptosporidiosis. J Hyg. (Lond) 95, 703-709.
- Bird, R.G., Smith, M.D., 1980. Cryptosporidiosis in man: parasite life cycle and fine structural pathology. J Pathol 132, 217-233.

- Blagburn, B.L., Lindsay, D.S., Hoerr, F.J., Davis, J.F., Giambrone, J.J., 1991. Pathobiology of cryptosporidiosis (*C. baileyi*) in broiler chickens. J Protozool. 38, 25S-28S.
- Blanchard, P.C., 2012. Diagnostics of Dairy and Beef Cattle Diarrhea. Veterinary Clinics of North America-Food Animal Practice 28, 443-+.
- Bornay-Llinares, F.J., da Silva, A.J., Moura, I.N., Myjak, P., Pietkiewicz, H., Kruminis-Lozowska, W., Graczyk, T.K., Pieniazek, N.J., 1999.
 Identification of *Cryptosporidium felis* in a cow by morphologic and molecular methods. Appl. Environ. Microbiol 65, 1455-1458.
- Borowski, H., Thompson, R.C., Armstrong, T., Clode, P.L., 2010. Morphological characterization of *Cryptosporidium parvum* life-cycle stages in an in vitro model system. Parasitology 137, 13-26.
- Broglia, A., Reckinger, S., Caccio, S.M., Nockler, K., 2008. Distribution of *Cryptosporidium parvum* subtypes in calves in Germany. Vet Parasitol 154, 8-13.
- Brook, E.J., Anthony, H.C., French, N.P., Christley, R.M., 2009. Molecular epidemiology of *Cryptosporidium* subtypes in cattle in England. Vet J 179, 378-382.
- Brownstein, D.G., Strandberg, J.D., Montali, R.J., Bush, M., Fortner, J., 1977. *Cryptosporidium* in snakes with hypertrophic gastritis. Vet Pathol 14, 606-617.
- Bukhari, Z., Smith, H.V., 1996. Detection of *Cryptosporidium muris* oocysts in the faeces of adult dairy cattle in Scotland. Vet Rec. 138, 207-208.
- Caccio, S.M., 2005. Molecular epidemiology of human cryptosporidiosis. Parassitologia 47, 185-192.
- Cama, V., Gilman, R.H., Vivar, A., Ticona, E., Ortega, Y., Bern, C., Xiao, L., 2006. Mixed *Cryptosporidium* infections and HIV. Emerging Infectious Diseases 12, 1025-1028.
- Cama, V.A., Bern, C., Roberts, J., Cabrera, L., Sterling, C.R., Ortega, Y., Gilman, R.H., Xiao, L., 2008. *Cryptosporidium* species and subtypes and clinical manifestations in children, Peru. Emerg. Infect. Dis 14, 1567-1574.
- Cama, V.A., Ross, J.M., Crawford, S., Kawai, V., Chavez-Valdez, R., Vargas, D., Vivar, A., Ticona, E., Navincopa, M., Williamson, J., Ortega, Y., Gilman, R.H., Bern, C., Xiao, L., 2007. Differences in clinical manifestations among *Cryptosporidium* species and subtypes in HIV-infected persons. J Infect. Dis 196, 684-691.
- Casemore, D.P., Watkins, J., 1998. Review of Disinfection and Associated Studies. In: pp. 1-56.
- Castro-Hermida, J.A., Carro-Corral, C., Gonzalez-Warleta, M., Mezo, M., 2006a. Prevalence and intensity of infection of *Cryptosporidium* spp. and *Giardia*

duodenalis in dairy cattle in Galicia (NW Spain). J Vet Med B Infect. Dis Vet Public Health 53, 244-246.

- Castro-Hermida, J.A., Gonzalez-Losada, Y.A., Ares-Mazas, E., 2002. Prevalence of and risk factors involved in the spread of neonatal bovine cryptosporidiosis in Galicia (NW Spain). Vet Parasitol 106, 1-10.
- Castro-Hermida, J.A., Pors, I., Mendez-Hermida, F., Ares-Mazas, E., Chartier, C., 2006b. Evaluation of two commercial disinfectants on the viability and infectivity of *Cryptosporidium parvum* oocysts. Vet J 171, 340-345.
- Cevallos, A.M., Bhat, N., Verdon, R., Hamer, D.H., Stein, B., Tzipori, S., Pereira, M.E., Keusch, G.T., Ward, H.D., 2000. Mediation of *Cryptosporidium parvum* infection in vitro by mucin-like glycoproteins defined by a neutralizing monoclonal antibody. Infect. Immun. 68, 5167-5175.
- Chalmers, R.M., Campbell, B., Crouch, N., Davies, A.P., 2010. Clinical laboratory practices for detection and reporting of *Cryptosporidium* in community cases of diarrhoea in the United Kingdom, 2008. Euro. Surveill 15.
- Chalmers, R.M., Campbell, B.M., Crouch, N., Charlett, A., Davies, A.P., 2011. Comparison of diagnostic sensitivity and specificity of seven *Cryptosporidium* assays used in the UK. J Med Microbiol 60, 1598-1604.
- Chalmers, R.M., Elwin, K., Thomas, A.L., Guy, E.C., Mason, B., 2009a. Long-term *Cryptosporidium* typing reveals the aetiology and species-specific epidemiology of human cryptosporidiosis in England and Wales, 2000 to 2003. Euro. Surveill 14.
- Chalmers, R.M., Giles, M., 2010. Zoonotic cryptosporidiosis in the UK challenges for control. J Appl. Microbiol 109, 1487-1497.
- Chalmers, R.M., Grinberg, A., 2005. Significance of *Cryptosporidium parvum* in horses. Vet Rec. 156, 688.
- Chalmers, R.M., Katzer, F., 2013. Looking for *Cryptosporidium*: the application of advances in detection and diagnosis. Trends in Parasitology 29, 237-251.
- Chalmers, R.M., Robinson, G., Elwin, K., Hadfield, S.J., Xiao, L., Ryan, U., Modha, D., Mallaghan, C., 2009b. *Cryptosporidium* sp. rabbit genotype, a newly identified human pathogen. Emerg. Infect. Dis 15, 829-830.
- Chappell, C.L., Okhuysen, P.C., Sterling, C.R., DuPont, H.L., 1996. *Cryptosporidium parvum*: intensity of infection and oocyst excretion patterns in healthy volunteers. J Infect. Dis 173, 232-236.
- Chappell, C.L., Okhuysen, P.C., Sterling, C.R., Wang, C., Jakubowski, W., DuPont, H.L., 1999. Infectivity of *Cryptosporidium parvum* in healthy adults with pre-existing anti-C. *parvum* serum immunoglobulin G. Am. J Trop. Med Hyg. 60, 157-164.

- Cho, M., Yoon, J., 2007. Quantitative evaluation and application of *Cryptosporidium parvum* inactivation with ozone treatment. Water Sci Technol. 55, 241-250.
- Clark, D.P., Sears, C.L., 1996. The pathogenesis of cryptosporidiosis. Parasitology Today 12, 221-225.
- Coetzee, N., Edeghere, O., Orendi, J., Chalmers, R., Morgan, L., 2008. A swimming pool-associated outbreak of cryptosporidiosis in Staffordshire, England, October to December 2007. Euro. Surveill 13, ii.
- Cole, D.J., Cohen, N.D., Snowden, K., Smith, R., 1998. Prevalence of and risk factors for fecal shedding of *Cryptosporidium parvum* oocysts in horses. J Am. Vet Med Assoc. 213, 1296-1302.
- Connelly, L., Craig, B.H., Jones, B., Alexander, C.L., 2013. Genetic diversity of *Cryptosporidium* spp. within a remote population of Soay Sheep on St. Kilda Islands, Scotland. Appl. Environ. Microbiol 79, 2240-2246.
- Couto, M.C., Lima, M.D., Bomfim, T.C., 2013. New *Cryptosporidium parvum* subtypes of IIa subfamily in dairy calves from Brazil. Acta Trop. 130C, 117-122.
- Current, W.L., Bick, P.H., 1989. Immunobiology of *Cryptosporidium* spp. Pathol Immunopathol Res 8, 141-160.
- Current, W.L., Reese, N.C., Ernst, J.V., Bailey, W.S., Heyman, M.B., Weinstein, W.M., 1983. Human cryptosporidiosis in immunocompetent and immunodeficient persons. Studies of an outbreak and experimental transmission. N. Engl. J Med 308, 1252-1257.
- Current, W.L., Snyder, D.B., 1988. Development of and serologic evaluation of acquired immunity to *Cryptosporidium baileyi* by broiler chickens. Poult. Sci 67, 720-729.
- Current, W.L., Upton, S.J., Haynes, T.B., 1986. The life cycle of *Cryptosporidium baileyi* n. sp. (Apicomplexa, Cryptosporidiidae) infecting chickens. J Protozool. 33, 289-296.
- de Graaf, D.C., Vanopdenbosch, E., Ortega-Mora, L.M., Abbassi, H., Peeters, J.E., 1999. A review of the importance of cryptosporidiosis in farm animals. International Journal for Parasitology 29, 1269-1287.
- De Waele, V., Berzano, M., Speybroeck, N., Berkvens, D., Mulcahy, G.M., Murphy, T.M., 2012. Peri-parturient rise of *Cryptosporidium* oocysts in cows: New insights provided by duplex quantitative real-time PCR. Veterinary Parasitology 189, 366-368.
- Denkinger, C.M., Harigopal, P., Ruiz, P., Dowdy, L.M., 2008. *Cryptosporidium parvum*-associated sclerosing cholangitis in a liver transplant patient. Transpl. Infect. Dis 10, 133-136.
- Diaz, P., Quilez, J., Chalmers, R.M., Panadero, R., Lopez, C., Sanchez-Acedo, C., Morrondo, P., Diez-Banos, P., 2010. Genotype and subtype analysis of

Cryptosporidium isolates from calves and lambs in Galicia (NW Spain). Parasitology 137, 1187-1193.

- DuPont, H.L., Chappell, C.L., Sterling, C.R., Okhuysen, P.C., Rose, J.B., Jakubowski, W., 1995. The infectivity of *Cryptosporidium parvum* in healthy volunteers. N. Engl. J Med 332, 855-859.
- Duranti, A., Caccio, S.M., Pozio, E., Di, E.A., De, C.M., Battisti, A., Scaramozzino, P., 2009. Risk factors associated with *Cryptosporidium parvum* infection in cattle. Zoonoses Public Health 56, 176-182.
- Elwin, K., Hadfield, S.J., Robinson, G., Chalmers, R.M., 2012a. The epidemiology of sporadic human infections with unusual cryptosporidia detected during routine typing in England and Wales, 2000-2008. Epidemiol. Infect. 140, 673-683.
- Elwin, K., Hadfield, S.J., Robinson, G., Crouch, N.D., Chalmers, R.M., 2012b. Cryptosporidium viatorum n. sp (Apicomplexa: Cryptosporidiidae) among travellers returning to Great Britain from the Indian subcontinent, 2007-2011. International Journal for Parasitology 42, 675-682.
- Enemark, H.L., Ahrens, P., Bille-Hansen, V., Heegaard, P.M., Vigre, H., Thamsborg, S.M., Lind, P., 2003. *Cryptosporidium parvum*: infectivity and pathogenicity of the 'porcine' genotype. Parasitology 126, 407-416.
- Enemark, H.L., Ahrens, P., Lowery, C.J., Thamsborg, S.M., Enemark, J.M., Bille-Hansen, V., Lind, P., 2002. *Cryptosporidium andersoni* from a Danish cattle herd: identification and preliminary characterisation. Vet Parasitol 107, 37-49.
- Esteban, E., Anderson, B.C., 1995. *Cryptosporidium muris*: prevalence, persistency, and detrimental effect on milk production in a drylot dairy. J Dairy Sci 78, 1068-1072.
- Faubert, G.M., Litvinsky, Y., 2000. Natural transmission of *Cryptosporidium parvum* between dams and calves on a dairy farm. J Parasitol 86, 495-500.
- Fayer, R., 2010. Taxonomy and species delimitation in *Cryptosporidium*. Experimental Parasitology 124, 90-97.
- Fayer, R., Ellis, W., 1993. Paromomycin is effective as prophylaxis for cryptosporidiosis in dairy calves. J Parasitol 79, 771-774.
- Fayer, R., Gasbarre, L., Pasquali, P., Canals, A., Almeria, S., Zarlenga, D., 1998. *Cryptosporidium parvum* infection in bovine neonates: dynamic clinical, parasitic and immunologic patterns. Int J Parasitol 28, 49-56.
- Fayer, R., Morgan, U., Upton, S.J., 2000a. Epidemiology of *Cryptosporidium*: transmission, detection and identification. Int J Parasitol 30, 1305-1322.
- Fayer, R., Santin, M., 2009. *Cryptosporidium xiaoi* n. sp. (Apicomplexa: Cryptosporidiidae) in sheep (*Ovis aries*). Vet Parasitol 164, 192-200.

- Fayer, R., Santin, M., Dargatz, D., 2010a. Species of *Cryptosporidium* detected in weaned cattle on cow-calf operations in the United States. Vet Parasitol 170, 187-192.
- Fayer, R., Santin, M., Macarisin, D., 2010b. *Cryptosporidium ubiquitum* n. sp. in animals and humans. Vet Parasitol 172, 23-32.
- Fayer, R., Santin, M., Trout, J.M., 2007. Prevalence of Cryptosporidium species and genotypes in mature dairy cattle on farms in eastern United States compared with younger cattle from the same locations. Vet Parasitol 145, 260-266.
- Fayer, R., Santin, M., Trout, J.M., 2008. Cryptosporidium ryanae n. sp. (Apicomplexa: Cryptosporidiidae) in cattle (Bos taurus). Vet Parasitol 156, 191-198.
- Fayer, R., Santin, M., Trout, J.M., Greiner, E., 2006. Prevalence of species and genotypes of *Cryptosporidium* found in 1-2-year-old dairy cattle in the eastern United States. Veterinary Parasitology 135, 105-112.
- Fayer, R., Santin, M., Xiao, L., 2005. *Cryptosporidium bovis* n. sp. (Apicomplexa: Cryptosporidiidae) in cattle (*Bos taurus*). J Parasitol 91, 624-629.
- Fayer, R., Trout, J.M., Graczyk, T.K., Lewis, E.J., 2000b. Prevalence of *Cryptosporidium, Giardia* and *Eimeria* infections in post-weaned and adult cattle on three Maryland farms. Vet Parasitol 93, 103-112.
- Fayer, R., Trout, J.M., Xiao, L., Morgan, U.M., Lai, A.A., Dubey, J.P., 2001. Cryptosporidium canis n. sp. from domestic dogs. J Parasitol 87, 1415-1422.
- Featherstone, C.A., Giles, M., Marshall, J.A., Mawhinney, I.C., Holliman, A., Pritchard, G.C., 2010. *Cryptosporidium* species in calves submitted for postmortem examination in England and Wales. Vet Rec. 167, 979-980.
- Feng, Y., Alderisio, K.A., Yang, W., Blancero, L.A., Kuhne, W.G., Nadareski, C.A., Reid, M., Xiao, L., 2007. *Cryptosporidium* genotypes in wildlife from a new york watershed. Appl. Environ. Microbiol 73, 6475-6483.
- Flanigan, T.P., 1994. Human immunodeficiency virus infection and cryptosporidiosis: protective immune responses. Am. J Trop. Med Hyg. 50, 29-35.
- Follet, J., Guyot, K., Leruste, H., Follet-Dumoulin, A., Hammouma-Ghelboun, O., Certad, G., Dei-Cas, E., Halama, P., 2011. *Cryptosporidium* infection in a veal calf cohort in France: molecular characterization of species in a longitudinal study. Vet Res 42, 116.
- Fujino, T., Matsui, T., Kobayashi, F., Haruki, K., Yoshino, Y., Kajima, J., Tsuji, M., 2002. The effect of heating against *Cryptosporidium* oocysts. J Vet Med Sci 64, 199-200.
- Gait, R., Soutar, R.H., Hanson, M., Fraser, C., Chalmers, R., 2008. Outbreak of cryptosporidiosis among veterinary students. Vet Rec. 162, 843-845.

- Garcia, L.S., Brewer, T.C., Bruckner, D.A., 1987. Fluorescence detection of *Cryptosporidium* oocysts in human fecal specimens by using monoclonal antibodies. J Clin. Microbiol 25, 119-121.
- Gatei, W., Barrett, D., Lindo, J.F., Eldemire-Shearer, D., Cama, V., Xiao, L., 2008. Unique *Cryptosporidium* population in HIV-infected persons, Jamaica. Emerg. Infect. Dis 14, 841-843.
- Geurden, T., Berkvens, D., Martens, C., Casaert, S., Vercruysse, J., Claerebout,
 E., 2007. Molecular epidemiology with subtype analysis of
 Cryptosporidium in calves in Belgium. Parasitology 134, 1981-1987.
- Giangaspero, A., Iorio, R., Paoletti, B., Traversa, D., Capelli, G., 2006. Molecular evidence for *Cryptosporidium* infection in dogs in Central Italy. Parasitol Res 99, 297-299.
- Giles, M., Chalmers, R., Pritchard, G., Elwin, K., Mueller-Doblies, D., Clifton-Hadley, F., 2009. *Cryptosporidium hominis* in a goat and a sheep in the UK. Vet Rec. 164, 24-25.
- Gomez Morales, M.A., La, R.G., Ludovisi, A., Onori, A.M., Pozio, E., 1999. Cytokine profile induced by *Cryptosporidium* antigen in peripheral blood mononuclear cells from immunocompetent and immunosuppressed persons with cryptosporidiosis. J Infect. Dis 179, 967-973.
- Goodwin, M.A., Brown, J., Resurreccion, R.S., Smith, J.A., 1996. Respiratory coccidiosis (*Cryptosporidium baileyi*) among northern Georgia broilers in one company. Avian Dis 40, 572-575.
- Gormley, F.J., Little, C.L., Chalmers, R.M., Rawal, N., Adak, G.K., 2011. Zoonotic cryptosporidiosis from petting farms, England and Wales, 1992-2009. Emerg. Infect. Dis 17, 151-152.
- Gow, S., Waldner, C., 2006. An examination of the prevalence of and risk factors for shedding of *Cryptosporidium* spp. and *Giardia* spp. in cows and calves from western Canadian cow-calf herds. Vet Parasitol 137, 50-61.
- Grinberg, A., Biggs, P.J., Dukkipati, V.S., George, T.T., 2013. Extensive intrahost genetic diversity uncovered in *Cryptosporidium parvum* using Next Generation Sequencing. Infect. Genet. Evol 15, 18-24.
- Grinberg, A., Pomroy, W.E., Carslake, H.B., Shi, Y., Gibson, I.R., Drayton, B.M., 2009. A study of neonatal cryptosporidiosis of foals in New Zealand. N. Z. Vet J 57, 284-289.
- Guarino, A., Canani, R.B., Pozio, E., Terracciano, L., Albano, F., Mazzeo, M., 1994. Enterotoxic effect of stool supernatant of *Cryptosporidium*-infected calves on human jejunum. Gastroenterology 106, 28-34.
- Hadfield, S.J., Robinson, G., Elwin, K., Chalmers, R.M., 2011. Detection and differentiation of *Cryptosporidium* spp. in human clinical samples by use of real-time PCR. J Clin. Microbiol 49, 918-924.

- Harp, J.A., Chen, W., Harmsen, A.G., 1992. Resistance of severe combined immunodeficient mice to infection with *Cryptosporidium parvum*: the importance of intestinal microflora. Infect. Immun. 60, 3509-3512.
- Harp, J.A., Goff, J.P., 1995. Protection of calves with a vaccine against *Cryptosporidium parvum*. J Parasitol 81, 54-57.
- Harp, J.A., Goff, J.P., 1998. Strategies for the control of *Cryptosporidium parvum* infection in calves. J Dairy Sci 81, 289-294.
- Harp, J.A., Wannemuehler, M.W., Woodmansee, D.B., Moon, H.W., 1988.
 Susceptibility of germfree or antibiotic-treated adult mice to *Cryptosporidium parvum*. Infect. Immun. 56, 2006-2010.
- Harp, J.A., Woodmansee, D.B., Moon, H.W., 1990. Resistance of calves to *Cryptosporidium parvum*: effects of age and previous exposure. Infect. Immun. 58, 2237-2240.
- Hatkin, J.M., Giambrone, J.J., Blagburn, B.L., 1993. Kinetics of serum antibody responses in broiler chicks against *Cryptosporidium baileyi*. Avian Pathol 22, 525-532.
- Helmy, Y.A., Krucken, J., Nockler, K., von Samson-Himmelstjerna, G., Zessin, K.H., 2013. Molecular epidemiology of *Cryptosporidium* in livestock animals and humans in the Ismailia province of Egypt. Vet Parasitol 193, 15-24.
- Hijjawi, N., Ng, J., Yang, R., Atoum, M.F., Ryan, U., 2010. Identification of rare and novel *Cryptosporidium* GP60 subtypes in human isolates from Jordan. Exp Parasitol 125, 161-164.
- Hijjawi, N.S., Meloni, B.P., Ng'anzo, M., Ryan, U.M., Olson, M.E., Cox, P.T., Monis, P.T., Thompson, R.C., 2004. Complete development of *Cryptosporidium parvum* in host cell-free culture. Int J Parasitol 34, 769-777.
- Hijjawi, N.S., Meloni, B.P., Ryan, U.M., Olson, M.E., Thompson, R.C., 2002. Successful in vitro cultivation of *Cryptosporidium andersoni*: evidence for the existence of novel extracellular stages in the life cycle and implications for the classification of *Cryptosporidium*. Int J Parasitol 32, 1719-1726.

Hoepelman, I.M., 1996. Human cryptosporidiosis. Int J STD AIDS 7 Suppl 1, 28-33.

- Hojlyng, N., Jensen, B.N., 1988. Respiratory cryptosporidiosis in HIV-positive patients. Lancet 1, 590-591.
- Huang, D.B., Zhou, J., 2007. Effect of intensive handwashing in the prevention of diarrhoeal illness among patients with AIDS: a randomized controlled study. J Med Microbiol 56, 659-663.
- Huetink, R.E., van der Giessen, J.W., Noordhuizen, J.P., Ploeger, H.W., 2001. Epidemiology of *Cryptosporidium* spp. and *Giardia* duodenalis on a dairy farm. Vet Parasitol 102, 53-67.

- Imre, K., Darabus, G., 2011. Distribution of *Cryptosporidium* species, genotypes and *C. parvum* subtypes in cattle in European countries. Sci Parasitol 12, 1-9.
- Imre, K., Lobo, L.M., Matos, O., Popescu, C., Genchi, C., Darabus, G., 2011. Molecular characterisation of *Cryptosporidium* isolates from pre-weaned calves in Romania: is there an actual risk of zoonotic infections? Vet Parasitol 181, 321-324.
- Inman, L.R., Takeuchi, A., 1979. Spontaneous cryptosporidiosis in an adult female rabbit. Vet Pathol 16, 89-95.
- Innes, E.A., Bartley, P.M., Rocchi, M., Benavidas-Silvan, J., Burrells, A., Hotchkiss, E., Chianini, F., Canton, G., Katzer, F., 2011. Developing vaccines to control protozoan parasites in ruminants: dead or alive? Vet Parasitol 180, 155-163.
- Iseki, M., Maekawa, T., Moriya, K., Uni, S., Takada, S., 1989. Infectivity of *Cryptosporidium muris* (strain RN 66) in various laboratory animals. Parasitol Res 75, 218-222.
- Jarvie, B.D., Trotz-Williams, L.A., McKnight, D.R., Leslie, K.E., Wallace, M.M., Todd, C.G., Sharpe, P.H., Peregrine, A.S., 2005. Effect of halofuginone lactate on the occurrence of *Cryptosporidium parvum* and growth of neonatal dairy calves. J Dairy Sci 88, 1801-1806.
- Jeckel, S., Williams, C.L., 2009. Subclinical *Cryptosporidium* species infection in lambs. Vet Rec. 164, 186-187.
- Jenkins, M., Higgins, J., Kniel, K., Trout, J., Fayer, R., 2004. Protection of calves against cryptosporiosis by oral inoculation with gamma-irradiated *Cryptosporidium parvum* oocysts. J Parasitol 90, 1178-1180.
- Jervis, H.R., Merrill, T.G., Sprinz, H., 1966. Coccidiosis in the guinea pig small intestine due to a *Cryptosporidium*. Am. J Vet Res 27, 408-414.
- Jirku, M., Valigurova, A., Koudela, B., Krizek, J., Modry, D., Slapeta, J., 2008. New species of *Cryptosporidium Tyzzer*, 1907 (Apicomplexa) from amphibian host: morphology, biology and phylogeny. Folia Parasitol (Praha) 55, 81-94.
- Joachim, A., 2004. Human cryptosporidiosis: an update with special emphasis on the situation in Europe. J Vet Med B Infect. Dis Vet Public Health 51, 251-259.
- Johnson, E.H., Muirhead, D.E., Windsor, J.J., King, G.J., Al-Busaidy, R., Cornelius, R., 1999. Atypical outbreak of caprine cryptosporidiosis in the Sultanate of Oman. Vet Rec. 145, 521-524.
- Kaneta, Y., Nakai, Y., 1998. Survey of *Cryptosporidium* oocysts from adult cattle in a slaughter house. J Vet Med Sci 60, 585-588.

- Keidel, J., Daugschies, A., 2013. Integration of halofuginone lactate treatment and disinfection with p-chloro-m-cresol to control natural cryptosporidiosis in calves. Vet Parasitol 196, 321-326.
- Kim, K., Gooze, L., Petersen, C., Gut, J., Nelson, R.G., 1992. Isolation, sequence and molecular karyotype analysis of the actin gene of *Cryptosporidium parvum*. Mol. Biochem. Parasitol 50, 105-113.
- Klein, P., Kleinova, T., Volek, Z., Simunek, J., 2008. Effect of *Cryptosporidium parvum* infection on the absorptive capacity and paracellular permeability of the small intestine in neonatal calves. Veterinary Parasitology 152, 53-59.
- Koinari, M., Karl, S., Ng-Hublin, J., Lymbery, A.J., Ryan, U.M., 2013. Identification of novel and zoonotic *Cryptosporidium* species in fish from Papua New Guinea. Vet Parasitol 198, 1-9.
- Koudela, B., Kucerova, S., 1999. Role of acquired immunity and natural age resistance on course of *Isospora suis* coccidiosis in nursing piglets. Vet Parasitol 82, 93-99.
- Kvac, M., Hromadova, N., Kvetonova, D., Rost, M., Sak, B., 2011. Molecular characterization of *Cryptosporidium* spp. in pre-weaned dairy calves in the Czech Republic: absence of *C. ryanae* and management-associated distribution of *C. andersoni*, *C. bovis* and C. *parvum* subtypes. Vet Parasitol 177, 378-382.
- Kvac, M., Kestranova, M., Pinkova, M., Kvetonova, D., Kalinova, J., Wagnerova, P., Kotkova, M., Vitovec, J., Ditrich, O., McEvoy, J., Stenger, B., Sak, B., 2013. Cryptosporidium scrofarum n. sp (Apicomplexa: Cryptosporidiidae) in domestic pigs (Sus scrofa). Veterinary Parasitology 191, 218-227.
- Kvac, M., Kouba, M., Vitovec, J., 2006. Age-related and housing-dependence of *Cryptosporidium* infection of calves from dairy and beef herds in South Bohemia, Czech Republic. Vet Parasitol 137, 202-209.
- Kvac, M., Nemejc, K., Kestranova, M., Kvetonova, D., Wagnerova, P., Kotkova, M., Rost, M., Samkova, E., McEvoy, J., Sak, B., 2014. Age related susceptibility of pigs to *Cryptosporidium scrofarum* infection. Vet Parasitol 202, 330-334.
- Kvac, M., Vitovec, J., 2003. Prevalence and pathogenicity of *Cryptosporidium* andersoni in one herd of beef cattle. J Vet Med B Infect. Dis Vet Public Health 50, 451-457.
- Langkjaer, R.B., Vigre, H., Enemark, H.L., Maddox-Hyttel, C., 2007. Molecular and phylogenetic characterization of *Cryptosporidium* and *Giardia* from pigs and cattle in Denmark. Parasitology 134, 339-350.
- Le Blancq, S.M., Khramtsov, N.V., Zamani, F., Upton, S.J., Wu, T.W., 1997. Ribosomal RNA gene organization in *Cryptosporidium parvum*. Mol. Biochem. Parasitol 90, 463-478.

- Lefay, D., Naciri, M., Poirier, P., Chermette, R., 2000. Prevalence of *Cryptosporidium* infection in calves in France. Vet Parasitol 89, 1-9.
- Lefay, D., Naciri, M., Poirier, P., Chermette, R., 2001. Efficacy of halofuginone lactate in the prevention of cryptosporidiosis in suckling calves. Vet Rec. 148, 108-112.
- Levine, N.D., 1980. Some corrections of coccidian (Apicomplexa: Protozoa) nomenclature. J Parasitol 66, 830-834.
- Lindergard, G., Nydam, D.V., Wade, S.E., Schaaf, S.L., Mohammed, H.O., 2003. A novel multiplex polymerase chain reaction approach for detection of four human infective *Cryptosporidium* isolates: *Cryptosporidium parvum*, types H and C, *Cryptosporidium canis*, and *Cryptosporidium felis* in fecal and soil samples. Journal of Veterinary Diagnostic Investigation 15, 262-267.
- Lindergard, G., Wade, S.E., Schaaf, S., Barwick, R.S., Mohammed, H.O., 2001. Detection of *Cryptosporidium* oocysts in soil samples by enzyme-linked immunoassay. Vet Parasitol 94, 163-176.
- Lindsay, D.S., Blagburn, B.L., Ernst, J.V., Current, W.L., 1985. Experimental coccidiosis (*Isospora suis*) in a litter of feral piglets. J Wildl Dis 21, 309-310.
- Lindsay, D.S., Blagburn, B.L., Sundermann, C.A., Giambrone, J.J., 1988. Effect of broiler chicken age on susceptibility to experimentally induced *Cryptosporidium baileyi* infection. Am. J Vet Res 49, 1412-1414.
- Lindsay, D.S., Blagburn, B.L., Sundermann, C.A., Hoerr, F.J., Ernest, J.A., 1986. Experimental *Cryptosporidium* infections in chickens: oocyst structure and tissue specificity. Am. J Vet Res 47, 876-879.
- Lindsay, D.S., Upton, S.J., Owens, D.S., Morgan, U.M., Mead, J.R., Blagburn, B.L., 2000. *Cryptosporidium andersoni* n. sp. (Apicomplexa: Cryptosporiidae) from cattle, Bos taurus. J Eukaryot. Microbiol 47, 91-95.
- Lopez-Velez, R., Tarazona, R., Garcia, C.A., Gomez-Mampaso, E., Guerrero, A., Moreira, V., Villanueva, R., 1995. Intestinal and extraintestinal cryptosporidiosis in AIDS patients. Eur. J Clin. Microbiol Infect. Dis 14, 677-681.
- Lorenzo, M.J., Ares-Mazas, E., Villacorta Martinez, d.M., I, 1993. Detection of oocysts and IgG antibodies to *Cryptosporidium parvum* in asymptomatic adult cattle. Vet Parasitol 47, 9-15.
- Lucio-Forster, A., Griffiths, J.K., Cama, V.A., Xiao, L., Bowman, D.D., 2010. Minimal zoonotic risk of cryptosporidiosis from pet dogs and cats. Trends Parasitol 26, 174-179.
- Mac Kenzie, W.R., Hoxie, N.J., Proctor, M.E., Gradus, M.S., Blair, K.A.,
 Peterson, D.E., Kazmierczak, J.J., Addiss, D.G., Fox, K.R., Rose, J.B., .,
 1994. A massive outbreak in Milwaukee of *Cryptosporidium* infection
 transmitted through the public water supply. N. Engl. J Med 331, 161-167.

- MacLeod, A., Turner, C.M., Tait, A., 1999. A high level of mixed *Trypanosoma brucei* infections in tsetse flies detected by three hypervariable minisatellites. Molecular and Biochemical Parasitology 102, 237-248.
- Majewska, A.C., Solarczyk, P., Tamang, L., Graczyk, T.K., 2004. Equine *Cryptosporidium parvum* infections in western Poland. Parasitol Res 93, 274-278.
- Mallon, M., MacLeod, A., Wastling, J., Smith, H., Reilly, B., Tait, A., 2003. Population structures and the role of genetic exchange in the zoonotic pathogen *Cryptosporidium parvum*. J Mol. Evol 56, 407-417.
- Mason, B.W., Chalmers, R.M., Carnicer-Pont, D., Casemore, D.P., 2010. A *Cryptosporidium hominis* outbreak in north-west Wales associated with low oocyst counts in treated drinking water. J Water Health 8, 299-310.
- Mason, R.W., Hartley, W.J., Tilt, L., 1981. Intestinal cryptosporidiosis in a kid goat. Aust. Vet J 57, 386-388.
- McAllister, T.A., Olson, M.E., Fletch, A., Wetzstein, M., Entz, T., 2005. Prevalence of *Giardia* and *Cryptosporidium* in beef cows in southern Ontario and in beef calves in southern British Columbia. Can. Vet J 46, 47-55.
- McDonald, V., Bancroft, G.J., 1994. Mechanisms of innate and acquired resistance to *Cryptosporidium parvum* infection in SCID mice. Parasite Immunol 16, 315-320.
- McDonald, V., Robinson, H.A., Kelly, J.P., Bancroft, G.J., 1996. Immunity to *Cryptosporidium muris* infection in mice is expressed through gut CD4+ intraepithelial lymphocytes. Infect. Immun. 64, 2556-2562.
- McKenzie, D.M., Diffay, B.C., 2000. Diarrhoea associated with cryptosporidial oocyst shedding in a quarterhorse stallion. Aust. Vet J 78, 27-28.
- Mead, J.R., You, X., 1998. Susceptibility differences to *Cryptosporidium parvum* infection in two strains of gamma interferon knockout mice. J Parasitol 84, 1045-1048.
- Meuten, D.J., Van Kruiningen, H.J., Lein, D.H., 1974. Cryptosporidiosis in a calf. J Am. Vet Med Assoc. 165, 914-917.
- Miller, R.A., Bronsdon, M.A., Morton, W.R., 1990. Experimental cryptosporidiosis in a primate model. J Infect. Dis 161, 312-315.
- Moore, D.A., Atwill, E.R., Kirk, J.H., Brahmbhatt, D., Herrera, A.L., Hou, L., Singer, M.D., Miller, T.D., 2003. Prophylactic use of decoquinate for infections with *Cryptosporidium parvum* in experimentally challenged neonatal calves. J Am. Vet Med Assoc. 223, 839-845.
- Moore, J.A., Frenkel, J.K., 1991. Respiratory and enteric cryptosporidiosis in humans. Arch Pathol Lab Med 115, 1160-1162.

- Mor, S.M., Tumwine, J.K., Ndeezi, G., Srinivasan, M.G., Kaddu-Mulindwa, D.H., Tzipori, S., Griffiths, J.K., 2010. Respiratory cryptosporidiosis in HIVseronegative children in Uganda: potential for respiratory transmission. Clin. Infect. Dis 50, 1366-1372.
- Morgan, U.M., Thompson, R.C., 1999. The importance of genotyping isolates before assigning species. Parasitol Today 15, 80-81.
- Morgan, U.M., Xiao, L., Hill, B.D., O'Donoghue, P., Limor, J., Lal, A., Thompson, R.C., 2000. Detection of the *Cryptosporidium parvum* "human" genotype in a dugong (*Dugong dugon*). J Parasitol 86, 1352-1354.
- Morgan-Ryan, U.M., Fall, A., Ward, L.A., Hijjawi, N., Sulaiman, I., Fayer, R., Thompson, R.C., Olson, M., Lal, A., Xiao, L., 2002. *Cryptosporidium hominis* n. sp. (Apicomplexa: Cryptosporidiidae) from Homo sapiens. J Eukaryot. Microbiol 49, 433-440.
- Morrill, K.M., Conrad, E., Lago, A., Campbell, J., Quigley, J., Tyler, H., 2012. Nationwide evaluation of quality and composition of colostrum on dairy farms in the United States. J Dairy Sci 95, 3997-4005.
- Morrison, L.J., Mallon, M.E., Smith, H.V., MacLeod, A., Xiao, L., Tait, A., 2008. The population structure of the *Cryptosporidium parvum* population in Scotland: a complex picture. Infect. Genet. Evol 8, 121-129.
- Mosier, D.A., Oberst, R.D., 2000. Cryptosporidiosis. A global challenge. Ann. N. Y Acad. Sci 916, 102-111.
- Mtambo, M.M., Nash, A.S., Blewett, D.A., Smith, H.V., Wright, S., 1991. *Cryptosporidium* infection in cats: prevalence of infection in domestic and feral cats in the Glasgow area. Vet Rec. 129, 502-504.
- Naciri, M., Mancassola, R., Fort, G., Danneels, B., Verhaeghe, J., 2011. Efficacy of amine-based disinfectant KENOCOX on the infectivity of *Cryptosporidium parvum* oocysts. Vet Parasitol 179, 43-49.
- Ng, J., Yang, R., McCarthy, S., Gordon, C., Hijjawi, N., Ryan, U., 2011. Molecular characterization of *Cryptosporidium* and *Giardia* in pre-weaned calves in Western Australia and New South Wales. Vet Parasitol 176, 145-150.
- Ng, J.S., Eastwood, K., Walker, B., Durrheim, D.N., Massey, P.D., Porigneaux,
 P., Kemp, R., McKinnon, B., Laurie, K., Miller, D., Bramley, E., Ryan, U.,
 2012. Evidence of *Cryptosporidium* transmission between cattle and
 humans in northern New South Wales. Exp Parasitol 130, 437-441.
- Ng-Hublin, J.S., Combs, B., MacKenzie, B., Ryan, U., 2013. Human cryptosporidiosis diagnosed in Western Australia: a mixed infection with *Cryptosporidium meleagridis*, the *Cryptosporidium* mink genotype, and an unknown *Cryptosporidium* species. J Clin. Microbiol 51, 2463-2465.
- Nime, F.A., Burek, J.D., Page, D.L., Holscher, M.A., Yardley, J.H., 1976. Acute enterocolitis in a human being infected with the protozoan *Cryptosporidium*. Gastroenterology 70, 592-598.

- Nydam, D.V., Wade, S.E., Schaaf, S.L., Mohammed, H.O., 2001. Number of *Cryptosporidium parvum* oocysts or *Giardia* spp cysts shed by dairy calves after natural infection. Am. J Vet Res 62, 1612-1615.
- O'Connor, R.M., Wanyiri, J.W., Cevallos, A.M., Priest, J.W., Ward, H.D., 2007. *Cryptosporidium parvum* glycoprotein gp40 localizes to the sporozoite surface by association with gp15. Mol. Biochem. Parasitol 156, 80-83.
- O'Handley, R.M., Cockwill, C., McAllister, T.A., Jelinski, M., Morck, D.W., Olson, M.E., 1999. Duration of naturally acquired giardiosis and cryptosporidiosis in dairy calves and their association with diarrhea. J Am. Vet Med Assoc. 214, 391-396.
- Okhuysen, P.C., Chappell, C.L., Crabb, J.H., Sterling, C.R., DuPont, H.L., 1999. Virulence of three distinct *Cryptosporidium parvum* isolates for healthy adults. J Infect. Dis 180, 1275-1281.
- Okhuysen, P.C., Chappell, C.L., Sterling, C.R., Jakubowski, W., DuPont, H.L., 1998. Susceptibility and serologic response of healthy adults to reinfection with *Cryptosporidium parvum*. Infect. Immun. 66, 441-443.
- Ongerth, J.E., Stibbs, H.H., 1989. Prevalence of *Cryptosporidium* infection in dairy calves in western Washington. Am. J Vet Res 50, 1069-1070.
- Ortega-Mora, L.M., Requejo-Fernandez, J.A., Pilar-Izquierdo, M., Pereira-Bueno, J., 1999. Role of adult sheep in transmission of infection by *Cryptosporidium parvum* to lambs: confirmation of periparturient rise. Int J Parasitol 29, 1261-1268.
- Ortega-Mora, L.M., Wright, S.E., 1994. Age-related resistance in ovine cryptosporidiosis: patterns of infection and humoral immune response. Infect. Immun. 62, 5003-5009.
- Pakandl, M., Hlaskova, L., 2007. The reproduction of Eimeria flavescens and Eimeria intestinalis in suckling rabbits. Parasitol Res 101, 1435-1437.
- Panciera, R.J., Thomassen, R.W., Garner, F.M., 1971. Cryptosporidial Infection in a Calf. Vet Pathol 8, 479-484.
- Paraud, C., Pors, I., Chartier, C., 2010. Evaluation of oral tilmicosin efficacy against severe cryptosporidiosis in neonatal kids under field conditions. Vet Parasitol 170, 149-152.
- Park, J.H., Guk, S.M., Han, E.t., Shin, E.H., Kim, J.L., Chai, J.Y., 2006.
 Genotype analysis of *Cryptosporidium* spp. prevalent in a rural village in Hwasun-Gun, Republic of Korea. Korean Journal of Parasitologu 44, 27-33.
- Patel, S., Pedraza-Diaz, S., McLauchlin, J., 1999. The identification of *Cryptosporidium* species and *Cryptosporidium parvum* directly from whole faeces by analysis of a multiplex PCR of the 18S rRNA gene and by PCR/RFLP of the *Cryptosporidium* outer wall protein (COWP) gene. International Journal for Parasitology 29, 1241-1247.
- Pavlasek, I., Ryan, U., 2008. *Cryptosporidium varanii* takes precedence over C. saurophilum. Exp Parasitol 118, 434-437.
- Perrucci, S., Buggiani, C., Sgorbini, M., Cerchiai, I., Otranto, D., Traversa, D., 2011. *Cryptosporidium parvum* infection in a mare and her foal with foal heat diarrhoea. Vet Parasitol 182, 333-336.
- Perryman, L.E., Kapil, S.J., Jones, M.L., Hunt, E.L., 1999. Protection of calves against cryptosporidiosis with immune bovine colostrum induced by a *Cryptosporidium parvum* recombinant protein. Vaccine 17, 2142-2149.
- Petry, F., Robinson, H.A., McDonald, V., 1995. Murine infection model for maintenance and amplification of *Cryptosporidium parvum* oocysts. J Clin. Microbiol 33, 1922-1924.
- Pohjola, S., Oksanen, H., Jokipii, L., Jokipii, A.M., 1986. Outbreak of cryptosporidiosis among veterinary students. Scand. J Infect. Dis 18, 173-178.
- Pohlenz, J., Bemrick, W.J., Moon, H.W., Cheville, N.F., 1978. Bovine cryptosporidiosis: A transmission and scanning electron microscope study of some stage in the life cycle and of the host parasite relationship. Vet. Pathol. 15, 417-427.
- Pohlenz, J., Moon, H.W., Cheville, N.F., Bemrick, W.J., 1978. Cryptosporidiosis as a probable factor in neonatal diarrhea of calves. J Am. Vet Med Assoc. 172, 452-457.
- Pollock, K.G., Young, D., Smith, H.V., Ramsay, C.N., 2008. Cryptosporidiosis and filtration of water from Loch Lomond, Scotland. Emerg. Infect. Dis 14, 115-120.
- Pollok, R.C., Farthing, M.J., Bajaj-Elliott, M., Sanderson, I.R., McDonald, V., 2001. Interferon gamma induces enterocyte resistance against infection by the intracellular pathogen Cryptosporidium parvum. Gastroenterology 120, 99-107.
- Power, M.L., 2010. Biology of *Cryptosporidium* from marsupial hosts. Exp Parasitol 124, 40-44.
- Power, M.L., Ryan, U.M., 2008. A new species of *Cryptosporidium* (Apicomplexa: Cryptosporidiidae) from eastern grey kangaroos (*Macropus giganteus*). J Parasitol 94, 1114-1117.
- Pozio, E., Gomez Morales, M.A., Barbieri, F.M., La, R.G., 1992. *Cryptosporidium*: different behaviour in calves of isolates of human origin. Trans. R. Soc. Trop. Med Hyg. 86, 636-638.
- Preiser, G., Preiser, L., Madeo, L., 2003. An outbreak of cryptosporidiosis among veterinary science students who work with calves. J Am. Coll. Health 51, 213-215.
- Pritchard, G.C., Fleetwood, A.J., 1995. Cryptosporidiosis and farm visits. Vet Rec. 136, 179.

- Pritchard, G.C., Marshall, J.A., Giles, M., Mueller-Doblies, D., Sayers, A.R., Marshall, R.N., Elwin, K., Chalmers, R.M., 2008. *Cryptosporidium* species in lambs submitted for diagnostic postmortem examination in England and Wales. Vet Rec. 163, 688-689.
- Quigley, J.D., III, Martin, K.R., Bemis, D.A., Potgieter, L.N., Reinemeyer, C.R., Rohrbach, B.W., Dowlen, H.H., Lamar, K.C., 1994. Effects of housing and colostrum feeding on the prevalence of selected infectious organisms in feces of Jersey calves. J Dairy Sci 77, 3124-3131.
- Quigley, J.D., III, Martin, K.R., Bemis, D.A., Potgieter, L.N., Reinemeyer, C.R., Rohrbach, B.W., Dowlen, H.H., Lamar, K.C., 1995. Effects of housing and colostrum feeding on serum immunoglobulins, growth, and fecal scores of Jersey calves. J Dairy Sci 78, 893-901.
- Quigley, J.D., III, Schwab, C.G., Hylton, W.E., 1985. Development of rumen function in calves: nature of protein reaching the abomasum. J Dairy Sci 68, 694-702.
- Quilez, J., Sanchez-Acedo, C., Avendano, C., Del, C.E., Lopez-Bernad, F., 2005. Efficacy of two peroxygen-based disinfectants for inactivation of *Cryptosporidium parvum* oocysts. Appl. Environ. Microbiol 71, 2479-2483.
- Quilez, J., Sanchez-Acedo, C., Del, C.E., Clavel, A., Causape, A.C., 1996. Prevalence of *Cryptosporidium* and *Giardia* infections in cattle in Aragon (northeastern Spain). Vet Parasitol 66, 139-146.
- Quilez, J., Torres, E., Chalmers, R.M., Hadfield, S.J., Del, C.E., Sanchez-Acedo, C., 2008a. *Cryptosporidium* genotypes and subtypes in lambs and goat kids in Spain. Appl. Environ. Microbiol 74, 6026-6031.
- Quilez, J., Torres, E., Chalmers, R.M., Robinson, G., Del, C.E., Sanchez-Acedo, C., 2008b. *Cryptosporidium* species and subtype analysis from dairy calves in Spain. Parasitology 135, 1613-1620.
- Ralston, B., Thompson, R.C., Pethick, D., McAllister, T.A., Olson, M.E., 2010. *Cryptosporidium* and ersoni in Western Australian feedlot cattle. Aust. Vet J 88, 458-460.
- Ramirez, N.E., Ward, L.A., Sreevatsan, S., 2004. A review of the biology and epidemiology of cryptosporidiosis in humans and animals. Microbes Infect. 6, 773-785.
- Ren, X., Zhao, J., Zhang, L., Ning, C., Jian, F., Wang, R., Lv, C., Wang, Q., Arrowood, M.J., Xiao, L., 2012. *Cryptosporidium tyzzeri* n. sp. (Apicomplexa: Cryptosporidiidae) in domestic mice (*Mus musculus*). Exp Parasitol 130, 274-281.
- Rieux, A., Paraud, C., Pors, I., Chartier, C., 2013. Molecular characterization of *Cryptosporidium* isolates from pre-weaned calves in western France in relation to age. Vet Parasitol.
- Riggs, M.W., 2002. Recent advances in cryptosporidiosis: the immune response. Microbes Infect. 4, 1067-1080.

- Robertson, L.J., Campbell, A.T., Smith, H.V., 1992. Survival of *Cryptosporidium parvum* oocysts under various environmental pressures. Appl. Environ. Microbiol 58, 3494-3500.
- Robertson, L.J., Chalmers, R.M., 2013. Foodborne cryptosporidiosis: is there really more in Nordic countries? Trends Parasitol 29, 3-9.
- Robertson, L.J., Gjerde, B.K., Furuseth, H.E., 2010. The zoonotic potential of *Giardia* and *Cryptosporidium* in Norwegian sheep: a longitudinal investigation of 6 flocks of lambs. Vet Parasitol 171, 140-145.
- Robinson, G., Chalmers, R.M., Stapleton, C., Palmer, S.R., Watkins, J., Francis, C., Kay, D., 2011. A whole water catchment approach to investigating the origin and distribution of *Cryptosporidium* species. J Appl. Microbiol 111, 717-730.
- Robinson, G., Elwin, K., Chalmers, R.M., 2008. Unusual *Cryptosporidium* genotypes in human cases of diarrhea. Emerg. Infect. Dis 14, 1800-1802.
- Robinson, G., Wright, S., Elwin, K., Hadfield, S.J., Katzer, F., Bartley, P.M., Hunter, P.R., Nath, M., Innes, E.A., Chalmers, R.M., 2010. Re-description of *Cryptosporidium cuniculus* Inman and Takeuchi, 1979 (Apicomplexa: Cryptosporidiidae): morphology, biology and phylogeny. Int J Parasitol 40, 1539-1548.
- Rossignol, J.F., Ayoub, A., Ayers, M.S., 2001. Treatment of diarrhea caused by *Cryptosporidium parvum*: a prospective randomized, double-blind, placebo-controlled study of Nitazoxanide. J Infect. Dis 184, 103-106.
- Ryan, U., 2010. *Cryptosporidium* in birds, fish and amphibians. Exp Parasitol 124, 113-120.
- Ryan, U., Fayer, R., Xiao, L., 2014. *Cryptosporidium* species in humans and animals: current understanding and research needs. Parasitology 141, 1667-1685.
- Ryan, U., Paparini, A., Tong, K., Yang, R., Gibson-Kueh, S., O'Hara, A., Lymbery, A., Xiao, L., 2015. *Cryptosporidium huwi* n. sp. (Apicomplexa: Eimeriidae) from the guppy (*Poecilia reticulata*). Exp Parasitol 150, 31-35.
- Ryan, U., Xiao, L., Read, C., Zhou, L., Lal, A.A., Pavlasek, I., 2003a. Identification of novel *Cryptosporidium* genotypes from the Czech Republic. Appl. Environ. Microbiol 69, 4302-4307.
- Ryan, U.M., Bath, C., Robertson, I., Read, C., Elliot, A., McInnes, L., Traub, R., Besier, B., 2005. Sheep may not be an important zoonotic reservoir for *Cryptosporidium* and *Giardia* parasites. Appl. Environ. Microbiol 71, 4992-4997.
- Ryan, U.M., Monis, P., Enemark, H.L., Sulaiman, I., Samarasinghe, B., Read, C., Buddle, R., Robertson, I., Zhou, L., Thompson, R.C., Xiao, L., 2004. *Cryptosporidium suis* n. sp. (Apicomplexa: Cryptosporidiidae) in pigs (Sus scrofa). J Parasitol 90, 769-773.

- Ryan, U.M., Power, M., Xiao, L., 2008. *Cryptosporidium fayeri* n. sp. (Apicomplexa: Cryptosporidiidae) from the Red Kangaroo (Macropus rufus). J Eukaryot. Microbiol 55, 22-26.
- Ryan, U.M., Xiao, L., Read, C., Sulaiman, I.M., Monis, P., Lal, A.A., Fayer, R.,
 Pavlasek, I., 2003b. A redescription of *Cryptosporidium galli* Pavlasek,
 1999 (Apicomplexa: Cryptosporidiidae) from birds. J Parasitol 89, 809-813.
- Ryley, J.F., Meade, R., Hazelhurst, J., Robinson, T.E., 1976. Methods in coccidiosis research: separation of oocysts from faeces. Parasitology 73, 311-326.
- Sanford, S.E., Josephson, G.K., 1982. Bovine Cryptosporidiosis: Clinical and Pathological Findings in Forty-two Scouring Neonatal Calves. Can. Vet J 23, 343-347.
- Santin, M., 2013. Clinical and subclinical infections with *Cryptosporidium* in animals. New Zealand Veterinary Journal 61, 1-10.
- Santin, M., Trout, J.M., Fayer, R., 2008. A longitudinal study of cryptosporidiosis in dairy cattle from birth to 2 years of age. Vet Parasitol 155, 15-23.
- Santin, M., Trout, J.M., Xiao, L., Zhou, L., Greiner, E., Fayer, R., 2004. Prevalence and age-related variation of *Cryptosporidium* species and genotypes in dairy calves. Veterinary Parasitology 122, 103-117.
- Santin, M., Zarlenga, D.S., 2009. A multiplex polymerase chain reaction assay to simultaneously distinguish *Cryptosporidium* species of veterinary and public health concern in cattle. Veterinary Parasitology 166, 32-37.
- Sari, B., Arslan, M.O., Gicik, Y., Kara, M., Tasci, G.T., 2009. The prevalence of Cryptosporidium species in diarrhoeic lambs in Kars province and potential risk factors. Trop. Anim Health Prod. 41, 819-826.
- Scott, C.A., Smith, H.V., Mtambo, M.M., Gibbs, H.A., 1995. An epidemiological study of *Cryptosporidium parvum* in two herds of adult beef cattle. Vet Parasitol 57, 277-288.
- Shepherd, R.C., Reed, C.L., Sinha, G.P., 1988. Shedding of oocysts of *Cryptosporidium* in immunocompetent patients. J Clin. Pathol 41, 1104-1106.
- Sherwood, D., Angus, K.W., Snodgrass, D.R., Tzipori, S., 1982. Experimental cryptosporidiosis in laboratory mice. Infect. Immun. 38, 471-475.
- Silverlas, C., Blanco-Penedo, I., 2013. *Cryptosporidium* spp. in calves and cows from organic and conventional dairy herds. Epidemiology and Infection 141, 529-539.
- Silverlas, C., Mattsson, J.G., Insulander, M., Lebbad, M., 2012. Zoonotic transmission of *Cryptosporidium meleagridis* on an organic Swedish farm. Int J Parasitol 42, 963-967.

- Silverlas, C., Naslund, K., Bjorkman, C., Mattsson, J.G., 2010. Molecular characterisation of *Cryptosporidium* isolates from Swedish dairy cattle in relation to age, diarrhoea and region. Veterinary Parasitology 169, 289-295.
- Singh, B.B., Sharma, R., Kumar, H., Banga, H.S., Aulakh, R.S., Gill, J.P., Sharma, J.K., 2006. Prevalence of *Cryptosporidium parvum* infection in Punjab (India) and its association with diarrhea in neonatal dairy calves. Vet Parasitol 140, 162-165.
- Slavin, D., 1955. Cryptosporidium meleagridis (sp. nov.). J Comp Pathol 65, 262-266.
- Smith, H.V., Nichols, R.A., Mallon, M., MacLeod, A., Tait, A., Reilly, W.J., Browning, L.M., Gray, D., Reid, S.W., Wastling, J.M., 2005. Natural *Cryptosporidium hominis* infections in Scottish cattle. Veterinary Record 156, 710-711.
- Smith, R.P., Clifton-Hadley, F.A., Cheney, T., Giles, M., 2014. Prevalence and molecular typing of Cryptosporidium in dairy cattle in England and Wales and examination of potential on-farm transmission routes. Vet Parasitol 204, 111-119.
- Snodgrass, D.R., Angus, K.W., Gray, E.W., 1984. Experimental cryptosporidiosis in germfree lambs. J Comp Pathol 94, 141-152.
- Snyder, S.P., England, J.J., McChesney, A.E., 1978. Cryptosporidiosis in immunodeficient Arabian foals. Vet Pathol 15, 12-17.
- Spano, F., Putignani, L., McLauchlin, J., Casemore, D.P., Crisanti, A., 1997. PCR-RFLP analysis of the *Cryptosporidium* oocyst wall protein (COWP) gene discriminates between *C. wrairi* and *C. parvum*, and between *C. parvum* isolates of human and animal origin. FEMS Microbiol Lett. 150, 209-217.
- Striepen, B., 2013. Parasitic infections: Time to tackle cryptosporidiosis. Nature 503, 189-191.
- Strong, W.B., Gut, J., Nelson, R.G., 2000. Cloning and sequence analysis of a highly polymorphic *Cryptosporidium parvum* gene encoding a 60kilodalton glycoprotein and characterization of its 15- and 45-kilodalton zoite surface antigen products. Infect. Immun. 68, 4117-4134.
- Strong, W.B., Nelson, R.G., 2000. Preliminary profile of the *Cryptosporidium parvum* genome: an expressed sequence tag and genome survey sequence analysis. Mol. Biochem. Parasitol 107, 1-32.
- Sturdee, A.P., Bodley-Tickell, A.T., Archer, A., Chalmers, R.M., 2003. Long-term study of *Cryptosporidium* prevalence on a lowland farm in the United Kingdom. Vet Parasitol 116, 97-113.
- Sulaiman, I.M., Hira, P.R., Zhou, L., Al-Ali, F.M., Al-Shelahi, F.A., Shweiki, H.M., Iqbal, J., Khalid, N., Xiao, L., 2005. Unique endemicity of cryptosporidiosis in children in Kuwait. J Clin. Microbiol 43, 2805-2809.

- Sulaiman, I.M., Morgan, U.M., Thompson, R.C., Lal, A.A., Xiao, L., 2000. Phylogenetic relationships of *Cryptosporidium* parasites based on the 70kilodalton heat shock protein (HSP70) gene. Appl. Environ. Microbiol 66, 2385-2391.
- Sulaiman, I.M., Xiao, L., Lal, A.A., 1999. Evaluation of *Cryptosporidium parvum* genotyping techniques. Appl. Environ. Microbiol 65, 4431-4435.
- Sweeny, J.P., Ryan, U.M., Robertson, I.D., Jacobson, C., 2011a. Cryptosporidium and Giardia associated with reduced lamb carcase productivity. Vet Parasitol 182, 127-139.
- Tali, A., Addebbous, A., Asmama, S., Chabaa, L., Zougaghi, L., 2011. Respiratory cryptosporidiosis in two patients with HIV infection in a tertiary care hospital in Morocco. Ann. Biol Clin. (Paris) 69, 605-608.
- Tarazona, R., Lally, N.C., Dominguez-Carmona, M., Blewett, D.A., 1997. Characterization of secretory IgA responses in mice infected with *Cryptosporidium parvum*. Int J Parasitol 27, 417-423.
- Tessema, T.S., Schwamb, B., Lochner, M., Forster, I., Jakobi, V., Petry, F., 2009. Dynamics of gut mucosal and systemic Th1/Th2 cytokine responses in interferon-gamma and interleukin-12p40 knock out mice during primary and challenge *Cryptosporidium parvum* infection. Immunobiology 214, 454-466.
- Teunis, P.F., Chappell, C.L., Okhuysen, P.C., 2002. *Cryptosporidium* dose response studies: variation between isolates. Risk Anal. 22, 175-183.
- Theodos, C.M., 1998. Innate and cell-mediated immune responses to *Cryptosporidium parvum*. Adv. Parasitol 40, 87-119.
- Thompson, H.P., Dooley, J.S., Kenny, J., McCoy, M., Lowery, C.J., Moore, J.E., Xiao, L., 2007. Genotypes and subtypes of *Cryptosporidium* spp. in neonatal calves in Northern Ireland. Parasitol Res 100, 619-624.
- Thompson, R.C., Armson, A., Ryan, U.M., 2003. *Cryptosporidium*: From Molecules to Disease. Elsevier.
- Thompson, R.C., Olson, M.E., Zhu, G., Enomoto, S., Abrahamsen, M.S., Hijjawi, N.S., 2005. *Cryptosporidium* and cryptosporidiosis. Adv. Parasitol 59, 77-158.
- Traversa, D., 2010. Evidence for a new species of *Cryptosporidium* infecting tortoises: *Cryptosporidium ducismarci*. Parasit. Vectors 3, 21.
- Trotz-Williams, L.A., Jarvie, B.D., Peregrine, A.S., Duffield, T.F., Leslie, K.E., 2011. Efficacy of halofuginone lactate in the prevention of cryptosporidiosis in dairy calves. Vet Rec. 168, 509.
- Tumova, E., Skrivan, M., Marounek, M., Pavlasek, I., Ledvinka, Z., 2002. Performance and oocyst shedding in broiler chickens orally infected with *Cryptosporidium baileyi* and *Cryptosporidium meleagridis*. Avian Dis 46, 203-207.

- Tyzzer, E.E., 1910. An extracellular Coccidium, *Cryptosporidium Muris* (Gen. Et Sp. Nov.), of the gastric Glands of the Common Mouse. J Med Res 23, 487-510.
- Tzipori, S., 2002. Introduction. Cryptosporidiosis: current trends and challenges. Microbes Infect. 4, 1045.
- Tzipori, S., Smith, M., Halpin, C., Angus, K.W., Sherwood, D., Campbell, I., 1983. Experimental cryptosporidiosis in calves: clinical manifestations and pathological findings. Vet Rec. 112, 116-120.
- Tzipori, S., Ward, H., 2002. Cryptosporidiosis: biology, pathogenesis and disease. Microbes and Infection 4, 1047-1058.
- Ungar, B.L., 1990. Enzyme-linked immunoassay for detection of *Cryptosporidium* antigens in fecal specimens. J Clin. Microbiol 28, 2491-2495.
- Upton, S.J., Gillock, H.H., 1996. Infection dynamics of *Cryptosporidium parvum* in ICR outbred suckling mice. Folia Parasitol (Praha) 43, 101-106.
- Upton, S.J., McAllister, C.T., Freed, P.S., Barnard, S.M., 1989. *Cryptosporidium* spp. in wild and captive reptiles. J Wildl Dis 25, 20-30.
- Vetterling, J.M., Jervis, H.R., Merrill, T.G., Sprinz, H., 1971. *Cryptosporidium wrairi* sp. n. from the guinea pig Cavia porcellus, with an emendation of the genus. J Protozool. 18, 243-247.
- Vieira, L.S., Silva, M.B., Tolentino, A.C., Lima, J.D., Silva, A.C., 1997. Outbreak of cryptosporidiosis in dairy goats in Brazil. Vet Rec. 140, 427-428.
- Villacorta, I., Ares-Mazas, E., Lorenzo, M.J., 1991. *Cryptosporidium parvum* in cattle, sheep and pigs in Galicia (N.W. Spain). Vet Parasitol 38, 249-252.
- Vitovec, J., Hamadejova, K., Landova, L., Kvac, M., Kvetonova, D., Sak, B., 2006. Prevalence and pathogenicity of *Cryptosporidium suis* in pre- and post-weaned pigs. J Vet Med B Infect. Dis Vet Public Health 53, 239-243.
- Viu, M., Quilez, J., Sanchez-Acedo, C., Del, C.E., Lopez-Bernad, F., 2000. Field trial on the therapeutic efficacy of paromomycin on natural *Cryptosporidium parvum* infections in lambs. Vet Parasitol 90, 163-170.
- Wade, S.E., Mohammed, H.O., Schaaf, S.L., 2000. Prevalence of Giardia sp. Cryptosporidium parvum and Cryptosporidium andersoni (syn. C. muris) [correction of Cryptosporidium parvum and Cryptosporidium muris (C. andersoni)] in 109 dairy herds in five counties of southeastern New York. Vet Parasitol 93, 1-11.
- Waldron, L.S., Cheung-Kwok-Sang, C., Power, M.L., 2010. Wildlife-associated *Cryptosporidium fayeri* in human, Australia. Emerg. Infect. Dis 16, 2006-2007.
- Wang, R., Ma, G., Zhao, J., Lu, Q., Wang, H., Zhang, L., Jian, F., Ning, C., Xiao, L., 2011a. *Cryptosporidium andersoni* is the predominant species in post-weaned and adult dairy cattle in China. Parasitol Int 60, 1-4.

- Wang, R., Wang, H., Sun, Y., Zhang, L., Jian, F., Qi, M., Ning, C., Xiao, L., 2011b. Characteristics of *Cryptosporidium* transmission in preweaned dairy cattle in Henan, China. J Clin. Microbiol 49, 1077-1082.
- Wang, R., Zhang, L., Axen, C., Bjorkman, C., Jian, F., Amer, S., Liu, A., Feng, Y., Li, G., Lv, C., Zhao, Z., Qi, M., Dong, H., Wang, H., Sun, Y., Ning, C., Xiao, L., 2014a. *Cryptosporidium parvum* IId family: clonal population and dispersal from Western Asia to other geographical regions. Sci Rep. 4, 4208.
- Wang, Y., Yang, W., Cama, V., Wang, L., Cabrera, L., Ortega, Y., Bern, C., Feng, Y., Gilman, R., Xiao, L., 2014b. Population genetics of *Cryptosporidium meleagridis* in humans and birds: evidence for crossspecies transmission. Int J Parasitol 44, 515-521.
- Webster, J.P., Macdonald, D.W., 1995. Cryptosporidiosis reservoir in wild brown rats (*Rattus norvegicus*) in the UK. Epidemiol. Infect. 115, 207-209.
- Weiss, W.P., 2015. Factors affecting manure excretion by dairy cows. In.
- Wells, B., Shaw, H., Hotchkiss, E., Gilray, J., Ayton, R., Green, J., Katzer, F., Wells, A., Innes, E., 2015. Prevalence, species identification and genotyping *Cryptosporidium* from livestock and deer in a catchment in the Cairngorms with a history of a contaminated public water supply. Parasit. Vectors 8, 66.
- Widerstrom, M., Schonning, C., Lilja, M., Lebbad, M., Ljung, T., Allestam, G., Ferm, M., Bjorkholm, B., Hansen, A., Hiltula, J., Langmark, J., Lofdahl, M., Omberg, M., Reuterwall, C., Samuelsson, E., Widgren, K., Wallensten, A., Lindh, J., 2014. Large outbreak of *Cryptosporidium hominis* infection transmitted through the public water supply, Sweden. Emerg. Infect. Dis 20, 581-589.
- Wielinga, P.R., de, V.A., van der Goot, T.H., Mank, T., Mars, M.H., Kortbeek, L.M., van der Giessen, J.W., 2008. Molecular epidemiology of *Cryptosporidium* in humans and cattle in The Netherlands. Int J Parasitol 38, 809-817.
- Winter, G., Gooley, A.A., Williams, K.L., Slade, M.B., 2000. Characterization of a major sporozoite surface glycoprotein of *Cryptosporidum parvum*. Funct. Integr. Genomics 1, 207-217.
- Xiao, L., 2010. Molecular epidemiology of cryptosporidiosis: an update. Exp Parasitol 124, 80-89.
- Xiao, L., Alderisio, K., Limor, J., Royer, M., Lal, A.A., 2000a. Identification of species and sources of *Cryptosporidium* oocysts in storm waters with a small-subunit rRNA-based diagnostic and genotyping tool. Applied and Environmental Microbiology 66, 5492-5498.
- Xiao, L., Escalante, L., Yang, C., Sulaiman, I., Escalante, A.A., Montali, R.J., Fayer, R., Lal, A.A., 1999. Phylogenetic analysis of *Cryptosporidium* parasites based on the small-subunit rRNA gene locus. Appl. Environ. Microbiol 65, 1578-1583.

- Xiao, L., Fayer, R., 2008. Molecular characterisation of species and genotypes of *Cryptosporidium* and *Giardia* and assessment of zoonotic transmission. Int J Parasitol 38, 1239-1255.
- Xiao, L., Herd, R.P., 1994. Infection pattern of *Cryptosporidium* and *Giardia* in calves. Vet Parasitol 55, 257-262.
- Xiao, L., Herd, R.P., McClure, K.E., 1994. Periparturient rise in the excretion of *Giardia* sp. cysts and *Cryptosporidium parvum* oocysts as a source of infection for lambs. J Parasitol 80, 55-59.
- Xiao, L., Limor, J., Morgan, U.M., Sulaiman, I.M., Thompson, R.C., Lal, A.A., 2000b. Sequence differences in the diagnostic target region of the oocyst wall protein gene of *Cryptosporidium* parasites. Appl. Environ. Microbiol 66, 5499-5502.
- Xiao, L., Ryan, U.M., 2004. Cryptosporidiosis: an update in molecular epidemiology. Curr Opin. Infect. Dis 17, 483-490.
- Xiao, L., Ryan, U.M., Graczyk, T.K., Limor, J., Li, L., Kombert, M., Junge, R., Sulaiman, I.M., Zhou, L., Arrowood, M.J., Koudela, B., Modry, D., Lal, A.A., 2004. Genetic diversity of *Cryptosporidium* spp. in captive reptiles. Appl. Environ. Microbiol 70, 891-899.
- Xiao, L., Zhou, L., Santin, M., Yang, W., Fayer, R., 2007. Distribution of *Cryptosporidium parvum* subtypes in calves in eastern United States. Parasitol Res 100, 701-706.
- Yang, R., Fenwick, S., Potter, A., Ng, J., Ryan, U., 2011. Identification of novel *Cryptosporidium* genotypes in kangaroos from Western Australia. Vet Parasitol 179, 22-27.
- Yang, R., Jacobson, C., Gardner, G., Carmichael, I., Campbell, A.J., Ng-Hublin, J., Ryan, U., 2014. Longitudinal prevalence, oocyst shedding and molecular characterisation of *Cryptosporidium* species in sheep across four states in Australia. Vet Parasitol 200, 50-58.
- Ye, J., Xiao, L., Ma, J., Guo, M., Liu, L., Feng, Y., 2012. Anthroponotic enteric parasites in monkeys in public park, China. Emerg. Infect. Dis 18, 1640-1643.
- Ye, J., Xiao, L., Wang, Y., Wang, L., Amer, S., Roellig, D.M., Guo, Y., Feng, Y., 2013. Periparturient transmission of *Cryptosporidium xiaoi* from ewes to lambs. Vet Parasitol 197, 627-633.
- Yui, T., Nakajima, T., Yamamoto, N., Kon, M., Abe, N., Matsubayashi, M., Shibahara, T., 2014. Age-related detection and molecular characterization of *Cryptosporidium suis* and *Cryptosporidium scrofarum* in pre- and post-weaned piglets and adult pigs in Japan. Parasitol Res 113, 359-365.
- Zambriski, J.A., Nydam, D.V., Bowman, D.D., Bellosa, M.L., Burton, A.J., Linden, T.C., Liotta, J.L., Ollivett, T.L., Tondello-Martins, L., Mohammed, H.O., 2013a. Description of fecal shedding of

Cryptosporidium parvum oocysts in experimentally challenged dairy calves. Parasitology Research 112, 1247-1254.

- Zambriski, J.A., Nydam, D.V., Wilcox, Z.J., Bowman, D.D., Mohammed, H.O., Liotta, J.L., 2013b. *Cryptosporidium parvum*: Determination of ID50 and the dose response relationship in experimentally challenged dairy calves. Veterinary Parasitology 197, 104-112.
- Zhou, L., Kassa, H., Tischler, M.L., Xiao, L., 2004. Host-adapted Cryptosporidium spp. in Canada geese (Branta canadensis). Appl. Environ. Microbiol 70, 4211-4215.
- Zintl, A., Gray, J.S., Skerrett, H.E., Mulcahy, G., 2005. Possible mechanisms underlying age-related resistance to bovine babesiosis. Parasite Immunol 27, 115-120.
- Zintl, A., Proctor, A.F., Read, C., Dewaal, T., Shanaghy, N., Fanning, S., Mulcahy, G., 2009. The prevalence of *Cryptosporidium* species and subtypes in human faecal samples in Ireland. Epidemiol. Infect. 137, 270-277.